UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, DC 20549

F	\mathbf{O}	R	M	8-	.K
1	\	1	1 V I	()	- I J

CURRENT REPORT

Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): December 10, 2025

GLOBAL WATER RESOURCES, INC.

(Exact name of registrant as specified in its charter)

Delaware 001-37756 90-0632193

(State or other jurisdiction of incorporation or organization) (Commission File Number) (IRS Employer Identification No.)

21410 N. 19th Avenue #220
Phoenix, Arizona 85027

(Address of principal executive offices) (Zip Code)

Registrant's telephone number, including area code: (480) 360-7775

Not Applicable (Former name or former address, if changed since last report)

C	11 1	e e	d to simultaneously satisfy the filing obligation of the (see General Instruction A.2 below):			
	Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)					
	Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)					
	Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))					
	Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))					
Securities registered pursuant to Section 12(b) of the Act.						
	Title of each class	Trading Symbol	Name of each exchange on which registered			
Common Stock, par value \$0.01 per share		GWRS	The NASDAQ Stock Market, LLC			

Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§240.12b-2 of this chapter).

Emerging growth company □

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. \Box

Item 1.01 Entry into a Material Definitive Agreement.

Credit Agreement and Promissory Note

On December 10, 2025, Global Water Resources, Inc. (the "Company") entered into a credit agreement (the "Credit Agreement") with CoBANK, ACB, a federally-chartered instrumentality of the United States ("CoBANK"), and a related promissory note issued by the Company to CoBANK (the "Promissory Note" and together with the Credit Agreement, the "CoBANK Loan Agreements"), pursuant to which CoBANK provided the Company a term loan in the aggregate principal amount of \$15,000,000 (the "Term Loan") on the terms and subject to the conditions set forth in the CoBANK Loan Agreements.

The Term Loan bears interest at a fixed rate of 5.49% per annum, payable semi-annually on June 15 and December 15 of each year, beginning on June 15, 2026. The scheduled maturity date of the Term Loan is December 10, 2035, subject to certain prepayment requirements upon a Change in Control (as defined in the Credit Agreement).

The obligations under the CoBANK Loan Agreements are guaranteed by the following subsidiaries of the Company: Global Water, LLC, West Maricopa Combine, LLC, and Global Water Holdings, Inc. (together, the "Guarantors"). The obligations under the CoBANK Loan Agreements are also secured by a pledge of the equity interests in all direct and indirect subsidiaries of the Company and certain related rights and property.

The Credit Agreement contains a debt service coverage ratio financial maintenance covenant, as well as certain restrictive covenants that limit, among other things, the Company's ability to: create liens and other encumbrances; incur additional indebtedness; merge, liquidate or consolidate with another entity; dispose of or transfer assets; make distributions or other restricted payments; engage in certain affiliate transactions; and change the nature of the business. Additionally, the Credit Agreement contains a provision limiting the payment of dividends if the Company falls below a debt service coverage ratio of 1.25:1.00. The foregoing covenants are subject to various qualifications and limitations as set forth in the Credit Agreement.

Pursuant to the Credit Agreement, the Term Loan will be subject to certain customary events of default after which the Term Loan may be declared due and payable if not cured within the grace period or, in certain circumstances, may be declared due and payable immediately.

The foregoing summary of the terms of the Credit Agreement and the Promissory Note is qualified in its entirety by reference to the full text of such agreements, each of which is attached as Exhibits 10.1 and 10.2 to this Current Report on Form 8-K (this "Form 8-K"), respectively, and incorporated herein by reference.

Guaranty Agreements

On December 10, 2025, each of the Guarantors entered into a guaranty agreement (collectively, the "Guaranty Agreements") for the benefit of CoBANK, pursuant to which each Guarantor jointly and severally guaranteed the Company's obligations under the CoBANK Loan Agreements.

The foregoing summary of the terms of the Guaranty Agreements is qualified in its entirety by reference to the full text of such agreements, each of which is attached as Exhibits 10.3, 10.4, and 10.5 to this Form 8-K, and incorporated herein by reference.

Pledge and Security Agreements

On December 10, 2025, each of the Company and the Guarantors entered into a pledge and security agreement (collectively, the "Pledge and Security Agreements") with U.S. Bank Trust Company, National Association, as collateral agent for CoBANK, relating to the collateral securing the Company's obligations under the CoBANK Loan Agreements.

The foregoing summary of the terms of the Pledge and Security Agreements is qualified in its entirety by reference to the full text of such agreements, each of which is attached as Exhibits 10.6, 10.7, 10.8, and 10.9 to this Form 8-K, and incorporated herein by reference.

Item 2.03 Creation of a Direct Financial Obligation under an Off-Balance Sheet Arrangement of a Registrant

The information required by this Item 2.03 is included in Item 1.01 of this Form 8-K and incorporated herein by reference.

Item 9.01 Financial Statements and Exhibits.

(d) Exhibits

Exhibit No.	Description				
10.1	Credit Agreement, dated December 10, 2025, by and between Global Water Resources, Inc. and CoBANK, <u>ACB</u>				
10.2	<u>Promissory Note, dated December 10, 2025, by and between Global Water Resources, Inc. and CoBANK, ACB</u>				
10.3	Guaranty Agreement, dated December 10, 2025, by Global Water, LLC				
10.4	Guaranty Agreement, dated December 10, 2025, by West Maricopa Combine, LLC				
10.5	Guaranty Agreement, dated December 10, 2025, by Global Water Holdings, Inc.				
10.6	Pledge and Security Agreement, dated December 10, 2025, by and between Global Water Resources, Inc. and U.S. Bank Trust Company, National Association, as collateral agent				
10.7	Pledge and Security Agreement, dated December 10, 2025, by and between Global Water, LLC and U.S. Bank Trust Company, National Association, as collateral agent				
10.8	Pledge and Security Agreement, dated December 10, 2025, by and between West Maricopa Combine, LLC and U.S. Bank Trust Company, National Association, as collateral agent				
10.9	Pledge and Security Agreement, dated December 10, 2025, by and between Global Water Holdings, Inc. and U.S. Bank Trust Company, National Association, as collateral agent				
104	Cover Page Interactive Data File (formatted as Inline XBRL)				

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Date: December 16, 2025

GLOBAL WATER RESOURCES, INC.

/s/ Michael J. Liebman

Michael J. Liebman Chief Financial Officer

CREDIT AGREEMENT

THIS CREDIT AGREEMENT (this "Agreement") dated as of December 10, 2025 is entered into by and between GLOBAL WATER RESOURCES, INC., a Delaware corporation (the "Company"), and COBANK, ACB, a federally-chartered instrumentality of the United States (the "Bank"). In consideration of the agreements set forth herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Company and Bank agree as follows:

ARTICLE 1 Defined Terms; Accounting Principles. Certain capitalized terms used in this Agreement bear the definitions given to them in the Defined Terms attached hereto as Exhibit A and incorporated herein by this reference. Accounting terms used herein but not defined in this Agreement have the meanings given to them in accordance with GAAP. Except as otherwise specifically provided herein: (a) all computations made pursuant to this Agreement shall be made in accordance with GAAP; and (b) all financial statements shall be prepared in accordance with GAAP. For purposes of determining compliance with this Agreement, any election by the Company to measure any financial liability using fair value (as permitted by Financial Accounting Standards Board Accounting Standards Codification Topic No. 825-10-25 - Fair Value Option, International Accounting Standard 39 - Financial Instruments: Recognition and Measurement, or any similar accounting standard) will be disregarded and such determination made as if such election had not been made. If any computation of any financial ratio or requirement in this Agreement would be affected solely by any change in GAAP and the Company or Bank requests, the parties, shall, at no cost to the Company (other than payment obligations pursuant to Section 10.1), negotiate in Good Faith to amend such ratio or requirement to preserve the original intent thereof in light of such change, provided that, until so amended: (i) such ratio or requirement shall be computed in accordance with GAAP prior to such change; and (ii) the Company shall provide to Bank financial statements and other documents required under this Agreement or as reasonably requested hereunder setting forth a reconciliation between calculations of such ratio or requirement before and after giving effect to such change in GAAP.

ARTICLE 2 The Facilities.

2.1 Promissory Note. In the event the Company desires to borrow from Bank and Bank is willing to lend to the Company, or in the event the parties desire to consolidate any existing loans hereunder, the parties will enter into a promissory note (a "Promissory Note"). Each Promissory Note will set forth Bank's commitment to make a loan or loans (each, a "Loan") to the Company, the amount of the Loan(s), the purpose of the Loan(s), the interest rate or rate options applicable to the Loan(s), the repayment terms of the Loan(s), and any other terms and conditions applicable to the particular Loan(s). Each Promissory Note will also contain the Company's promise to make payments of interest on the unpaid principal balance of the Loan(s), and fees and premiums, if any, and to repay the principal balance of the Loan(s). Each Loan will be governed by the terms and conditions contained in this Agreement and in the Promissory Note relating to that Loan.

- **2.2 Availability**. Loans will be made available on any day on which Bank and the Federal Reserve Banks are open for business (a "**Business Day**") upon the telephonic or written request of an authorized employee of the Company. Requests for loans must be received by 12:00 p.m. Denver, Colorado time on the date the Loan is desired. Loans will be made available by wire transfer of immediately available funds. Wire transfers will be made to such account or accounts as may be authorized by the Company.
- **2.3 Security**. The Company's obligations under this Agreement, and each Promissory Note will be secured by a statutory first lien on all equity that the Company may now own or hereafter acquire or be allocated in Bank. In addition, except as otherwise provided in a Promissory Note, the Company's obligations hereunder and under each Promissory Note will be:
- (a) secured pursuant to the Security Agreements by a first priority lien on all the Collateral in favor of the Collateral Agent, for the benefit of the Bank and the other Secured Parties; and
 - **(b)** guaranteed by the Subsidiary Guaranties.
- 2.4 Payments Generally. The Company's obligation to repay each Loan will be evidenced by a Promissory Note. Bank will maintain a record of all Loans, the interest accrued thereon, and all payments made with respect thereto, and such record will, absent proof of manifest error, be conclusive evidence of the outstanding principal and interest on the Loans. Payments under each Promissory Note will be made by wire transfer of immediately available funds, by check, or by automated clearing house (ACH) or other similar cash handling processes as specified by separate agreement between the Company and Bank. Wire transfers will be made to ABA No. 307088754 for advice to and credit of "CoBANK" (or to such other account as Bank may direct by notice). The Company will give Bank telephonic notice no later than 12:00 p.m. Denver, Colorado time on the day the Company intends to pay by wire of such intent, and funds received after 3:00 p.m. Denver, Colorado time will be credited on the next Business Day. Checks will be mailed to CoBANK, Department 167, Denver, Colorado 80291-0167 (or to such other place as Bank may direct by notice). Credit for payment by check will not be given until the later of the next Business Day after receipt of the check or the day on which Bank receives immediately available funds. If any installment of principal or interest is due on a date that is not a Business Day, then such installment will be due and payable on the next Business Day.
- 2.5 Broken Funding Surcharge. Notwithstanding the terms of any Promissory Note giving the Company the right to repay any Loan prior to the date it would otherwise be due and payable, the Company agrees to provide three Business Days' prior written notice for any prepayment of a Loan and to pay to Bank a broken funding surcharge in the amount equal to the greater of (a) any funding losses sustained by Bank as a result of such prepayment, which funding losses shall be determined as the present value of the excess, if any, of the contractual interest that would have accrued on the amount prepaid at the applicable fixed interest rate on the applicable Promissory Note over the interest that would accrue on the same amount at the Reinvestment Rate for the remaining scheduled term of the amount of the prepayment, or (b) \$300. For purposes of determining such funding losses on the prepaid amount pursuant to this Section 2.5, "Reinvestment Rate" means the yield to maturity (on a bond-equivalent basis) of

- U.S. Treasury obligations with a maturity (or interpolated maturity, determined on a straight-line basis) comparable to such remaining term, as published for the most recent Business Day available on the Federal Reserve H.15 statistical release (or any successor publication), plus 0.50% (50 basis points). If the remaining term does not correspond exactly to a published maturity, the Reinvestment Rate shall be determined by linear interpolation between the nearest shorter and longer published maturities. If the relevant published yields are no longer available, the Reinvestment Rate shall be determined by reference to a reasonably comparable successor source customarily used in the market for such purpose, as selected by Bank in good faith and in a commercially reasonable manner. In addition, for purposes of this Section 2.5, present values shall be computed using the Reinvestment Rate with cash flow timing consistent with the applicable original interest payment frequency, on an actual/360 day-count basis, and with discounting to the prepayment date. Notwithstanding the foregoing, in the event of a conflict between the provisions of this section and of the broken funding charge section of a forward fix agreement between Bank and the Company, the provisions of the forward fix agreement will control.
- 2.6 Taxes; Change in Law. Any payment by the Company to Bank will be made net of any taxes (other than income and similar taxes imposed on or measured by Bank's overall net income, franchise taxes and branch profits taxes). If any change after the date hereof in any applicable law, rule, regulation, code, ordinance, order or the like (collectively, "Laws"), including, without limitation, all laws relating to taxes (other than income and similar taxes imposed on or measured by Bank's overall net income, franchise taxes and branch profits taxes), increases the cost of making or maintaining any Loan (or any associated commitment to lend), or reduces the amount received or receivable by Bank hereunder or under any other Loan Document, or requires or modifies any reserve or similar requirements of the Bank, then, upon request, the Company will pay to Bank such additional amount as will compensate Bank for such additional costs incurred or reduction suffered.

ARTICLE 3 Conditions Precedent.

- **3.1** Conditions to Initial Promissory Note. Bank's obligation to extend credit under the initial Promissory Note hereunder is subject to the condition precedent that Bank receive, in form and substance satisfactory to Bank, a duly executed copy of this Agreement, the other Loan Documents, and all instruments and documents contemplated hereby and thereby.
- **3.2** Conditions to Each Promissory Note. Bank's obligation to extend credit under each Promissory Note hereunder, including the initial Promissory Note, is subject to the condition precedent that Bank receive, in form and substance satisfactory to Bank, each of the following, except as otherwise provided in the Promissory Note:
- (a) **Promissory Note**. A duly executed copy of the Promissory Note and all instruments and documents contemplated by the Promissory Note.
- **(b) Evidence of Perfection and Priority**. Such evidence as Bank may require that the Collateral Agent, in favor of Bank and the other Secured Parties, has duly perfected first priority liens as required under this Agreement.

- (c) Evidence of Authority. Such certified board resolutions, certificates of incumbency, and other evidence that Bank may require that the Promissory Note, all instruments and documents executed in connection therewith, and, in the case of the initial Promissory Note hereto, this Agreement, the other Loan Documents and all instruments and documents executed in connection herewith and therewith, including any Security Documents, have been duly authorized and executed.
- (d) Fees and Other Charges. Any fees or other charges provided for herein, in the Promissory Note or in any invoice provided by Bank on or before the making of the extension of credit.
- **(e) Insurance.** Such evidence as Bank may require that the Company is in compliance with Section 7.2 below.
- **(f) Consents and Approvals**. Evidence as Bank may require that all regulatory and other consents and approvals referred to in Section 4.7 below have been obtained and are in full force and effect.
- **(g) Opinion of Counsel**. An opinion of counsel to the Company (which counsel must be acceptable to Bank).
- **3.3** Conditions to Each Loan. Bank's obligation under each Promissory Note to make any Loan to the Company thereunder is subject to the condition that no Event of Default or Default will have occurred and be continuing or would be caused by the making of such Loan.
- **ARTICLE 4 Representations and Warranties**. The execution by the Company of this Agreement and each Promissory Note hereunder, or any renewal or extension by Bank of any Promissory Note hereunder, will constitute a representation and warranty by the Company that:
- 4.1 Organization; Authority. The Company is a corporation duly organized, validly existing and in good standing under the Laws of Delaware and is duly qualified as a foreign corporation and in good standing in each other jurisdiction in which qualification is required by Law, other than any jurisdictions as to which the failure to be so qualified or in good standing could not individually or in the aggregate reasonably be expected to have a Material Adverse Effect. The Company has the corporate power and authority to own or lease the Properties it purports to own or hold under lease, to transact the business it transacts and proposes to transact, to execute and deliver this Agreement and the other Loan Documents to which it is a party and to perform the provisions thereof.
- **4.2 Authorization**. This Agreement and the other Loan Documents to which it is a party have been duly authorized by all necessary action on the part of the Company, and constitute the legal, valid and binding Obligations of the Company enforceable against the Company in accordance with their terms, except as such enforceability may be limited by: (a) bankruptcy, insolvency, reorganization, moratorium, receivership, conservatorship, arrangement or similar Laws affecting the enforcement of creditors' rights generally; and (b) general

principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at Law).

4.3 Disclosure. This Agreement, the financial statements delivered to the Bank by or on behalf of the Company and any other documents, certificates and writings delivered to the Bank by or on behalf of the Company in connection with any Loan (collectively "Disclosure Documents"), taken as a whole, do not contain any untrue statement of material fact or omit to state any material fact necessary to make any statements therein not misleading in light of the circumstances under which they were made. Except as may be disclosed in the Disclosure Documents, since December 31, 2024, there has been no change in the financial condition, operations, business, properties or prospects of the Company Group that could, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect. There is no fact known to the Company that could reasonably be expected to have a Material Adverse Effect that is not described herein or in the Disclosure Documents.

4.4 Subsidiaries; Affiliates.

- (a) Schedule 4.4 contains (except as may be noted therein) complete lists of the: (i) Subsidiaries, showing as to each its name, jurisdiction of organization, and percentage of Equity Interests outstanding owned directly or indirectly by the Company Group; (ii) Company's other Affiliates, if any; and (iii) Company directors and senior officers.
- **(b)** All outstanding shares of capital stock, membership interests or other Equity Interests of each Subsidiary owned by the Company Group have been validly issued, are fully paid and non-assessable and are owned by the Company or a Subsidiary free and clear of any Lien prohibited by Section 8.5.
- (c) Each Subsidiary is a legal entity duly organized, validly existing and in good standing under the Laws of its jurisdiction of organization, and is duly qualified as a foreign corporation or other legal entity and, where applicable, is in good standing in each jurisdiction in which such qualification is required by Law, other than jurisdictions as to which the failure to be so qualified or in good standing could not individually or in the aggregate reasonably be expected to have a Material Adverse Effect. Each such Subsidiary has the corporate, company or other similar power and authority to own or lease the Properties it purports to own or lease, to transact the business it transacts and proposes to transact and to execute and deliver the Loan Documents to which it is a party and to perform the provisions thereof.
- (d) No Subsidiary is subject to any legal, contractual or other restriction (other than this Agreement, the Note Purchase Agreement, the Northern Trust Loan Agreement and the agreements listed on Schedule 4.4(D) and customary limitations imposed by corporate Law or similar statutes) restricting the ability of such Subsidiary to pay dividends or make similar distributions of profits to any of the Company Group that owns outstanding Equity Interests of such Subsidiary.
- (e) The Loan Documents to which it is a party have been duly authorized by all necessary action on the part of each Subsidiary, and constitute the legal, valid and binding Obligations of such Subsidiary enforceable against such Subsidiary in accordance with their terms, except as such enforceability may be limited by: (a) bankruptcy, insolvency, reorganization, moratorium, receivership, conservatorship, arrangement or similar Laws affecting the enforcement of creditors' rights generally; and (b) general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at Law).

- 4.5 Financial Statements; Liabilities. All financial statements that have been delivered to the Bank by or on behalf of the Company (including any related schedules and notes) fairly present in all material respects the consolidated financial position of the Company Group as of the respective dates specified and the consolidated results of their operations and cash flows for the respective periods specified and have been prepared in accordance with GAAP consistently applied throughout the periods involved except as set forth in any notes thereto (subject in the case of interim statements to normal year-end adjustments and the absence of footnotes). The Company Group does not have any Material liabilities not disclosed in the Disclosure Documents.
- 4.6 Laws; Other Instruments. The execution, delivery and performance by the Company of this Agreement and the other Loan Documents to which it is a party do not: (a) contravene, result in any breach of, or constitute a default under, or result in the creation of any Lien in respect of any Property of the Company Group under, any (i) indenture, mortgage, deed of trust, loan, purchase or credit agreement, or lease, or (ii) corporate or company charter or bylaws, shareholder or operating agreement, or similar organizational agreement or instrument, in each case of this clause (a) to or by which the Company or any Subsidiary or any of their Properties are bound or affected; (b) conflict with or result in a breach of any order, judgment, decree or ruling of any arbitrator or Governmental Authority to which any of the Company Group is subject; or (c) violate any Law applicable to any of the Company Group.
- **4.7 Governmental Authorizations**. Except for any consents, approvals, authorizations, registrations, filings or declarations that have been obtained or made, no consent, approval or authorization of, or registration, filing or declaration with, the Arizona Corporation Commission or any other Governmental Authority is required in connection with the execution, delivery or performance by the Company or any Subsidiary Guarantor of this Agreement and the other Loan Documents.

4.8 Litigation.

- (a) There are no actions, suits, investigations or proceedings pending or, to the best knowledge of the Company, threatened against or affecting any of the Company Group or any Property of any of the Company Group in any court or before any arbitrator or before or by any Governmental Authority that could, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.
- **(b)** Neither the Company nor any Subsidiary is in: (i) default under any agreement or instrument to which it is a party or by which it is bound; (ii) violation of any order, judgment, decree or ruling of any arbitrator or Governmental Authority to which any of the Company Group is subject; or (iii) violation of any applicable Law of any Governmental Authority (including any Environmental Law or Law referred to in Section 4.15), which default or violation could, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.
- **4.9 Taxes**. The Company and its Subsidiaries have filed all tax returns required to have been filed in any jurisdiction, and have paid all taxes shown to be due and payable on such returns and all other taxes and assessments levied on them or their Properties, income or franchises, to the extent such taxes and assessments have become due and payable and before

they have become delinquent, except for any taxes and assessments: (a) the amount of which, individually or in the aggregate, is not Material; or (b) the amount, applicability or validity of which is being contested in Good Faith by appropriate proceedings and with respect to which the Company Group has established adequate reserves in accordance with GAAP. The Company knows of no basis for any other tax or assessment that could, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect. The charges, accruals and reserves on the books of the Company Group in respect of U.S. federal, state or other taxes for all fiscal periods are adequate. The U.S. federal income tax liabilities of the Company Group have been finally determined (whether by reason of completed audits or the statute of limitations having run) for all Fiscal Years up to and including the Fiscal Year ended December 31, 2024.

4.10 Property; Leases. The Company and its Subsidiaries have good and sufficient title to their respective owned Properties that individually or in the aggregate are Material, including all such Properties reflected in the most recent audited balance sheet referred to in Section 4.5 or purported to have been acquired by the Company Group after the date thereof (except as disposed of in the ordinary course of business), in each case free and clear of Liens prohibited by Section 8.5. All leases that individually or in the aggregate are Material are valid, subsisting and are in full force and effect in all material respects.

4.11 Licenses; Permits.

- (a) The Company and its Subsidiaries own or possess all licenses, permits, franchises, authorizations, patents, copyrights, proprietary software, service marks, trademarks and trade names, or rights thereto, that individually or in the aggregate are Material, without known conflict with the rights of others.
- **(b)** To the best knowledge of the Company, no product or service of the Company or any Subsidiary infringes in any Material respect any license, permit, franchise, authorization, patent, copyright, proprietary software, service mark, trademark, trade name or other right owned by any other Person.
- (c) To the best knowledge of the Company, there is no Material violation by any Person of any right of the Company or any Subsidiary with respect to any patent, copyright, proprietary software, service mark, trademark, trade name or other right owned or used by any of the Company Group.

4.12 ERISA.

(a) The Company and each ERISA Affiliate have operated and administered each Plan in compliance with all applicable Laws except for such instances of noncompliance as have not resulted in and could not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect. Neither the Company nor any ERISA Affiliate has incurred any liability pursuant to Title I or IV of ERISA or the penalty or excise tax provisions of the Code relating to employee benefit plans (as defined in section 3 of ERISA), and no event, transaction or condition has occurred or exists that could, individually or in the aggregate, reasonably be expected to result in the incurrence of any such liability by the Company or any ERISA Affiliate, or in the imposition of any Lien on any of the rights or Properties of the Company or any ERISA Affiliate, in either case pursuant to Title I or IV of ERISA or to section 430(k) of the Code or to any such penalty or excise tax provisions under the Code or federal Law or section 4068 of

ERISA or by the granting of a security interest in connection with the amendment of a Plan, other than such liabilities or Liens as would not be individually or in the aggregate Material.

- **(b)** Neither the Company nor any ERISA Affiliate has ever maintained or contributed to a Plan subject to Title IV of ERISA.
- (c) The Company and its ERISA Affiliates have not incurred withdrawal liabilities (and are not subject to contingent withdrawal liabilities) under section 4201 or 4204 of ERISA in respect of Multiemployer Plans that individually or in the aggregate are Material.
- (d) The expected postretirement benefit obligation (determined as of the last day of the Company's most recently ended Fiscal Year in accordance with Financial Accounting Standards Board Accounting Standards Codification Topic 715-60, without regard to liabilities attributable to continuation coverage mandated by section 4980B of the Code) of the Company Group is not Material.
- (e) The execution and delivery of this Agreement and the issuance of the Promissory Note will not involve any transaction subject to the prohibitions of section 406 of ERISA or in connection with which a tax could be imposed pursuant to section 4975(c)(1)(A)-(D) of the Code.
- 4.13 Margin Regulations. No proceeds of any Loan will be used, directly or indirectly, for the purpose of buying or carrying any margin stock within the meaning of Regulation U of the Federal Reserve System (12 CFR 221), buying, carrying or trading in any Securities under circumstances as to involve the Company in a violation of Regulation X of the Federal Reserve System (12 CFR 224), or to involve any broker or dealer in a violation of Regulation T of the Federal Reserve System (12 CFR 220). Margin stock does not constitute more than 5.00% of the value of the Consolidated Assets, and the Company Group has no present intention that margin stock will constitute more than 5.00% of the value of the Consolidated Assets. In this Section 4.13, "margin stock" and "purpose of buying or carrying" have the meanings assigned in Regulation U.

4.14 Existing Indebtedness; Future Liens.

- (a) Schedule 4.14(A) sets forth a list of all outstanding Indebtedness for borrowed money of the Company Group at September 30, 2025, since which date there has been no Material change in the amounts, interest rates, sinking funds, installment payments or maturities of such Indebtedness. Neither the Company nor any Subsidiary is in default and no waiver of default is currently in effect, in the payment of any Indebtedness for borrowed money of the Company or such Subsidiary, and no event or condition exists with respect to any Indebtedness for borrowed money of the Company or any Subsidiary that would permit (or with notice or the lapse of time or both would permit) one or more Persons to cause such Indebtedness for borrowed money to be payable before its stated maturity or regularly scheduled dates of payment.
- **(b)** Neither the Company nor any Subsidiary has agreed or consented to cause or permit any of its Property, now owned or hereafter acquired, to be subject to a Lien that secures Indebtedness or to cause or permit in the future (on the happening of a contingency or otherwise) any of its Property, now owned or hereafter acquired, to be subject to a Lien that secures Indebtedness prohibited by this Agreement.

(c) Neither the Company nor any Subsidiary is a party to, or otherwise subject to any provision in, any instrument evidencing Indebtedness for borrowed money, any agreement relating thereto or any other agreement (including its charter or any other organizational document) which limits the amount of, or otherwise imposes restrictions on the incurring of, Indebtedness of the Company, except this Agreement, the Note Purchase Agreement, the Northern Trust Loan Agreement and agreements governing other Material Credit Facilities.

4.15 Corrupt Practices.

- (a) Neither the Company nor any Controlled Entity is: (i) a Prohibited Person; (ii) an agent, department or instrumentality of, or otherwise beneficially owned by, controlled by or acting on behalf of, directly or indirectly (x) any Prohibited Person, or (y) any Person, entity, organization, foreign country or regime subject to any OFAC Sanctions Program; or (iii) otherwise blocked, subject to sanctions under or engaged in any activity in violation of other United States economic sanctions, including the Trading with the Enemy Act, International Emergency Economic Powers Act, Comprehensive Iran Sanctions, Accountability and Divestment Act or any similar Law with respect to Iran or any other country, Sudan Accountability and Divestment Act, any OFAC Sanctions Program, or any economic sanctions regulations administered and enforced by the United States or any enabling legislation or executive order relating to any of the foregoing ("U.S. Economic Sanctions") (each Prohibited Person and other Person, entity, organization and government of a country described in clauses (i), (ii) or (iii), a "Blocked Person"). Neither the Company nor any Controlled Entity has been notified that its name appears or may in the future appear on a state list of Persons that engage in investment or other commercial activities in any country subject to U.S. Economic Sanctions.
- **(b)** No proceeds of any Loan constitute or will constitute funds obtained on behalf of any Blocked Person or will otherwise be used by the Company or any Controlled Entity, directly or indirectly: (i) in connection with any investment in, or any transaction or dealing with, any Blocked Person; or (ii) otherwise in violation of U.S. Economic Sanctions.
- (c) Neither the Company nor any Controlled Entity: (i) has been found in violation of, charged with or convicted of money laundering, drug trafficking, terrorist related activities or other money laundering predicate crimes under the Currency and Foreign Transactions Reporting Act of 1970 (otherwise known as the Bank Secrecy Act), Executive Order No. 13224, the USA PATRIOT Act or any other United States or Canadian law or regulation governing such activities ("Anti-Terrorism Laws"), or any U.S. Economic Sanctions violations; (ii) to the Company's actual knowledge after due inquiry, is under investigation by any Governmental Authority for possible violation of Anti-Terrorism Laws or U.S. Economic Sanctions; (iii) has been assessed civil penalties under any Anti-Terrorism Laws or U.S. Economic Sanctions; or (iv) has had any of its funds seized or forfeited in an action under any Anti-Terrorism Law. The Company has established procedures and controls it reasonably believes are adequate (and otherwise comply with applicable Law) to ensure the Company and each Controlled Entity is and will continue to be in compliance with all applicable current and future Anti-Terrorism Laws and U.S. Economic Sanctions.
- (d) Neither the Company nor any Controlled Entity: (i) has been charged with, or convicted of bribery or any other anti-corruption related activity under any applicable Law in a U.S. or any non-U.S. country or jurisdiction, including the U.S. Foreign Corrupt Practices Act and U.K. Bribery Act 2010 (collectively, "Anti-Corruption Laws"); (ii) to the Company's actual knowledge after due inquiry, is under investigation by any Governmental Authority for possible violation of Anti-Corruption Laws; (iii) has been assessed civil or criminal penalties under any Anti-Corruption Laws; or (iv) has been or is the target of sanctions imposed by the United Nations or European Union;

- (e) To the Company's actual knowledge after due inquiry, neither the Company nor any Controlled Entity has, within the last five years, directly or indirectly offered, promised, given, paid or authorized the offer, promise, giving or payment of anything of value to a Governmental Official or commercial counterparty for the purposes of: (x) influencing any act, decision or failure to act by such Governmental Official in his or her official capacity or for such commercial counterparty; (y) inducing a Governmental Official to do or omit to do any act in violation of the Governmental Official's lawful duty; or (z) inducing a Governmental Official or commercial counterparty to use his or her influence with a government or instrumentality to affect any act or decision of such government or entity; in each case in order to obtain, retain or direct business or to otherwise secure an improper advantage in violation of any Law or which would cause the Bank to be in violation of any Law applicable to the Bank; and
- (f) No part of the proceeds of any Loan will be used, directly or indirectly, for any improper payments, including bribes, to any Governmental Official or commercial counterparty in order to obtain, retain or direct business or obtain any improper advantage. The Company has established procedures and controls it reasonably believes are adequate (and otherwise comply with Law) to ensure the Company and each Controlled Entity is and will be in compliance with all current and future Anti-Corruption Laws.
- **4.16 Status under Certain Statutes**. Neither the Company nor any Subsidiary is subject to regulation under the Investment Company Act of 1940, Public Utility Holding Company Act of 2005, ICC Termination Act of 1995 or Federal Power Act, all as amended.

4.17 Environmental Matters.

- (a) Neither the Company nor any Subsidiary has knowledge or has received notice of any claim and no proceeding has been instituted asserting any claim against the Company or any of its Subsidiaries or any of their Properties now or formerly owned, leased or operated by any of them, alleging any damage to the environment or violation of any Environmental Laws, except, in each case, such as could not reasonably be expected to result in a Material Adverse Effect.
- **(b)** Neither the Company nor any Subsidiary has knowledge of any facts which would give rise to any claim, public or private, of violation of Environmental Laws or damage to the environment emanating from, occurring on or in any way related to Properties now or formerly owned, leased or operated by any of them or to other assets or their use, except, in each case, such as could not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect.
- (c) Neither the Company nor any Subsidiary has stored any Hazardous Materials on Properties now or formerly owned, leased or operated by any of them in a manner which is contrary to any Environmental Law that could, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect.
- (d) Neither the Company nor any Subsidiary has disposed of any Hazardous Materials in a manner which is contrary to any Environmental Law that could, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect.
- **(e)** All buildings on real property now owned, leased or operated by the Company or any Subsidiary are in compliance with applicable Environmental Laws, except where failure to comply could not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect.

4.18 Collateral. The Security Documents are effective to grant to the Collateral Agent for the ratable benefit of the Secured Parties legal, valid and enforceable first priority Liens on all Collateral. The security interests in the Collateral granted to the Collateral Agent (for the benefit of the Secured Parties) pursuant to the Security Documents constitute, as to all such Collateral, legal, valid and enforceable security interests and Liens under the UCC, subject to no other Lien; and all actions as are required pursuant to the Security Documents or applicable Law have been taken to establish and perfect the Collateral Agent's rights in and to, and the first priority of its Liens and security interests on all of the Collateral to secure the Obligation, including any recording, filing, registration, possession, control, delivery to the Collateral Agent, giving of notice or similar action.

ARTICLE 5 Company Information.

5.1 Financial and Business. The Company shall deliver to the Bank:

- (a) Quarterly Statements Within 60 days (or such earlier date by which such statements are required, filed or delivered under the Note Purchase Agreement or SEC requirement) after the end of each Fiscal Quarter ending on or after March 31, 2026 (other than the last Fiscal Quarter of each Fiscal Year): (i) a consolidated balance sheet of the Company Group as at the end of such Fiscal Quarter, and (ii) consolidated statements of income and cash flows of the Company Group for such Fiscal Quarter and (in the case of the second and third Fiscal Quarters of any Fiscal Year) for the portion of the Fiscal Year then ended; setting forth in comparative form the figures for the corresponding periods in the previous Fiscal Year, all in reasonable detail, prepared in accordance with GAAP applicable to quarterly financial statements generally, and certified by a Senior Financial Officer as fairly presenting, in all Material respects, the financial position of the Company Group and its results of operations and cash flows, subject to changes resulting from year-end adjustments and the absence of footnotes;
- (b) Annual Statements Within 120 days (or such earlier date by which such statements are required, filed or delivered under the Note Purchase Agreement or SEC requirement) after the end of the 2025 and each subsequent Fiscal Year: (i) a consolidated balance sheet of the Company Group as at the end of such Fiscal Year, and (ii) consolidated statements of income, changes in shareholders' equity and cash flows of the Company Group for such Fiscal Year, setting forth in comparative form the figures for the previous Fiscal Year, all in reasonable detail, prepared in accordance with GAAP, and accompanied by an opinion thereon (without a "going concern" or similar qualification or exception and without qualification or exception as to the scope of the audit on which such opinion is based) of independent public accountants of recognized national standing, which opinion shall state that such financial statements present fairly, in all material respects, the financial position of the Company Group and its results of operations and cash flows and have been prepared in conformity with GAAP, that the examination of such accountants in connection with such statements has been made in accordance with generally accepted auditing standards, and that such audit provides a reasonable basis for such opinion in the circumstances;
- (c) Default or Event of Default Promptly, and in any event within five Business Days after a Responsible Officer becoming aware of the existence of any Default or Event of Default or that any Person has given notice or taken action with respect to a claimed default hereunder or that any Person has given notice or taken action with respect to a claimed default of the type referred to in Section 9.1(f), a written notice specifying the nature and period of existence thereof and what action the Company is taking or proposes to take with respect thereto;

11

- ERISA Promptly, and in any event within five Business Days after a Responsible Officer becoming aware of any of the following, notice setting forth the nature thereof and the action, if any, the Company or an ERISA Affiliate proposes to take with respect thereto: (i) with respect to any Plan, any reportable event, as defined in section 4043(c) of ERISA and the regulations thereunder, for which notice thereof has not been waived pursuant to such regulations as in effect on the date hereof; (ii) the taking by the PBGC of steps to institute, or the threatening by the PBGC of the institution of, proceedings under section 4042 of ERISA for the termination of, or the appointment of a trustee to administer, any Plan, or the receipt by the Company or any ERISA Affiliate of a notice from a Multiemployer Plan that such action has been taken by the PBGC with respect to such Multiemployer Plan; or (iii) any event, transaction or condition that could result in the incurrence of any liability by the Company or any ERISA Affiliate pursuant to Title I or IV of ERISA or the penalty or excise tax provisions of the Code relating to employee benefit plans, or in the imposition of any Lien on any of the rights, properties or assets of the Company or any ERISA Affiliate pursuant to Title I or IV of ERISA or such penalty or excise tax provisions, if such liability or Lien, taken together with any other such liabilities or Liens then existing, could reasonably be expected to have a Material Adverse Effect;
- **(e)** Governmental Authority Promptly, and in any event within 30 days of receipt thereof, copies of any written notice to the Company or any Subsidiary from any Governmental Authority relating to any order, ruling, statute or other law or regulation that could reasonably be expected to have a Material Adverse Effect; and
- (f) Requested Information With reasonable promptness, such other information (which may include tax returns) relating to the business, operations, affairs, financial condition or Properties of the Company Group or its ability to perform its Obligations as from time to time may be reasonably requested by the Bank.
- **5.2 Officer's Certificate**. Each financial statement delivered to the Bank pursuant to Section 5.1(a) or 5.1(b) shall be accompanied by a certificate of a Senior Financial Officer:
- (a) Covenant Compliance Setting forth the information from such financial statements that is required in order to establish whether the Company was in compliance with the requirements of Sections 8.5(m), 8.6, 8.7 and 8.8(g) during the quarterly or annual period covered by the statements then being furnished (including with respect to each provision that involves mathematical calculations the information from such statements that is required to perform such calculations), and detailed calculations of the maximum or minimum amount, ratio or percentage, as the case may be, permissible under such Section, and of the amount, ratio or percentage then in existence. If the Company or any Subsidiary has made an election to measure any financial liability using fair value (which election is being disregarded for purposes of determining compliance with this Agreement pursuant to Section 1) as to the period covered by any such statement, such Senior Financial Officer's certificate as to such period shall include a reconciliation from GAAP with respect to such election; and
- **(b)** Event of Default Certifying such Senior Financial Officer has reviewed the terms hereof and has made, or caused to be made, under his or her supervision, a review of the transactions and conditions of the Company Group from the beginning of the quarterly or annual period covered by the statements then being furnished to the date of the certificate and such review has not disclosed the existence during such period of any Default or Event of Default or, if any such Default or Event of Default existed or exists (including any resulting from the failure to comply with any Environmental Law), specifying its nature and period of existence and what action the Company has taken or proposes to take with respect thereto.

- **5.3 Visitation**. The Company shall permit representatives of the Bank:
- (a) No Default If no Default or Event of Default then exists, at the expense of the Bank and with prior notice to the Company, to visit the principal executive office of the Company during normal business hours, discuss the affairs, and finances of the Company Group with the Company's officers, and (with the consent of the Company which shall not be unreasonably withheld or delayed) with its independent public accountants (with the understanding only one discussion with those accountants each Fiscal Year will be permitted pursuant to this Section 5.3(a)), and (with the consent of the Company which shall not be unreasonably withheld or delayed) to visit the other offices and Properties of the Company Group as often as may be reasonably requested; and
- **(b)** Default If a Default or Event of Default exists, at the expense of the Company to visit and inspect any offices or Properties of the Company Group to examine its records, reports and other papers, make copies and extracts therefrom, and to discuss their respective affairs, finances and accounts with their respective officers and independent public accountants (and by this provision the Company authorizes said accountants to discuss the affairs, finances and accounts of the Company Group), all at such times and as often as may be requested.
- **5.4 Electronic Delivery**. Financial statements, opinions of independent certified public accountants, other information and Officer's Certificates that are required to be delivered by the Company pursuant to any of Sections 5.1(a) or (b) and Section 5.2 shall be deemed to have been delivered if the Company satisfies any of the following requirements with respect thereto:
 - (i) such financial statements satisfying the requirements of Section 5.1(a) or (b) and related Officer's Certificate satisfying the requirements of Section 5.2 are delivered to the Bank by e-mail; or
 - (ii) the Company shall have timely filed Form 10–Q or Form 10–K, satisfying the requirements of Section 5.1(a) or Section 5.1(b), as the case may be, with the SEC on EDGAR and shall have made such form and the related Officer's Certificate satisfying the requirements of Section 5.2 available on its home page on the internet, which on the date hereof is located at http://www.gwresources.com as of the date required in such Section 5.1(a) or Section 5.1(b), as the case may be.

provided however, that in the case of clause (ii), the Company shall have given the Bank prior notice, which may be by e-mail or in accordance with Section 11.2, of such posting or filing in connection with each delivery, provided further, that upon request of the Bank to receive paper copies of such forms, financial statements and Officer's Certificates or to receive them by e-mail, the Company will promptly e-mail them or deliver such paper copies, as the case may be, to the Bank.

5.5 Indemnity for Taxes. At all times indemnify and hold and save the Bank harmless from and against any and all actions or causes of action, claims, demands, liabilities, loss, damage or expense of whatsoever kind and nature incurred by the Bank as a result of the non-payment of any documentary stamp tax, intangible tax, interest or penalties associated therewith or any other local, state or federal assessment required to be paid, but not paid in conjunction with the indebtedness evidenced by the Loan Documents. The Company agrees to pay to the Bank, its successors and assigns, all sums of money requested by the Bank hereunder

within ten days of such request, which the Bank will or may advance, pay or cause to be paid, or become liable to pay, on account of or in connection with failure to pay as required by the regulations of the Governmental Authority so imposing said payment. The Bank will be entitled to charge for any and all disbursements made by it in Good Faith, under the reasonable belief that it or the Company is or was liable for the amount so assessed. Any default by the Company in making any payments required under this covenant will constitute a payment Event of Default under the Loan Documents and Bank may, at its option, declare the entire amount of principal plus accrued interest thereon due and payable without notice or demand.

ARTICLE 6 Prepayments. Unless otherwise agreed to in writing by Bank, while this Agreement is in effect, the Company agrees that:

6.1 Reserved.

6.2 Prepayment; Change in Control.

- (c) Notice. The Company shall, within five Business Days after any Responsible Officer has knowledge of the occurrence of any Change in Control, give notice of such Change in Control to the Bank. Such notice shall contain and constitute an offer to prepay the Obligation and simultaneously terminate any remaining commitment to make Loans and be accompanied by the certificate described in Section 6.2(e).
- (d) Offer to Prepay. The offer contemplated by Section 6.2(a) shall be an offer to prepay, in accordance with this Section 6.2, all but not less than all of the Obligation and terminate any remaining commitment to make Loans on a date specified in such offer (the "Proposed Prepayment Date") which shall be a Business Day not fewer than ten Business Days or more than 45 days after the Change in Control.
- **(e)** Acceptance/Rejection. The Bank may accept or reject the offer made pursuant to this Section 6.2 by notice to the Company not later than ten Business Days after receipt of the offer. Failure by the Bank to respond to an offer made pursuant to this Section 6.2 will constitute an acceptance of such offer.
- (f) Prepayment. Prepayment of the Loans pursuant to this Section 6.2 shall include interest accrued to the date of prepayment, any unreimbursed transaction expenses and any amounts due under Section 2.5. The prepayment shall be made on the Proposed Prepayment Date.
- **(g)** Officer's Certificate. Each offer pursuant to this Section 6.2 shall be accompanied by a certificate executed by a Senior Financial Officer and dated the date of such offer, specifying: (i) the Proposed Prepayment Date; (ii) that such offer is made pursuant to this Section 6.2; (iii) the principal amount offered to be prepaid; (iv) the interest that would accrue on the Proposed Prepayment Date; (v) that the conditions of this Section 6.2 required to be performed as of the date thereof have been performed or fulfilled; and (vi) in reasonable detail, the nature and date or proposed date of the Change in Control.
- **6.3 Asset Dispositions**. If the Company elects to make a Debt Prepayment Application, the Company shall offer to prepay the Loans in a principal amount which equals the Ratable Portion (as defined below) for the Loans (which offer shall be in writing and shall offer to make such prepayment on a Business Day which is not less than 10 and not more than 60 days

after the date of the notice of offer (the "**Disposition Prepayment Date**")), together with accrued interest thereon to the date of such prepayment. The Bank shall notify the Company of its acceptance or rejection of such offer within 10 Business Days of receipt thereof by giving notice of such acceptance or rejection to the Company; provided, however, that if the Bank fails to so notify the Company within 10 Business Days of receipt of the notice of offer of prepayment it shall be deemed to have rejected such offer. The Company shall prepay on the Disposition Prepayment Date the Ratable Portion of the Loans together with accrued interest thereon to the date of such prepayment.

For purposes of this Section 6.3, "Ratable Portion" for the Loans means, with respect to a Debt Prepayment Application, an amount equal to the product of (x) the Net Proceeds Amount being so applied to the payment of Senior Indebtedness multiplied by (y) a fraction the numerator of which is the outstanding principal amount of the Loans and the denominator of which is the aggregate principal amount of Senior Indebtedness of the Company and its Subsidiaries subject to such prepayment (or a required offer of such prepayment).

Notwithstanding any provision to the contrary, no additional amounts shall be due pursuant to Section 2.5 as a result of any prepayment pursuant to Sections 6.2 or 6.3.

ARTICLE 7 Affirmative Covenants. Unless otherwise agreed to in writing by Bank, while this Agreement is in effect, the Company agrees that:

- 1.1 Laws. Without limiting Section 8.4, the Company shall, and shall cause each of its Subsidiaries to, comply with all Laws to which each of them is subject, including ERISA, Environmental Laws, the USA PATRIOT Act and the other Laws referred to in Section 4.15, and obtain and maintain in effect all licenses, certificates, permits, franchises and other governmental authorizations necessary to the ownership of their respective Properties or conduct of their respective businesses, in each case to the extent necessary to ensure that non-compliance with such Laws or failures to obtain or maintain in effect such licenses, certificates, permits, franchises and other authorizations could not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.
- 1.2 Insurance. The Company will, and will cause each of its Subsidiaries to, maintain, with financially sound and reputable insurers, insurance with respect to their respective Properties and businesses against casualties and contingencies of such types, on such terms and in such amounts (including deductibles, co-insurance and self-insurance, if adequate reserves are maintained with respect thereto) as is customary in the case of entities of established reputations engaged in the Water Business or a similar business and similarly situated, other than any failure to maintain such insurance that would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.
- 1.3 Maintenance of Properties. The Company will, and will cause each of its Subsidiaries to, maintain and keep, or cause to be maintained and kept, their respective Properties in good repair, working order and condition (other than ordinary wear and tear), so that the business carried on in connection therewith may be properly conducted at all times, provided that this Section 7.3 shall not prevent the Company or any Subsidiary from

discontinuing the operation and the maintenance of any of its Properties if such discontinuance is desirable in the conduct of its business and the Company has concluded that such discontinuance could not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.

- 1.4 Taxes and Claims. The Company will, and will cause each of its Subsidiaries to, file all tax returns required to be filed in any jurisdiction and to pay and discharge all taxes shown to be due and payable on such returns and all other taxes, assessments, governmental charges or levies imposed on them or any of their Properties, income or franchises, to the extent the same have become due and payable and before they become delinquent, and all claims for which sums have become due and payable that have or might become a Lien on Properties of the Company or any Subsidiary, provided that neither the Company nor any Subsidiary need pay any such tax, assessment, charge, levy or claim if: (a) the amount, applicability or validity thereof is contested by the Company or such Subsidiary on a timely basis in Good Faith and appropriate proceedings, and the Company or a Subsidiary has established adequate reserves therefor in accordance with GAAP on the books of the Company or such Subsidiary; or (b) the nonfiling or nonpayment, as the case may be, of all such taxes, assessments, charges, levies and claims could not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.
- 1.5 Corporate Existence. The Company will at all times preserve and keep its corporate existence in full force and effect. Subject to Sections 8.2 and 8.8, the Company will at all times preserve and keep in full force and effect the corporate, limited liability company or other applicable existence of each of its Subsidiaries (unless merged into the Company or a Wholly Owned Subsidiary) and all rights and franchises of the Company and its Subsidiaries unless, in the Good Faith judgment of the Company, the termination of or failure to preserve and keep in full force and effect such corporate existence, right or franchise could not, individually or in the aggregate, have a Material Adverse Effect.
- 1.6 Books and Records. The Company will, and will cause each of its Subsidiaries to, maintain proper books of record and account in conformity with GAAP and in conformity in all material respects with all applicable requirements of any Governmental Authority having legal or regulatory jurisdiction over the Company or such Subsidiary, as the case may be. The Company will, and will cause each of its Subsidiaries to, keep books, records and accounts which, in reasonable detail, accurately reflect in all material respects all transactions and dispositions of assets. The Company and its Subsidiaries have devised a system of internal accounting controls sufficient to provide reasonable assurances that their respective books, records, and accounts accurately reflect all transactions and dispositions of assets and the Company will, and will cause each of its Subsidiaries to, continue to maintain such system.
- **1.7 Subsidiary Guarantors**. Subject to Article 8, the Company will cause each of its Subsidiaries (except for any Foreign Subsidiary or Foreign Holding Company) that guarantees or otherwise becomes liable at any time, whether as a borrower or additional or co-borrower or otherwise, for or in respect of any Indebtedness under the Term Note Facility or any other Material Credit Facility to concurrently therewith:

- (a) enter into an agreement in form and substance satisfactory to the Bank providing for the guaranty by such Subsidiary, on a joint and several basis with all other such Subsidiaries, of (i) the prompt payment in full when due of all amounts payable by the Company pursuant to the Promissory Notes (whether for principal, interest or otherwise) and this Agreement, including all indemnities, fees and expenses payable by the Company thereunder, and (ii) the prompt, full and faithful performance, observance and discharge by the Company of each and every covenant, agreement, undertaking and provision required pursuant to the Promissory Notes, this Agreement and the other Loan Documents to be performed, observed or discharged by it;
- (b) deliver the following to the Bank (i) an executed counterpart of such Subsidiary Guaranty, (ii) a certificate signed by an authorized officer of such Subsidiary containing representations and warranties on behalf of such Subsidiary to the same effect, mutatis mutandis, as those contained in Sections 4.1, 4.2, 4.6, 4.7, 4.10 and 4.11 of this Agreement (but with respect to such Subsidiary and such Subsidiary Guaranty rather than the Company), (iii) all documents reasonably requested by the Bank to evidence the due organization, continuing existence and good standing of such Subsidiary and the due authorization by all requisite action on the part of such Subsidiary of the execution and delivery of such Subsidiary Guaranty and the performance by such Subsidiary of its obligations thereunder, and (iv) an opinion of counsel reasonably satisfactory to the Bank covering such matters relating to such Subsidiary and such Subsidiary Guaranty as the Bank may reasonably request; and
- at the election of the Company and by notice to the Bank, any Subsidiary Guarantor may be discharged from its obligations under its Subsidiary Guaranty and be automatically released from its obligations thereunder without the need for the execution or delivery of any other document by the Bank, provided that (i) if such Subsidiary Guarantor is a guarantor or otherwise liable for or in respect of the Term Note Facility and/or any other Material Credit Facility, then such Subsidiary Guarantor shall have been released and discharged (or will be released and discharged concurrently with the release of such Subsidiary Guarantor under its Subsidiary Guaranty) from its liability for or in respect of all such Indebtedness, (ii) at the time of, and after giving effect to, such release and discharge, no Default or Event of Default shall be existing, (iii) no amount is then due and payable under such Subsidiary Guaranty, (iv) if in connection with such Subsidiary Guarantor being released and discharged from any Indebtedness described in the preceding clause (i) any fee, debt repayment or other consideration is given or paid to any holder of such Indebtedness, the Bank may elect to receive equivalent pro-rata consideration substantially concurrently therewith, and (v) the Company shall simultaneously provide to the Bank a certificate of a Responsible Officer certifying as to the matters set forth in the preceding clauses (i) through (iv). In the event of any such release, for purposes of Section 8.7, all Indebtedness of such Subsidiary shall be deemed to have been incurred concurrently with such release.
- 1.8 Collateral. The Company shall take all actions necessary to insure that the Collateral Agent, on behalf of the Secured Parties, has and continues to have in all relevant jurisdictions duly and validly created, attached, perfected and enforceable first priority Liens on all Collateral. The Company shall cause the Obligation to constitute direct senior secured obligations of the Company and to be senior in right of payment and to rank senior in right of security with respect to all Collateral granted in the Security Documents to all other Indebtedness of the Company; provided, however, if the Company incurs Indebtedness under Material Credit Facilities permitted hereunder (including the Term Note Facility) such Indebtedness may be secured equally and ratably with the Indebtedness evidenced by the Promissory Notes pursuant to documentation reasonably acceptable to the Bank in substance and form, including an

intercreditor agreement and opinions of counsel to the Company and/or any such Subsidiary, as the case may be.

- **ARTICLE 8** Negative Covenants. Unless otherwise agreed to in writing by Bank, while this Agreement is in effect, the Company agrees that:
- 8.1 Transactions with Affiliates. The Company will not and will not permit any Subsidiary to enter into directly or indirectly any transaction or group of related transactions (including the purchase, lease, sale or exchange of Property or rendering of any service) with any Affiliate (other than the Company or another Subsidiary), except in the ordinary course and pursuant to the reasonable requirements of the Company's or such Subsidiary's business and on fair and reasonable terms no less favorable to the Company or such Subsidiary than would be obtainable in a comparable arm's-length transaction with a Person not an Affiliate.
- **8.2** Merger; Consolidation. The Company will not, nor will it permit any Subsidiary Guarantor to, consolidate or merge with any other Person or Transfer all or substantially all of its assets in a single transaction or series of transactions to any Person unless:
- (a) the successor formed by such consolidation or survivor of such merger or Person acquiring such assets, as the case may be, shall be a solvent corporation, limited partnership or limited liability company organized and existing under the laws of the United States or state thereof (including the District of Columbia), and, if the Company or such Subsidiary Guarantor is not such a corporation, limited partnership or limited liability company (i) such corporation, limited partnership or limited liability company (if it involves the Company) shall have executed and delivered to the Bank its assumption of the due and punctual performance and observance of each covenant and agreement of the Company under the Loan Documents or (if it involves a Subsidiary Guarantor) shall have executed and delivered to the Bank its assumption of the due and punctual performance and observance of each covenant and condition of the Subsidiary Guaranty, and (ii) in either case, such corporation, limited partnership or limited liability company shall have caused to be delivered to the Bank an opinion of independent counsel reasonably satisfactory to the Bank to the effect that all documentation effecting such assumption is enforceable in accordance with its terms;
- **(b)** each Obligor under any Subsidiary Guaranty or Security Document outstanding when such transaction or each transaction in such a series of transactions occurs reaffirms its obligations under the Loan Documents and the Collateral Agent's Liens on the Collateral in writing at such time pursuant to documentation reasonably acceptable to the Bank; and
- (c) immediately before and immediately after giving effect to such transaction or each transaction in any such series of transactions, no Default or Event of Default shall have occurred and be continuing.

No such Transfer shall have the effect of releasing any Obligor or Collateral that shall theretofore have become such in the manner prescribed in this Section 8.2 from its liability under any Loan Document.

8.3 Business. The Company will not and will not permit any Subsidiary to engage in any business other than the Water Business.

8.4 Anti-Terrorism Law.

- (a) Bank hereby notifies the Company and any Related Party that, pursuant to the requirements of the USA Patriot Act, Bank may be required to obtain, verify and record information that identifies the Company and any Related Party, which information may include the name and address of the Company and any Related Party and other information that will allow Bank to identify the Company and any Related Party in accord with the USA Patriot Act. The Company hereby agrees to take any action necessary to enable Bank to comply with the requirements of the USA Patriot Act.
 - **(b)** The Company covenants, represents and warrants as follows:
 - (i) Neither the Company nor any Related Party is or, to the best of Company's knowledge, will be in violation of any Anti-Terrorism Law.
 - (ii) Neither the Company nor any Related Party is or, to the best of Company's knowledge, will be a Prohibited Person.
 - (iii) Neither the Company nor any Related Party: (A) conducts any business or engages in any transaction or dealing with any Prohibited Person, including making or receiving any contribution of funds, goods or services to or for the benefit of any Prohibited Person; (B) deals in, or otherwise engages in any transaction relating to, any property or interests in property blocked pursuant to Executive Order No. 13224; or (C) engages in or conspires to engage in any transaction that evades or avoids, or has the purpose of evading or avoiding, or attempts to violate, any of the prohibitions set forth in any Anti-Terrorism Law.
 - (iv) Neither the Company nor any Related Party will engage in any of the activities described in (iii) of this subsection (b) in the future.
 - (v) The Company and each Related Party will ensure that the proceeds of the Indebtedness are not used to violate any foreign asset control regulations of OFAC or of any enabling statute or any Executive Order relating thereto.
 - (vi) The Company will deliver to Bank any certification or other evidence requested from time to time by Bank in its sole reasonable discretion, confirming the Company's and any Related Party's compliance with this Section.
 - (vii) The Company has implemented procedures, and will consistently apply those procedures while this Agreement is in effect, to ensure that the representations and warranties in this Section remain true and correct while this Agreement is in effect.
- **8.5 Limitation on Liens**. The Company will not, and will not permit any of its Subsidiaries to, directly or indirectly create, incur, assume or permit to exist (on the happening of a contingency or otherwise) any Lien on or with respect to any Property (including any goods or accounts receivable) of the Company or any Subsidiary, now owned or held or hereafter created or acquired, or any income or profits therefrom, or assign or otherwise convey any right to receive income or profits, except:
- (a) Liens for taxes, assessments or other governmental charges not yet due and payable or the payment of which is not at the time required by Section 7.4;

- **(b)** any attachment or judgment Lien, unless the judgment it secures shall not, within 60 days after the entry thereof, have been discharged or execution thereof stayed pending appeal, or shall not have been discharged within 60 days after the expiration of any such stay;
- (c) Liens incidental to the conduct of business or the ownership of Properties (including landlords', carriers', warehousemen's, mechanics', materialmen's and other similar Liens for sums not yet due and payable or, if due and payable, which are being contested in Good Faith pursuant to appropriate proceedings and for which appropriate reserves have been established in accordance with GAAP), and Liens to secure the performance of bids, tenders, leases or trade contracts, or to secure statutory obligations (including under workers compensation, unemployment insurance and other social security Laws), surety or appeal bonds or other Liens incurred in the ordinary course of business and not in connection with borrowing money;
- (d) leases or subleases granted to others, easements, rights-of-way, licenses, encroachments, restrictions and similar charges or encumbrances, in each case incidental to the ownership of Property or ordinary conduct of business by the Company Group, or Liens incidental to minor survey exceptions and the like, *provided that* such Liens do not, in the aggregate, materially detract from the value of such Property;
 - (e) Liens securing Indebtedness of a Subsidiary to the Company;
 - (f) Liens existing as of the date hereof reflected in Schedule 8.5;
- Liens incurred after the date hereof to secure the purchase price or costs of (g) acquiring, constructing or improving Property (other than accounts receivable or inventory) useful and intended to be used in carrying on the business of the Company or a Subsidiary, including Liens existing on such Property at the time of acquisition, construction or improvement thereof or Liens incurred within 365 days of such acquisition or completion of such construction or improvements, provided that (i) any such Lien shall attach solely to the Property acquired, constructed or improved; (ii) at the time of acquisition, construction or improvement of such Property (or, in the case of any Lien incurred within 365 days of such acquisition or improvement, at the time of the incurrence of the Indebtedness secured by such Lien), the aggregate amount remaining unpaid on all Indebtedness secured by Liens on such Property, whether or not assumed by the Company or a Subsidiary, shall not exceed the lesser of (y) the cost of such acquisition, construction or improvements, or (z) the fair market value of such Property (as determined in Good Faith by one or more officers of the Company to whom authority to enter into the transaction has been delegated by the board of directors of the Company); and (iii) at the time of such incurrence and after giving effect thereto, no Default or Event of Default would exist;
- (h) any Lien existing on Property of a Person immediately prior to its being consolidated with or merged into the Company or a Subsidiary or its becoming a Subsidiary, or any Lien existing on Property acquired by the Company or any Subsidiary at the time such Property is so acquired (whether or not the Indebtedness secured thereby is assumed), provided that (i) no such Lien shall have been created or assumed in contemplation of such consolidation or merger or such Person's becoming a Subsidiary or such acquisition of Property, (ii) each such Lien shall extend solely to the Property so acquired and, if required by the instrument originally creating such Lien, other Property which is an improvement to or acquired for specific use in connection with such acquired Property, and (iii) at the time of such incurrence and after giving effect thereto, no Default or Event of Default would exist;
- (i) any extension, renewal or replacement of any Lien permitted by the preceding Sections 8.5(f) through 8.5(h), provided that (i) no additional Property is encumbered by such

- Lien, (ii) the unpaid principal amount of the obligations secured thereby is not increased on or after the date of extension, renewal or replacement, and (iii) at such time and immediately after giving effect thereto, no Default or Event of Default shall have occurred and be continuing;
- (j) Liens relating to customary rights of setoff, revocation, refund or chargeback of financial institutions under deposit or similar agreements where the Company or its Subsidiary maintains deposits in the ordinary course of business;
 - (k) Liens securing the Loans;
- (I) Liens securing Indebtedness of the Company or any Subsidiary under any Material Credit Facility (including the Term Note Facility), provided that notwithstanding the foregoing, the Company shall not, and shall not permit any of its Subsidiaries to, secure pursuant to this Section 8.5(I) any Indebtedness under any Material Credit Facility unless and until the Loans shall concurrently be secured and guaranteed equally and ratably with such Indebtedness pursuant to documentation reasonably acceptable to the Bank (and Collateral Agent and any required Noteholders) in substance and in form, which may include an intercreditor agreement and opinions of counsel to the Company and/or any such Subsidiary, as the case may be, reasonably acceptable to the Bank, Collateral Agent and any required Noteholders; and
- (m) other Liens securing Indebtedness of the Company Group not otherwise permitted by Sections 8.5(a) through 8.5(l), provided that Priority Debt shall not at any time exceed 5% of Consolidated Assets (determined as of the end of the then most recently ended Fiscal Quarter), provided further that notwithstanding the foregoing the Company shall not, and shall not permit any Subsidiary to, secure pursuant to this Section 8.5(m) any Indebtedness outstanding under or pursuant to any Material Credit Facility unless and until the Loans shall concurrently be secured and guaranteed equally and ratably with such Indebtedness pursuant to documentation acceptable to the Bank (and Collateral Agent and any required Noteholders) in substance and form, including an intercreditor agreement and opinions of counsel to the Company and/or any such Subsidiary, as the case may be, reasonably acceptable to the Bank.
- **8.6 Debt Service Coverage Ratio**. The Company will not permit the ratio of Consolidated EBITDA to Consolidated Debt Service for any period of four consecutive Fiscal Ouarters ended or ending on or after December 31, 2025, to be less than 1.10 to 1.00.
- **8.7 Priority Debt**. The Company will not at any time permit the aggregate amount of all Priority Debt to exceed 5.0% of Consolidated Assets (Consolidated Assets to be determined as of the end of the then most recently ended Fiscal Quarter).
- **8.8 Transfer of Assets**. The Company will not, nor permit any Subsidiary to, Transfer any Property to any other Person except:
- (a) Transfers of (i) inventory or similar Property in the ordinary course of business, (ii) used, worn-out, obsolete or surplus Property no longer used or useful in the business of the Company Group (including the abandonment or other disposition of intellectual property that is, in the reasonable judgment of the Company, no longer economically practicable to maintain or used or useful in the Water Business), and (iii) cash and cash equivalents;
- **(b)** Transfers of Property to the extent such Property is exchanged for credit against the purchase price of similar replacement Property, or the proceeds of such sale are applied with reasonable promptness to the purchase price of such replacement Property;

- **(c)** Transfers of Property (i) between Obligors, and (ii) between Subsidiaries that are not Subsidiary Guarantors;
- **(d)** the issuance of Equity Interests by any Subsidiaries to the Company or to any Subsidiary Guarantor;
 - (e) the granting of Liens not prohibited by Section 8.5;
- (f) leases or subleases or licenses or sublicenses in the ordinary course of business to third Persons not interfering in any material respect with the business of the Company Group;
- (g) any Transfer of Property that, with all other Property of the Company Group previously Transferred pursuant to this Section 8.8(g) during (i) the four Fiscal Quarter period ending with the Fiscal Quarter in which such Transfer occurs, do not represent more than 10% of Consolidated Assets on the last day immediately prior to the Fiscal Quarter in which such determination is made, or (ii) during the period beginning on the date hereof and continuing through the date of such Transfer, do not represent more than 20% of Consolidated Assets on the last day of the Fiscal Quarter ending immediately prior to the Fiscal Quarter in which such determination is made.

To the extent the Net Proceeds Amount with respect to any Transfer to a Person (other than an Affiliate or Subsidiary) is applied to a Debt Prepayment Application or a Property Reinvestment Application within one year after such disposition, then such Transfer (or, if less than all such Net Proceeds Amount is applied as contemplated above, the pro rata percentage thereof corresponding to the Net Proceeds Amount so applied), only for the purpose of determining compliance with Section 8.8(g) as of any date, will not be an asset disposition.

- **8.9 Restricted Payments**. The Company will not declare or make or incur any liability to declare or make any Restricted Payment unless at such time and immediately after giving effect to such action: (a) no Default or Event of Default would exist; and (b) the ratio of Consolidated EBITDA to Consolidated Debt Service at the end of each of the four consecutive Fiscal Quarters most recently ended (calculated in each such case based on the four consecutive Fiscal Quarters then ended) is not less than 1.25:1.00.
- **8.10 Restrictive Agreements**. No member of the Company Group will, directly or indirectly, enter into, incur or permit to exist any agreement or other arrangement that prohibits, restricts or imposes any condition on the ability of any Subsidiary to pay dividends or make other distributions with respect to any of its Equity Interests or to make or repay loans or advances to the Company or any other Subsidiary or to guaranty the Indebtedness of the Company or any Subsidiary.

ARTICLE 9 Default.

- 9.1 Event of Default. Each of the following will constitute an "Event of Default" hereunder:
- (a) the Company defaults in the payment of any Loan when due and payable, whether at maturity or any other date fixed for payment by declaration or otherwise; or

- **(b)** the Company defaults in the payment of any interest accrued on the Loans for more than five Business Days after such interest becomes due and payable; or
- (c) the Company defaults in the performance of or compliance with any term contained in Section 5.1(b) or (d) or Sections 8.1, 8.2, 8.5, 8.6, 8.8, 8.9 or 8.10; or
- (d) any Obligor defaults in the performance of or compliance with any term contained herein (other than those referred to in Sections 9.1(a), (b) and (c)) or in any other Loan Document and such Default is not remedied within 30 days after the earlier of (i) a Responsible Officer obtaining actual knowledge of such Default, and (ii) the Company receiving notice of such Default from the Bank; or
- (e) (i) any representation or warranty by or on behalf of the Company or by any officer of the Company in any Loan Document or writing furnished in connection with the transactions contemplated hereby is false or incorrect in any Material respect on the date as of which made, or (ii) any representation or warranty by or on behalf of any Subsidiary Guarantor or by any officer of such Subsidiary Guarantor in any Loan Document or writing furnished in connection with any such Loan Document is false or incorrect in any Material respect on the date as of which made; or
- (i) the Company or any Subsidiary is in default (as principal or as guarantor or other surety) in the payment of any principal of or premium or make-whole amount or interest on (A) the Note Purchase Agreement, (B) the Northern Trust Loan Agreement, or (C) any Indebtedness that is outstanding in an aggregate principal amount of at least \$15,000,000, beyond any period of grace provided with respect thereto, or (ii) the Company or any Subsidiary is in default in the performance of or compliance with any term of (A) the Note Purchase Agreement, (B) the Northern Trust Loan Agreement, or (C) any Indebtedness that is outstanding in an aggregate principal amount of at least \$15,000,000, or of any mortgage, indenture or other agreement relating thereto or any other condition exists, and as a consequence of such default or condition such Indebtedness has become, or has been declared (or one or more Persons are entitled to declare such Indebtedness to be), due and payable before its stated maturity or before its regularly scheduled dates of payment, or (iii) as a consequence of the occurrence or continuation of any event or condition (other than the passage of time or the right of the holder of Indebtedness to convert such Indebtedness into equity interests), (x) the Company or any Subsidiary has become obligated to purchase or repay Indebtedness before its regular maturity or before its regularly scheduled dates of payment under (A) the Note Purchase Agreement, (B) the Northern Trust Loan Agreement, or (C) any Indebtedness that is outstanding in an aggregate principal amount of at least \$15,000,000, or (y) one or more Persons have the right to require the Company or any Subsidiary so to purchase or repay such Indebtedness; or
- (g) the Company or any Subsidiary (i) is generally not paying, or admits in writing its inability to pay, its debts as they become due, (ii) files, or consents by answer or otherwise to the filing against it of, a petition for relief or reorganization or arrangement or any other petition in bankruptcy, for liquidation or to take advantage of any bankruptcy, insolvency, reorganization, moratorium or similar Law of any jurisdiction, (iii) makes an assignment for the benefit of its creditors, (iv) consents to the appointment of a custodian, receiver, trustee or other officer with similar powers with respect to it or any substantial part of its Property, (v) is adjudicated as insolvent or to be liquidated, or (vi) takes action for the purpose of any of the foregoing; or
- (h) a Governmental Authority of competent jurisdiction enters an order appointing, without consent by the Company or any of its Subsidiaries, a custodian, receiver, trustee or other officer with similar powers with respect to it or any substantial part of its Property, or constituting an order for relief or approving a petition for relief or reorganization or any other petition in bankruptcy or for liquidation or to take advantage of any bankruptcy or insolvency

law of any jurisdiction, or ordering the dissolution, winding-up or liquidation of the Company or any of its Subsidiaries, or any such petition is filed against the Company or any of its Subsidiaries and not dismissed within 60 days; or

- (i) any proceeding (judicial or administrative) shall be commenced against the Company or any Related Party, or with respect to any of their assets, which would reasonably be expected to have a Material Adverse Effect; or a judgment or settlement shall be entered or agreed to in any such proceeding which would reasonably be expected to have a Material Adverse Effect on the ability of the Company to repay its Indebtedness; or any garnishment, summons, writ of attachment, citation, levy or the like is issued against or served upon Bank for the attachment of any property of the Company or any Related Party in Bank's possession or control; or
- if (i) any Plan shall fail to satisfy the minimum funding standards of ERISA or the Code for any plan year or part thereof or a waiver of such standards or extension of any amortization period is sought or granted under section 412 of the Code, (ii) a notice of intent to terminate any Plan shall have been or is reasonably expected to be filed with the PBGC or the PBGC shall have instituted proceedings under ERISA section 4042 to terminate or appoint a trustee to administer any Plan or the PBGC shall have notified the Company or any ERISA Affiliate that a Plan may become a subject of any such proceedings, (iii) the aggregate "amount of unfunded benefit liabilities" (within the meaning of section 4001(a)(18) of ERISA) under all Plans, determined in accordance with Title IV of ERISA, shall exceed an amount that could reasonably be expected to have a Material Adverse Effect, (iv) the Company or any ERISA Affiliate shall have incurred or is reasonably expected to incur any liability pursuant to Title IV of ERISA or the penalty or excise tax provisions of the Code relating to employee benefit plans, (v) the Company or any ERISA Affiliate withdraws from any Multiemployer Plan, or (vi) the Company or any Subsidiary establishes or amends any employee welfare benefit plan that provides post-employment welfare benefits in a manner that would increase the liability of the Company or any Subsidiary thereunder; and any such event or events described in the preceding clauses (i) through (vi), individually or with any other such event or events, could reasonably be expected to have a Material Adverse Effect. As used in this Section 9.1(j), the terms "employee benefit plan" and "employee welfare benefit plan" have the meanings assigned in section 3 of ERISA; or
- (k) any Subsidiary Guaranty shall cease to be in full force and effect, any Subsidiary Guarantor or any Person acting on behalf of any Subsidiary Guarantor shall contest in any manner the validity, binding nature or enforceability of any Subsidiary Guaranty, or the obligations of any Subsidiary Guarantor under any Subsidiary Guaranty are not or cease to be legal, valid, binding and enforceable in accordance with the terms of such Subsidiary Guaranty; or
- (I) any Security Document ceases to give the Collateral Agent a perfected first priority Lien in any of the Collateral purported to be covered thereby for any reason; or any Security Document, at any time after its execution and delivery and for any reason, ceases to be in full force and effect; or any Obligor contests in writing the validity or enforceability of any Security Document; or any Obligor denies in writing that it has any further liability or obligation under any Security Document or purports to revoke, terminate or rescind any Security Document, other than, for each of the foregoing, as expressly permitted hereunder or thereunder (including by amendment, waiver and/or consent) or satisfaction in full of the Obligation; or
- (m) the Company or any Subsidiary fails to obtain or maintain in effect such licenses, certificates, permits, franchises and other governmental authorizations for any Utility Subsidiary as would, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.

- **9.2 Remedies**. Upon the occurrence and during the continuance of an Event of Default or Default, Bank will have no obligation to extend or continue to extend credit to the Company and may discontinue doing so at any time without prior notice or other limitation. In addition, upon the occurrence and during the continuance of any Event of Default, Bank may, upon notice to the Company:
- (a) Termination and Acceleration. Terminate any commitment and declare the unpaid principal balance of the Loans, all accrued interest thereon, and all other amounts payable under this Agreement, each Promissory Note, and all other Loan Documents to be immediately due and payable. Upon such a declaration, the unpaid principal balance of the Loans and all such other amounts will become immediately due and payable, without protest, presentment, demand, or further notice of any kind, all of which are hereby expressly waived by the Company.
- **(b) Enforcement**. Proceed to protect, exercise, and enforce such rights and remedies as may be provided by this Agreement, any security instrument or document, any other Loan Document, or under Law. Each and every one of such rights and remedies will be cumulative and may be exercised from time to time, and no failure on the part of Bank to exercise, and no delay in exercising, any right or remedy will operate as a waiver thereof, and no single or partial exercise of any right or remedy will preclude any future or other exercise thereof, or the exercise of any other right. Without limiting the foregoing, Bank may hold and/or set off and apply against the Company's obligations to Bank the proceeds of any equity in Bank, any cash collateral held by Bank, or any balances held by Bank for the Company's account (whether or not such balances are then due).
- (c) Application of Funds. Bank may apply all payments received by it to the Company's obligations to Bank in such order and manner as Bank may elect in its sole discretion.
- notwithstanding any Promissory Note: (1) upon the occurrence and during the continuance of an Event of Default, at Bank's option in each instance, the entire indebtedness outstanding hereunder and under each Promissory Note will bear interest from the date of such Event of Default until such Event of Default will have been waived or cured in a manner satisfactory to Bank at 2.00% per annum in excess of the rate(s) of interest that would otherwise be in effect on that Loan under the terms of the applicable Promissory Note; and (2) after the maturity of any Loan (whether as a result of acceleration or otherwise), the unpaid principal balance of such Loan (including without limitation, principal, interest, fees and expenses) will automatically bear interest at 2.00% per annum in excess of the rate(s) of interest that would otherwise be in effect on that Loan under the terms of the Promissory Note. All interest provided for herein will be payable on demand and will be calculated on the basis of a year consisting of 360 days.

ARTICLE 10 Expenses; Indemnification; Damage Waiver.

10.1 Costs and Expenses. To the extent allowed by Law, the Company agrees to pay all reasonable out-of-pocket costs and expenses (including the fees and expenses of counsel retained or employed by Bank) incurred by Bank and any participants of Bank in connection

with the origination, administration, collection and enforcement of this Agreement and the other Loan Documents, including, without limitation, all costs and expenses incurred in obtaining, perfecting, maintaining, determining the priority of, and releasing any security for the Company's obligations to Bank, and any stamp, intangible, transfer or like tax incurred in connection with this Agreement or any other Loan Document or the recording hereof or thereof.

- **Indemnification**. The Company indemnifies Bank, its Affiliates and its and their respective officers, directors, employees, agents and advisors (each an "Indemnitee") against, and holds each Indemnitee harmless from, any and all losses, claims, damages, liabilities and related expenses (including fees and expenses of employed or retained counsel) incurred by any Indemnitee or asserted against any Indemnitee by any third party arising out of or as a result of (a) the execution or delivery of any Loan Document, the performance or nonperformance by the Company of its obligations under any Loan Document or the consummation of the transactions contemplated thereby, including the use of the proceeds therefrom, (b) breach of representations, warranties or covenants of the Company under any Loan Document, or (c) any actual or prospective claim, litigation, investigation or proceeding relating to any of the foregoing, including any such items or losses relating to or arising under environmental Laws or pertaining to environmental matters, regardless whether any Indemnitee is a party thereto; provided that such indemnity will not, as to an Indemnitee, be available to the extent that such losses, claims, damages, liabilities or related expenses are determined by a court of competent jurisdiction by a final and nonappealable judgment to have resulted from the gross negligence or willful misconduct of such Indemnitee.
- 10.3 Waiver of Consequential Damages. To the fullest extent permitted by applicable Law, the Company will not assert, and hereby waives, any claim against any Indemnitee, on any theory of liability, for special, indirect, consequential or punitive damages arising out of, in connection with, or as a result of, any Loan Document, the transactions contemplated thereby or the use of the proceeds thereof.

ARTICLE 11 Miscellaneous.

- 11.1 Amendments; Waivers; Etc. No amendment, modification, or waiver of any provision of this Agreement or the other Loan Documents, and no consent to any departure by the Company herefrom or therefrom, will be effective unless approved by Bank and contained in a writing signed by or on behalf of Bank, and then such waiver or consent will be effective only in the specific instance and for the specific purpose for which given. In the event this Agreement is amended or restated, each such amendment or restatement will be applicable to all Promissory Notes hereto.
- 11.2 Notices. All notices hereunder will be in writing and will be deemed to have been duly given when addressed to the party intended to receive the same at the address of such party set forth below (or such other address either party may specify by like notice), (a) upon delivery if personally delivered to a party at such address, (b) three days after the same is deposited in the United States mail as first class, certified mail, return receipt requested, postage paid, (c) one Business Day after the same has been deposited with Federal Express or another nationally

recognized overnight courier service if designated for next-day delivery, and (d) upon delivery if sent by facsimile or electronic mail with confirmation of delivery of the same:

If to Bank, as follows:

For general correspondence purposes:

P.O. Box 5110 Denver, Colorado 80217-5110

For direct delivery purposes, when desired:

6340 South Fiddlers Green Circle Greenwood Village, Colorado 80111-1914 Attention: Credit Information Services

Fax No.: (303) 224-6101

If to the Company, as follows:

Global Water Resources, Inc. Attn: Michael J. Liebman Chief Financial Officer 21410 North 19th Avenue, Suite 220 Phoenix, Arizona 85027-2738

- 11.3 Survival. Notwithstanding anything to the contrary in this or any other Loan Document, Sections 5.5, 9.2, all of Article 10, Section 11.7 and Section 11.08 will survive the termination of this Agreement, repayment of every Promissory Note, and the foreclosure, or any other enforcement action, of any and all security described in Section 2.3 above. The representations, warranties, acknowledgments, and agreements set forth herein will survive the date of this Agreement, but not its termination unless otherwise agreed.
- all indebtedness and obligations of the Company under this Agreement and the other Loan Documents have been paid or satisfied; (b) Bank has no commitment to extend credit to or for the account of the Company under any Promissory Note; and (c) either party sends written notice to the other party terminating this Agreement. Any provision of this Agreement or any other Loan Document that is prohibited or unenforceable in any jurisdiction will be ineffective to the extent of such prohibition or unenforceable without invalidating the remaining provisions hereof or thereof.

11.5 Successors and Assigns.

(a) Successors and Assigns Generally. This Agreement and the other Loan Documents will be binding upon and inure to the benefit of the Company and Bank and their respective successors and assigns, except that the Company may not assign or transfer its rights

or obligations under this Agreement or the other Loan Documents without the prior written consent of Bank.

(b) Participations, Etc. From time to time, Bank may sell to one or more banks, financial institutions, or other lenders a participation in one or more of the Loans or other extensions of credit made pursuant to this Agreement. However, no such participation will relieve Bank of any commitment made to the Company hereunder. In connection with the foregoing, Bank may disclose information concerning the Company and its Subsidiaries to any participant or prospective participant, provided that such participant or prospective participant agrees to keep such information confidential. A sale of a participation interest may include certain voting rights of the participants regarding the Loans hereunder (including without limitation the administration, servicing, and enforcement thereof). Bank will maintain a record of all participation interests, a copy of which will be made available to the Company upon request.

11.6 Integration; Other Types of Credit; Counterparts; Electronic Signatures.

- (a) Integration. The Loan Documents are intended by the parties to be a complete and final expression of their agreement. Each Promissory Note will be deemed to incorporate all of the terms and conditions of this Agreement as if fully set forth therein. Without limiting the foregoing, any capitalized term utilized in any Promissory Note (or in any amendment to this Agreement or Promissory Note) and not otherwise defined in the Promissory Note (or amendment) will have the meaning set forth herein or, if applicable, in the accounting principles described in Article 1.
- **(b)** Other Types of Credit. From time to time, Bank may issue letters of credit or extend other types of credit to or for the account of the Company. In the event the parties desire to do so under the terms of this Agreement, then the agreement of the parties with respect thereto may be set forth in a Promissory Note and this Agreement will be applicable thereto.
- Counterparts; Electronic Signatures. This Agreement, each Promissory Note and any other Loan Document may be executed in counterparts, each of which will constitute an original, but all of which when taken together will constitute a single contract. Delivery of an executed counterpart of a signature page of this Agreement, any Promissory Note or any other Loan Document by facsimile or other electronic means will be as effective as delivery of a manually executed counterpart of each such Agreement, Promissory Note or Loan Document. The parties agree that the electronic signature of a party to this Agreement, any Promissory Note or any other Loan Document shall be as valid as an original signature of such party and shall be effective to bind such party to this Agreement or such Loan Document. The parties agree that any electronically signed Loan Document (including this Agreement) shall be deemed (i) to be "written" or "in writing," (ii) to have been signed and (iii) to constitute a record established and maintained in the ordinary course of business and an original written record when printed from electronic files. The parties presently intend to authenticate any Loan Documents to which they are a party by either signing such Loan Document or attaching thereto or logically associating therewith an electronic sound, symbol or process as their respective electronic signature. The words "execution," "signed," "signature," and words of like import in any Loan Document shall be deemed to include electronic signatures or the keeping of records in electronic form, each of

which shall be of the same legal effect, validity or enforceability as a manually executed signature or the use of a paper-based recordkeeping system, as the case may be, to the extent and as provided for in any applicable Law, including the Federal Electronic Signatures in Global and National Commerce Act, or any similar state Laws based on the Uniform Electronic Transactions Act.

11.7 Applicable Law; Submission to Jurisdiction; Service of Process; Waiver of Venue; Waiver of Jury Trial.

- (a) Applicable Law. Without giving effect to the principles of conflict of laws and except to the extent governed by federal law, the Laws of the State of Colorado, without reference to choice of law doctrine, will govern this Agreement, each Promissory Note and any other Loan Document for which Colorado is specified as the applicable law, and all disputes and matters between the parties to this Agreement, including all disputes and matters whatsoever arising under, in connection with or incident to the lending and/or leasing or other business relationship between the parties, and the rights and obligations of the parties to this Agreement or any other Loan Document by and between the parties for which Colorado is specified as the applicable law.
- **(b) Submission to Jurisdiction; Service of Process**. The Company hereby irrevocably consents to the nonexclusive jurisdiction of any state or federal court in Denver, Colorado, and consents that Bank may effect any service of process in the manner and at the Company's address set forth herein for providing notice or demand; provided that nothing contained in this Agreement will prevent Bank from bringing any action, enforcing any award or judgment or exercising any rights against the Company individually, against any collateral or against any property of the Company within any other county, state or other foreign or domestic jurisdiction.
- **(c) Waiver of Venue**. The Company acknowledges and agrees that the venue provided above is the most convenient forum for the Company and Bank. The Company waives any objection to venue and any objection based on a more convenient forum in any action instituted under this Agreement.
- (d) Waiver of Jury Trial. The Company and Bank each hereby irrevocably waives any right it may have to a trial by jury in connection with any action directly or indirectly arising out of or relating to this Agreement or any other Loan Document. Each party hereto (1) certifies that no representative, administrative agent or attorney of any other person has represented, expressly or otherwise, that such other person would not, in the event of litigation, seek to enforce the foregoing waiver and (2) acknowledges that it and the other party hereto has been induced to enter into this Agreement and other Loan Documents by, among other things, the mutual waivers and certifications in this Section.

11.8 Confidentiality.

(a) The Bank agrees to maintain the confidentiality of all information received from or on behalf of the Company relating to the Company or its Subsidiaries or their respective

businesses, other than any such information that is available to the Bank on a nonconfidential basis prior to disclosure by the Company or pertaining to this Agreement that is routinely provided to data service providers, including league table providers that serve the lending industry; provided that, and in the case of information received from or on behalf of the Company after the date hereof, such information is clearly identified at the time of delivery as being confidential information (the "Confidential Information"), except that the Bank may disclose Confidential Information:

- (i) to its Affiliates and the directors, officers, employees, partners, agents, trustees, administrators, managers, advisors, and representatives of it and its Affiliates in connection with the administration of this Agreement and the preservation, exercise, or enforcement of their rights under this Agreement (it being understood that the Persons to whom such disclosure is made will be informed of the confidential nature of the Confidential Information and instructed to keep such Confidential Information confidential);
- (ii) to the extent required or requested by any regulatory authority purporting to have jurisdiction over the Bank or any of its Affiliates or any director, officer, employee, partner, agent, trustee, administrator, manager, advisor, or representative of it or any of its Affiliates (including any self-regulatory authority or any securities exchange on which securities of the disclosing party or any Affiliate of the disclosing party are listed or traded, in which case the Bank shall inform the Company prior to such disclosure to the extent permitted by Law provided that the failure to provide such notice will not constitute a violation of this Agreement);
- (iii) pursuant to any subpoena, or order of any court administrative agency, in any pending legal, judicial, or administrative proceeding, or similar legal process, or to the extent required or requested by law, regulation, or a Governmental Authority; provided that, unless specifically prohibited by applicable law or court order, the Bank shall use reasonable efforts to notify the Company of any such request by any Governmental Authority or representative thereof prior to disclosure of such Confidential Information provided that the failure to provide such notice will not constitute a violation of this Agreement (other than any such request in connection with an examination or audit conducted by bank accountants or any banking regulatory authority exercising examining or regulatory authority over the Bank);
- (iv) in connection with the exercise of any remedies hereunder or under any other Loan Document or any suit, action, or proceeding relating to this Agreement or any other Loan Document or the enforcement of its rights hereunder or thereunder;
- (v) subject to an agreement containing provisions substantially the same as those of this Section, to (x) any actual or potential pledgee, assignee, transferee, or participant in connection with the pledge, assignment, or transfer by the Bank of any Loans or any participations therein or (y) any actual or prospective party to any swap, derivative, or other transaction under which payments are to be made by reference to the

Company or any Subsidiary or any of their respective obligations, this Agreement, or payments hereunder;

- (vi) for purposes of establishing a "due diligence" defense;
- (vii) with the written consent of the Company;
- (viii) to the extent such Confidential Information (x) becomes publicly available other than as a result of a breach of this Section, or (y) is available to the Bank on a non-confidential basis prior to disclosure by the Company, or (z) becomes available to the Bank or any of its Affiliates on a non-confidential basis from a source other than the Company that is not known to be subject to a confidentiality obligation to the Company.
- **(b)** Any Person required to maintain the confidentiality of Confidential Information as provided in this Section shall be considered to have complied with its obligation to do so if such Person has exercised the same degree of care to maintain the confidentiality of such Confidential Information as such Person would accord to its own confidential information.

SIGNATURE PAGE FOLLOWS

SIGNATURE PAGE TO CREDIT AGREEMENT

IN WITNESS WHEREOF, the parties hereto, by their duly authorized officers, have executed this Agreement.

GLOBAL WATER RESOURCES, INC.

By: <u>/s/ Michael J. Liebman</u> Name: Michael J. Liebman

Title: Senior Vice President, Chief Financial Officer and Secretary

SIGNATURE PAGE TO CREDIT AGREEMENT

IN WITNESS WHEREOF, the parties hereto, by their duly authorized officers, have executed this Agreement.

COBANK, ACB

By: <u>/s/ Fidel Escalante</u> Name: Fidel Escalante

Title: Assistant Corporate Secretary

EXHIBIT A

DEFINED TERMS

As used herein (or in any other Loan Document, unless otherwise defined therein) the term:

"Affiliate" means, at any time with respect to any Person, any other Person that then directly or indirectly through one or more intermediaries Controls, or is Controlled by, or is under common Control with, such first Person, and with respect to the Company includes any Person beneficially owning or holding, directly or indirectly, 10% or more of any class of Equity Interests of the Company or any Subsidiary or any Person of which the Company Group and its Subsidiaries beneficially own or hold, in the aggregate, directly or indirectly, 10% or more of any such class of Equity Interests. Unless the context otherwise clearly requires, reference to an "Affiliate" refers to an Affiliate of the Company.

"**Agreement**" means this Credit Agreement, including the Schedules and Exhibits hereto, as restated, supplemented and/or otherwise modified at any relevant time.

"Anti-Terrorism Laws" is defined in Section 4.15(c).

"Anti-Corruption Laws" is defined in Section 4.15(d).

"Blocked Person" is defined in Section 4.15(a).

"Capital Lease" means a lease with respect to which the lessee is required concurrently to recognize the acquisition of an asset and incurrence of a liability with GAAP.

"Change in Control" means any Person or related Persons constituting a group (as such term is used in Rule 13d-5 under the Exchange Act on the date hereof), other than William S. Levine and the Levine Family Members, shall have: (a) become the "beneficial owners" (as such term is used in Rule 13d-3 and Rule 13d-5 under the Exchange Act as in effect on the date hereof), directly or indirectly, of more than 50% of the total voting power of all common stock of the Company then outstanding and constituting Voting Stock; or (b) acquired after the date hereof (i) the power to elect, appoint or cause the election or appointment of at least a majority of the members of the board of directors of the Company, through beneficial ownership of the capital stock of the Company or otherwise, or (ii) all or substantially all of the Properties of the Company.

"Code" means the Internal Revenue Code of 1986, as amended from time to time, and the rules and regulations promulgated thereunder from time to time.

"Collateral" means all "Collateral" described in any of the Security Documents.

"Collateral Agency Agreement" means the Third Amended and Restated Collateral Agency Agreement dated as of October 26, 2023 among the Collateral Agent, the Company and

the Secured Parties from time to time (as amended, restated, supplemented, joined in and/or otherwise modified from time to time).

"Collateral Agent" means U.S. Bank National Association, in its capacity as Collateral Agent for the Secured Parties from time to time under the Security Documents, and any successor in such capacity appointed pursuant to the Collateral Agency Agreement.

"Company Group" means the Company and all Subsidiaries.

"Confidential Information" is defined in Section 11.8.

"Consolidated Assets" means, as of any date of determination, the total amount all assets of the Company Group, determined on a consolidated basis in accordance with GAAP.

"Consolidated Debt Service" means, for any period of time, the sum of: (a) Consolidated Interest Expense for such period; plus (b) all scheduled amortization payments of principal in respect of Indebtedness of the Company Group (including principal components of payments in respect of Capital Lease obligations) paid or payable during such period after eliminating offsetting debits and credits between the Company and its Subsidiaries and other items required to be eliminated in the preparation of consolidated financial statements of the Company Group in accordance with GAAP; plus (c) dividends, distributions or repurchases on or related to Equity Interests of the Company paid or payable during such period, in each case as of the date declared whether or not paid during any period.

"Consolidated EBITDA" means, for any period of time, Consolidated Net Income for such period plus, to the extent deducted in computing such Consolidated Net Income and without duplication: (a) depreciation, depletion and amortization expense for such period; (b) income tax expense for such period; (c) other non-cash charges net of non-cash income for such periods; and (d) Consolidated Interest Expense for such period all as determined in accordance with GAAP.

"Consolidated Interest Expense" means, for any period of time, the gross interest expense of the Company Group deducted in the calculation of Consolidated Net Income for such period less gross interest income for the Company Group, in each case determined on a consolidated basis in accordance with GAAP.

"Consolidated Net Income" means, for any period of time, the consolidated net income (or loss) of the Company Group for such period, determined on a consolidated basis in accordance with GAAP.

"Control" means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of a Person, whether through the ownership of voting securities, by contract or otherwise.

"Controlled Entity" means: (a) any Subsidiary of the Company and any of its or the Company's respective Controlled Affiliates; and (b) if the Company has a parent company, such parent company and its Controlled Affiliates.

"Debt Prepayment Application" means, with respect to any sales, leases, transfers and other dispositions of assets, the application by the Company of cash in an amount equal to the amount of proceeds from such Transfer (or portion thereof) to pay Senior Indebtedness; provided, that in the event such Senior Indebtedness would otherwise permit the reborrowing of such Senior Indebtedness, the commitment to relend such Senior Indebtedness shall be permanently reduced by the amount of such Debt Prepayment Application.

"**Default**" means an event or condition the occurrence or existence of which would, with the lapse of time or giving of notice or both, become an Event of Default.

"Disclosure Documents" is defined in Section 4.3.

"**EDGAR**" means the SEC's Electronic Data Gathering Analysis and Retrieval System or any successor SEC electronic filing system for such purposes.

"Environmental Laws" means any and all federal, state, local, and foreign statutes, Laws, permits, concessions, grants, franchises, licenses, agreements or governmental restrictions relating to pollution and the protection of the environment or the release of any materials into the environment, including those related to Hazardous Materials.

"Equity Interests" means, with respect to any Person, all shares of capital stock (including preferred stock) and/or membership interests of or in (or other ownership or profit interests in) such Person, all warrants, options or other rights for the purchase or acquisition from such Person of shares of capital stock of (or other ownership or profit interests in) such Person, all securities convertible into or exchangeable for shares of capital stock of (or other ownership or profit interests in) such Person or warrants, rights or options for the purchase or acquisition from such Person of such securities, and all other ownership or profit interests owned by such Person (including partnership, member or trust interests therein), whether voting or nonvoting.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended from time to time, and the rules and regulations promulgated thereunder from time to time in effect

"ERISA Affiliate" means any trade or business (whether or not incorporated) that is treated as a single employer together with the Company under section 414 of the Code.

"Event of Default" is defined in Section 9.

"Exchange Act" means the Securities Exchange Act of 1934, as amended.

"Executive Order No. 13224" means Executive Order No. 13224 on Terrorist Financing, effective September 24, 2001.

"Fiscal Quarter" means each of the four consecutive three-month periods comprising each Fiscal Year.

"Fiscal Year" means each calendar year ending on a December 31.

"Foreign Holding Company" means a Person which has as its principal purpose, and substantially all of the assets of which consist of, holding (directly or indirectly) ownership interests in "controlled foreign corporations" (as defined in Section 957 of the Code) and/or other Foreign Holding Companies.

"Foreign Subsidiary" means a Subsidiary organized under the Laws of a jurisdiction other than the United States of America or a state thereof or the District of Columbia.

"GAAP" means generally accepted accounting principles as in effect from time to time in the United States of America.

"Global Water" means Global Water, LLC, a Delaware limited liability company, and its successors and assigns.

"Global Water Holdings" means Global Water Holdings, Inc., an Arizona corporation, and its successors and assigns.

"Good Faith" has the meaning provided by Arizona Revised Statute § 47-1201 on the date hereof.

"Governmental Authority" means: (a) the government of (i) the United States of America or any state or other political subdivision of either, or (ii) any other jurisdiction in which the Company or any Subsidiary conducts all or any part of its business, or which asserts jurisdiction over any Properties of any of the Company Group; or (b) any entity exercising executive, legislative, judicial, regulatory or administrative functions of, or pertaining to, any such government.

"Governmental Official" means any governmental official or employee, employee of any government-owned or government-controlled entity, political party, any official of a political party, candidate for political office, official of any public international organization or anyone else acting in an official capacity.

"Guaranty" means, with respect to any Person, any obligation (except endorsement in the ordinary course of business of negotiable instruments for deposit or collection) of such Person guaranteeing or in effect guaranteeing any indebtedness, distribution, dividend or obligation of any other Person in any manner, directly or indirectly, including obligations incurred through an agreement, contingent or otherwise, by such Person: (a) to purchase such indebtedness or obligation or any Property constituting security therefor; (b) to advance or supply funds (i) for the purchase or payment of such indebtedness, distribution, dividend or obligation, or (ii) to maintain any working capital or other balance sheet or income statement condition of any other Person or otherwise to advance or make available funds for the purchase or payment of such indebtedness, distribution, dividend or obligation; (c) to lease or purchase property or services primarily for the purpose of assuring the owner of such indebtedness, distribution, dividend or obligation; or (d) otherwise to assure the owner of such indebtedness, distribution, dividend or obligation against loss in respect thereof. In any

computation of the liabilities of the obligor under any Guaranty, the indebtedness, distribution, dividend or other obligations that are the subject of such Guaranty shall be assumed to be direct obligations of such obligor; *provided that*, the amount outstanding for purposes of this Agreement shall not exceed the lesser of (i) the maximum amount of indebtedness that is the subject of such Guaranty, and (ii) any stated maximum recourse amount of such Guaranty.

"Hazardous Materials" means all pollutants, toxic or hazardous wastes or other substances that might pose a hazard to health and safety, the removal of which may be required or the generation, manufacture, refining, production, processing, treatment, storage, handling, transportation, transfer, use, disposal, release, discharge, spillage, seepage or filtration of which is or shall be restricted, prohibited or penalized by any applicable Law including, but not limited to, asbestos, urea formaldehyde foam insulation, polychlorinated biphenyls, petroleum, petroleum products, lead based paint, radon gas or similar restricted, prohibited or penalized substances.

"Indebtedness" with respect to any Person means at any time without duplication: (a) its liabilities for borrowed money and its redemption obligations in respect of mandatorily redeemable Preferred Stock; (b) its liabilities for the deferred purchase price of Property acquired by such Person (excluding accounts payable arising in the ordinary course of business but including liabilities arising under any conditional sale or other title retention agreement with respect to any such Property); (c) all liabilities (i) appearing on its balance sheet in accordance with GAAP in respect of Capital Leases, and (ii) which would appear on its balance sheet in accordance with GAAP in respect of Synthetic Leases assuming such Synthetic Leases were accounted for as Capital Leases; (d) all liabilities for borrowed money secured by any Lien with respect to any Property owned by such Person (whether or not it has assumed or otherwise become liable for such liabilities); (e) all its liabilities in respect of letters of credit or instruments serving a similar function issued or accepted for its account by financial institutions (whether or not representing obligations for borrowed money); (f) the aggregate Swap Termination Value of all Swap Contracts of such Person; and (g) any Guaranty of such Person with respect to liabilities described in any of the preceding clauses (a) through (f). Indebtedness of any Person includes all obligations of such Person described in clauses (a) through (g) to the extent such Person remains legally liable in respect thereof notwithstanding that any such obligation is deemed extinguished under GAAP.

"Levine Family Members" means: (a) the spouse of William S. Levine and each of his children and siblings; (b) the spouse and lineal descendants of any Person identified in the foregoing clause (a); (c) any trust or account primarily for the benefit of William S. Levine or any Person or Persons identified in the foregoing clauses (a) or (b); (d) any corporation, limited liability company, partnership or other entity in which William S. Levine or any of the Persons identified in the foregoing clauses (a), (b) or (c) are the beneficial owners of substantially all of the capital stock, membership interests, partnership interests or other Equity Interests and options or warrants to acquire, or securities convertible into, capital stock, membership interests, partnership interest, or other equity securities of an entity; and (e) the personal representative or guardian of William S. Levine or any of the Persons identified in the foregoing clauses (a) and (b) upon such Person's death for purposes of the administration of such Person's estate or upon

such Person's disability or incompetency for purposes of protection and management of the assets of such Person.

"Lien" means, with respect to any Person, any mortgage, lien, pledge, charge, security interest or other encumbrance, or any interest or title of any vendor, lessor, lender or other secured party to or of such Person under any conditional sale or other title retention agreement or Capital Lease, on or with respect to any Property of such Person (including in the case of stock, stockholder agreements, voting trust agreements and all similar arrangements).

"Loan Documents" means at any time, this Agreement, the Promissory Notes, the Subsidiary Guaranties, the Security Documents, and all other writings now or hereafter executed in connection with any of the foregoing, with any amendments, waivers and supplements thereto.

"Material" means material in relation to the business, operations, affairs, financial condition, assets or prospects of the Company Group as a whole.

"Material Adverse Effect" means a Material adverse effect on the: (a) business, operations, affairs, financial condition, assets or properties of the Company and its Subsidiaries taken as a whole; (b) ability of the Company to perform its obligations under this Agreement and the other Loan Documents; (c) ability of any Subsidiary Guarantor to perform its obligations under its Subsidiary Guaranty and the other Loan Documents; or (d) the validity or enforceability of this Agreement or any other Loan Document.

"Material Credit Facility" means, as to any of the Company Group: (a) the Term Note Facility; (b) loans under the Northern Trust Loan Agreement; (c) the Loans; and (d) in addition to and not in place of (a), (b) and (c), any other agreement entered into on or after the date hereof creating or evidencing Indebtedness for borrowed money by any of the Company Group or in respect of which any of the Company Group is an obligor or otherwise provides a guarantee or other credit support ("Credit Facility"), in a principal amount outstanding or available for borrowing equal to or greater than \$15,000,000.00 (or its equivalent in the currency of payment determined as of the closing of such Credit Facility based on the exchange rate of such currency), and if no Credit Facility equals or exceeds \$15,000,000.00, then the largest Credit Facility other than those in clauses (a), (b) and (c) above which equals or exceeds the Loans, will be a Material Credit Facility.

"Multiemployer Plan" means any Plan that is a "multiemployer plan" as such term is defined in section 4001(a)(3) of ERISA.

"Net Proceeds Amount" means, with respect to any Transfer by the Company or any Subsidiary, an amount equal to the difference of: (a) the aggregate consideration (valued at the fair market value thereof by the Company or such Subsidiary in Good Faith) received by the Company Group in respect of such Transfer; minus (b) all applicable taxes and all ordinary and reasonable out-of-pocket expenses incurred by the Company Group in connection with such Transfer.

"Northern Trust Loan Agreement" means the Loan Agreement dated as of April 30, 2020 by the Company to The Northern Trust Company, as amended by Modification Agreement dated April 30, 2021, Second Modification Agreement dated July 26, 2022, Letter Agreement dated August 25, 2022, Third Modification Agreement dated June 28, 2023, Fourth Modification Agreement dated October 26, 2023, Fifth Modification Agreement dated July 1, 2024, and Sixth Modification Agreement dated April 10, 2025, as further modified, supplemented and/or restated at any relevant time.

"Noteholders" has the meaning provided in the Collateral Agency Agreement.

"Note Purchase Agreement" means (a) the Note Purchase Agreement dated as of May 20, 2016, between the Company and Purchasers described therein, as modified by Amendment No. 1 dated as of December 19, 2017, Amendment No. 2 dated as of April 20, 2018, and Amendment No. 3 dated as of April 30, 2020, and as further modified, supplemented and/or restated at any relevant time, and (b) the Note Purchase Agreement dated as of October 26, 2023, between the Company and Purchasers described therein, as modified, supplemented and/or restated at any relevant time.

"Notes" has the meaning provided by the Note Purchase Agreement.

"Obligation" means any debt, liability or obligation related to the Loans and owed by any of the Obligors to the Bank (including to the Collateral Agent as agent for the Bank), of any kind or description (whether or not for the payment of money), direct or indirect, absolute or contingent, due or to become due, now or hereafter arising, including principal, interest, fees and expenses, payable or reimbursable by any of the Obligors pursuant to any of the Loan Documents.

"Obligors" means the Company and Subsidiary Guarantors.

"OFAC" means the Office of Foreign Assets Control, United States Department of the Treasury.

"OFAC Sanctions Program" means any economic or trade sanction OFAC is responsible for administering and enforcing.

"Officer's Certificate" means a certificate of a Senior Financial Officer or other officer of the Company whose responsibilities extend to the subject matter of such certificate.

"PBGC" means the Pension Benefit Guaranty Corporation referred to and defined in ERISA or any successor thereto.

"**Person**" means an individual, partnership, corporation, limited liability company, association, trust, unincorporated organization, business entity or Governmental Authority.

"Plan" means an "employee benefit plan" (as defined in section 3(3) of ERISA) subject to Title I of ERISA that is or, within the preceding five years, has been established or maintained, or to which contributions are or, within the preceding five years, have been made or

required to be made, by the Company or any ERISA Affiliate or with respect to which the Company or any ERISA Affiliate may have any liability.

"Preferred Stock" means any class of Equity Interests of a Person that is preferred over any other class of Equity Interests of such Person as to the payment of dividends, distributions or any amount upon liquidation or dissolution of such Person.

"Priority Debt" means (without duplication), as of the date of any determination, the sum of: (a) all unsecured Indebtedness of Subsidiaries (including all Guaranties of Indebtedness of the Company but excluding (x) unsecured Indebtedness owed to any of the Company Group, and (y) all Subsidiary Guaranties and all unsecured Guaranties of Indebtedness of the Company by any Subsidiary which has also guaranteed the Loans); and (b) all Indebtedness of the Company and its Subsidiaries secured by Liens other than Indebtedness secured by Liens permitted by any of Sections 8.5(a) through 8.5(l).

"Prohibited Person" means: (i) a Person that is listed in the Annex to, or is otherwise subject to the provisions of, Executive Order No. 13224; (ii) a Person owned or controlled by, or acting for or on behalf of, any Person that is listed in the Annex to, or is otherwise subject to the provisions of, Executive Order No. 13224; (iii) a Person with whom Bank is prohibited from dealing or otherwise engaging in any transaction by any Anti-Terrorism Law; (iv) a Person who commits, threatens or conspires to commit or supports "terrorism" as defined in Executive Order No. 13224; (v) a Person that is named as a "specially designated national and blocked person" on the most current list published by OFAC at its official website or at any replacement website or at any other official publication of such list; and (vi) a Person who is affiliated with a person described in clauses (i) - (v) above.

"**Property**" or "**Properties**" means, unless otherwise specifically limited, real or personal property of any kind, tangible or intangible, choate or inchoate.

"Property Reinvestment Application" means, with respect to any asset disposition, the application of the Net Proceeds Amount (or a portion thereof) with respect to such asset disposition to the acquisition by the Company Group of fixed or capital assets of the Company or any Subsidiary to be used in the business of such Person.

"Proposed Prepayment Date" has the meaning provided by Section 6.2(b).

"**Regulation D**" means Regulation D of the Board of Governors of the Federal Reserve System, as amended or supplemented from time to time.

"Related Party(ies)" means any Subsidiary Guarantor, any other Subsidiary, and, in addition as to the Company, to the extent applicable, any general or limited partner, controlling shareholder, joint venturer, member or manager, of the Company.

"**Responsible Officer**" means any Senior Financial Officer and any other officer of the Company with responsibility for the administration of the relevant portion of this Agreement.

"Restricted Payment" means any dividend or other distribution (in cash, securities or other Property) with respect to any Equity Interest of the Company or any Subsidiary, or any payment, other than balloon payments at final maturity (in cash, securities or other property), including any sinking fund or similar deposit, on account of the redemption, retirement, acquisition, cancellation or termination of any such Equity Interest or on account of any return of capital to any holder of any such Person's Equity Interests.

"SEC" means the Securities and Exchange Commission of the United States or any successor thereto.

"Secured Parties" means the Secured Parties (as defined in the Collateral Agency Agreement) that are party to the Collateral Agency Agreement from time to time.

"Securities" or "Security" shall have the meaning specified in section 2(1) of the Securities Act.

"Securities Act" means the Securities Act of 1933, and rules and regulations promulgated thereunder, as from time to time in effect.

"Security Agreements" means the four Pledge and Security Agreements dated as of the date hereof between the Company and the Collateral Agent, Global Water and the Collateral Agent, West Maricopa and the Collateral Agent, and Global Water Holdings and the Collateral Agent, in each case as supplemented, amended and restated or replaced from time to time.

"Security Documents" means: (a) the Collateral Agency Agreement; (b) the Security Agreements; and (c) other security documents entered into pursuant to Section 7.8 and any other security documents, financing statements and the like filed or recorded in connection with the foregoing.

"Senior Financial Officer" means the chief financial officer, principal accounting officer, treasurer or comptroller of the Company.

"Senior Indebtedness" means any Indebtedness of the Company or any Subsidiary owing to any Person other than the Company, a Subsidiary or an Affiliate and which is not expressed to be junior or subordinate to any other Indebtedness of the Company or any Subsidiary.

"Subsidiary" means, as to any Person, any other Person in which such first Person or one or more of its Subsidiaries or such first Person and one or more of its Subsidiaries owns sufficient equity or voting interests to enable it or them (as a group) ordinarily, in the absence of contingencies, to elect a majority of the directors (or Persons performing similar functions) of such second Person, and any partnership or joint venture if more than a 50% interest in the profits or capital thereof is owned by such first Person or one or more of its Subsidiaries or such first Person and one or more of its Subsidiaries (unless such partnership or joint venture can and does ordinarily take major business actions without the prior approval of such Person or one or

more of its Subsidiaries). Unless the context otherwise clearly requires, any reference to a "**Subsidiary**" is to a Subsidiary of the Company.

"Subsidiary Guarantor" means each of Global Water, West Maricopa, Global Water Holdings and any other Subsidiary that has executed and delivered a Subsidiary Guaranty pursuant to Section 7.7.

"Subsidiary Guaranties" means the Guaranty Agreements delivered by Global Water, West Maricopa and Global Water Holdings on the date hereof and any other guaranties delivered pursuant to Section 7.7(a).

"Swap Contract" means any: (a) interest rate swap transactions, basis swap transactions, basis swaps, credit derivative transactions, forward rate transactions, commodity swaps, commodity options, forward commodity contracts, equity or equity index swaps or options, bond or bond price or bond index swaps or options or forward foreign exchange transactions, cap transactions, floor transactions, currency options, spot contracts or any other similar transactions or any of the foregoing (including any options to enter into any of the foregoing); and (b) transactions of any kind, and related confirmations, subject to the terms and conditions of, or governed by any form of master agreement published by the International Swaps and Derivatives Association, Inc. or any International Foreign Exchange Master Agreement.

"Swap Termination Value" means, with respect of any one or more Swap Contracts, after taking into account any enforceable netting agreement relating to such Swap Contracts: (a) for any date on or after the date such Swap Contract has been closed out and termination value determined, such termination value; and (b) for any date prior to the date referenced in clause (a), the amount determined as the mark-to-market values for such Swap Contract, as determined based on one or more mid-market or other readily available quotations provided by any recognized dealer in such Swap Contracts.

"Synthetic Lease" means, at any time, any lease (including leases that may be terminated by the lessee at any time) of any Property: (a) accounted for as an operating lease under GAAP; and (b) in respect of which the lessee retains or obtains ownership of the Property so leased for U.S. federal income tax purposes, other than any such lease under which such Person is the lessor.

"Term Note Facility" (aka the "Private Placement Note Facility") means the credit facility provided pursuant to the Note Purchase Agreement.

"Transfer" means, with respect to any Person, any transaction (including by merger, consolidation or disposition of all or substantially all such Person's assets) in which such Person sells, conveys, transfers or leases (as lessor) any of its Property, including Subsidiary stock. "Transfer" includes the creation of minority interests in connection with any merger or consolidation involving a Subsidiary if the resulting entity is owned, directly or indirectly, by the Company in a proportion less than the proportion of ownership of such Subsidiary by the Company immediately preceding such merger or consolidation.

"UCC" means, with respect to any jurisdiction, the Uniform Commercial Code in effect in such jurisdiction.

"USA PATRIOT Act" means United States Public Law 107-56, Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism Act of 2001, as amended from time to time, and rules and regulations promulgated thereunder from time to time in effect.

"U.S. Economic Sanctions" is defined in Section 4.15(a).

"Utility Subsidiary" means each of the following Arizona corporations: (1) Global Water - 303 Utilities Company, Inc.; (2) Global Water - Palo Verde Utilities Company, Inc.; (3) Global Water - Santa Cruz Water Company, Inc.; (4) Global Water - CP Water Company, Inc.; (5) Global Water - Hassayampa Utilities Company, Inc.; (6) Global Water - Turner Ranches Irrigation, Inc.; (7) Global Water - Belmont Water Company, Inc.; (8) Water Utility Company of Greater Buckeye, Inc.; (9) Valencia Water Company, Inc.; (10) Willow Valley Water Company, Inc.; (11) Global Water - Saguaro District Water Company, Inc.; (12) Global Water - Farmers Water Company, Inc.; and (13) Global Water - Ocotillo Water Company, Inc.; and any successors at Law of each of the foregoing entities.

"Voting Stock" means, as of any date, the capital stock of the Company that is at the time entitled to vote in the election of the board of directors of the Company.

"Water Business" means the ownership, operation and management of water, wastewater and recycled water utilities, provided that the foregoing shall not prevent or restrict any of the Company Group from engaging in any business or activity reasonably related or ancillary to the Water Business.

"West Maricopa" means West Maricopa Combine, LLC, an Arizona limited liability company converted from West Maricopa Combine, Inc., an Arizona corporation, and its successors and assigns.

"Wholly Owned Subsidiary" means, at any time, any Subsidiary all of the Equity Interests (except any directors' qualifying shares) and voting interests of which are owned by any one or more of the Company and the Company's other Wholly Owned Subsidiaries at such time.

PROMISSORY NOTE (Single Advance Term Loan)

THIS PROMISSORY NOTE (this "Promissory Note") is entered into as of December 10, 2025 between GLOBAL WATER RESOURCES, INC., a Delaware corporation (the "Company"), and CoBANK, ACB, a federally-chartered instrumentality of the United States ("CoBank"), and supplements the Credit Agreement between the parties of even date herewith (as further amended or restated from time to time, the "Credit Agreement"). Terms used in this Promissory Note and not otherwise defined herein have the meanings given such terms in the Credit Agreement.

- **SECTION 1. The Commitment.** On the terms and subject to the conditions set forth in the Credit Agreement and this Promissory Note, CoBank agrees to make a loan (the "**Loan**") to the Borrower in a principal amount not to exceed \$15,000,000 (the "**Commitment**"). Under the Commitment, amounts borrowed and later repaid may not be re-borrowed.
- **SECTION 2. Purpose.** The purpose of the Commitment is to finance capital expenditures related to upgrading the following rural system assets and related closing costs: (a) rural assets of Global Water-Farmers Water Company, Inc., (b) Global Water-Belmont Water Company, Inc. and Global Water-Hassayampa Utilities Company, Inc. rural water and wastewater assets, and (c) Global Water-Santa Cruz Water Company, Inc. assets in Red Rock, Arizona
- **SECTION 3. Term of Commitment.** CoBank's obligation to make the Loan to the Borrower hereunder shall expire at 12:00 Noon, Denver, Colorado time, on the date of this Promissory Note, or on such later date as CoBank may, in its sole discretion, authorize in writing.

SECTION 4. Interest.

- (A) Rate. The Borrower agrees to pay interest on the unpaid balance of the Loan at the fixed rate of 5.49% per annum.
- (B) Calculation and Payment. Interest shall be calculated on the actual number of days the Loan is outstanding on the basis of a year consisting of 360 days. In calculating interest, the date the Loan is made shall be included and the date the Loan is repaid shall, if received before 3:00 p.m. Denver, Colorado time, be excluded. Interest shall be: (1) calculated semi-annually in arrears as of the last day of each May and November and on the final maturity date of the Loan; and (2) due and payable on the 15th day of each June and December and on the final maturity date of the Loan.
- **SECTION 5. Promissory Note.** The Borrower promises to pay to CoBank or order the principal amount on December 10, 2035. In addition to the above, the Borrower promises to pay to CoBank or order interest on the unpaid principal balance of the Loan at the times and in accordance with the provisions set forth above. If any date on which principal or interest is due is

not a Business Day, then such payment shall be due and payable on the next Business Day and, in the case of principal, interest shall continue to accrue on the amount thereof.

- **SECTION 6. Prepayment.** Subject to Section 2.5 of the Credit Agreement, the Borrower may, upon three Business Days' prior notice, prepay all or any portion of the Loan. Unless otherwise agreed, all prepayments will be applied to balances, fixed or variable, as CoBank shall direct.
- **SECTION 7. Security; Guaranties.** The Borrower's obligations hereunder and, to the extent related hereto, the Credit Agreement shall be secured as provided in Section 2.3 of the Credit Agreement. In addition, the Borrower's obligations hereunder and, to the extent related hereto, the Credit Agreement, are guaranteed by the Subsidiary Guarantors as provided in Section 2.3 of the Credit Agreement.
- **SECTION 8. Counterparts and Electronic Delivery**. This Promissory Note may be executed in counterparts (and by different parties in different counterparts), each of which shall constitute an original, and all of which when taken together shall constitute a single agreement. In addition, this Promissory Note may be delivered by electronic means.

SIGNATURE PAGE FOLLOWS

IN WITNESS WHEREOF, the parties have caused this Promissory Note to be executed by their duly authorized officers as of the date shown above.

CoBANK, ACB

GLOBAL WATER RESOURCES, INC.

By: /s/ Fidel Escalante /s/ Michael J. Liebman By: Name: Fidel Escalante Name: Michael J. Liebman

Title: Senior Vice President, Chief Financial
Officer and Secretary Assistant Corporate Secretary Title:

GUARANTY AGREEMENT

THIS GUARANTY AGREEMENT (this "Guaranty") dated as of December 10, 2025 is entered into by GLOBAL WATER, LLC, Delaware limited liability company (the "Guarantor"), in favor of COBANK, ACB, a federally-chartered instrumentality of the United States ("Lender").

BACKGROUND

Global Water Resources, Inc., a Delaware corporation (the "Borrower"), and Lender are entering into, among other agreements, a Credit Agreement of even date herewith between the Borrower and Lender (as amended, amended and restated, supplemented or otherwise modified from time to time, the "Credit Agreement") and a Promissory Note of even date herewith by the Borrower in favor of Lender (as amended, amended and restated, supplemented or otherwise modified from time to time, the "Initial Promissory Note"). Pursuant to those documents, Lender has agreed (subject to the terms and conditions therein) to make a \$15,000,000 loan to the Borrower. One of the conditions precedent to Lender's obligation to make the loan is that the Guarantor enter into this Guaranty. The Guarantor will derive substantial direct and indirect benefits from the loan, and therefore the Guarantor has agreed to enter into this Guaranty.

NOW, THEREFORE, in order to induce Lender to extend credit to the Borrower and for good and valuable other consideration, the receipt and sufficiency of which are hereby acknowledged, the Guarantor agrees as follows:

Guarantee of Payment. The Guarantor hereby unconditionally and irrevocably guarantees to Lender the punctual payment when due, whether at stated maturity, by acceleration or otherwise, of all indebtedness, obligations and liabilities of the Borrower to Lender, whether now existing or hereafter incurred, arising under the Credit Agreement, the Initial Promissory Note and the other Loan Documents (as hereinafter defined), including, but not limited to, those under or arising out of or in connection with any loans, advances, indemnities, or any other kind of contract or agreement under which the Borrower may be indebted to Lender in any manner under the Loan Documents, whether for principal, interest, fees, surcharges, expenses or otherwise. For ease of reference: (i) all such indebtedness, obligations and liabilities under the Loan Documents shall hereinafter be collectively referred to as the "Guaranteed Obligations"; and (ii) all instruments, documents and agreements evidencing or relating to the Guaranteed Obligations shall hereinafter collectively be referred to as the "Loan Documents." Without limiting the foregoing, the term: (1) "Loan Documents" shall include: (a) the Credit Agreement; (b) all Promissory Notes (as such term is defined in the Credit Agreement) issued under and pursuant to the Credit Agreement, whenever executed, including, without limitation, the Initial Promissory Note; (c) all other loan and loan related documentation executed at any time in connection with the Credit Agreement or any Promissory Note; and (d) all amendments to and restatements of the foregoing; and (2) "Guaranteed Obligations" shall include all indebtedness, obligations and liabilities of the Borrower to Lender arising under the: (a) the Credit Agreement; (b) the Initial Promissory; and (c) all other Loan Documents, including all obligations to pay principal, interest, fees, costs, surcharges, premiums, and other amounts arising thereunder.

2. Nature of Liability.

2.1 This is a continuing guarantee of payment and performance, and not a guaranty of collection. The liability of the Guarantor to Lender is separate and independent of the Guaranteed Obligations or of any liability of the Borrower thereunder and any liability of any other guarantors of the Guaranteed Obligations. This is a continuing guarantee, and the Guarantor acknowledges and agrees that the Guarantor's obligations hereunder shall cover Guaranteed Obligations incurred on the date hereof and at any time in the future. If an Event of Default (as defined in the Credit Agreement or any other Loan Document) has occurred and any applicable cure period has expired, then Lender shall have the right to declare the Guaranteed Obligations immediately due and payable in full, regardless of whether Lender has accelerated all or any part of the Borrower's indebtedness and regardless of whether Lender is prohibited from accelerating such indebtedness as a result of a bankruptcy, reorganization or like

proceeding involving the Borrower. Without limiting the generality of the foregoing, if the Borrower should at any time (i) become insolvent, (ii) make a general assignment for the benefit of creditors, (iii) petition for or be subject to a receivership proceeding, or (iv) be subject to a petition in bankruptcy or any insolvency or reorganization proceeding, whether voluntary or involuntary, then Lender shall have the right to declare the Guaranteed Obligations hereunder immediately due and payable in full, regardless of whether Lender has accelerated (or is permitted to accelerate) all or any part of Guaranteed Obligations, and Lender shall have the right to demand and to collect from the Guarantor payment in full of the Guaranteed Obligations hereunder, including all principal, interest, fees and charges, whether or not then due and payable by the Borrower. In addition, in the event the Borrower fails to make any payment due to Lender under the Loan Documents, Lender shall also have the right to make a partial demand hereunder in the amount of the defaulted payment or any part thereof and at a later date make one or more additional demands for payment hereunder (whether for all or a portion of the balance of the same defaulted payment, another defaulted payment, or the entire Guaranteed Obligations), all without relieving the Guarantor of its obligations hereunder; it being understood that until all Guaranteed Obligations are actually, finally, indefeasibly and unconditionally paid, performed or satisfied in full in accordance with the applicable terms of the Loan Documents, multiple demands may be made hereunder. In addition, Lender may make a partial demand for payment hereunder with or without terminating Lender's commitment to extend additional Guaranteed Obligations to the Borrower.

- The obligations of the Guarantor hereunder will remain in full force and effect and be enforceable until the Guaranteed Obligations are actually, finally, indefeasibly, and unconditionally paid, performed, or satisfied in full in accordance with the applicable terms of the Loan Documents. If, on account of the federal Bankruptcy Code, as amended, or any other debtor relief law (whether statutory, common law, case law or otherwise) of any jurisdiction whatsoever, now or hereafter in effect, that may be or become applicable, the Borrower will be or is relieved of or fails to perform any of the Guaranteed Obligations, the Guarantor will nevertheless be and remain fully liable to pay or perform all Guaranteed Obligations pursuant to this Guaranty. If the Guaranteed Obligations are partially paid or discharged for any reason, including voluntary payment or prepayment accepted by Lender, application of insurance proceeds or condemnation awards, permitted additional financing or refinancing, or sale of any collateral pledged by the Borrower for the Guaranteed Obligations or a portion thereof (the "Collateral"), with or without the consent or cooperation of Lender, this Guaranty will nevertheless remain in full force and effect, and the Guarantor will remain liable for all remaining Guaranteed Obligations, even though there may be rights that the Guarantor might have had against the Borrower that are destroyed or diminished by the exercise of any remedy or right by Lender, or as the result of any inaction by Lender.
- 2.3 If at any time all or any part of any payment made by the Borrower or the Guarantor or received by Lender from the Borrower or the Guarantor under or with respect to the Guaranteed Obligations or this Guaranty is or must be rescinded or returned for any reason whatsoever (including, but not limited to, the insolvency, bankruptcy or reorganization of the Guarantor or the Borrower), then the obligations of the Guarantor hereunder will, to the extent of the payment rescinded or returned, be deemed to have continued in existence, notwithstanding such previous payment made by the Borrower or the Guarantor, or receipt of payment by Lender, and the obligations of the Guarantor hereunder will continue to be effective or be reinstated, as the case may be, as to such payment, all as though such previous payment had never been made.
- **2.4** Specifically, to the maximum extent permitted by applicable law, this Guaranty and the Guarantor's liability hereunder will in no way be affected or impaired by reason of the happening from time to time of any of the following, whether or not any such event has occurred with or without notice to or consent of the Guarantor:
- (a) the waiver, compromise, settlement, termination or other release of the performance or observance by the Borrower or the Guarantor of any or all of the respective provisions, covenants, terms, or conditions contained in the Loan Documents or this Guaranty;

- **(b)** any failure, omission, delay or lack on the part of Lender to enforce, assert or exercise any right, power or remedy conferred on Lender in the Loan Documents or this Guaranty in accordance with their terms or the inability of Lender to enforce any provision of the Loan Documents or this Guaranty for any reason, or any other act or omission on the part of Lender;
- (c) the transfer, assignment or mortgaging, or the purported transfer, assignment or mortgaging, of all or any part of the Collateral, or the invalidity, unenforceability or termination for any reason whatsoever (other than performance) of any Loan Document or any provision, covenant, term or condition thereof or any right or remedy thereunder or any defect in the title of the Collateral or any part thereof or any loss of possession, use or operation of the Collateral or any part thereof by the Borrower;
- (d) the modification or amendment (whether material or otherwise) of any provision, covenant, term, or condition in any Loan Document;
- (e) the voluntary or involuntary liquidation, dissolution, sale of all or substantially all of the assets, marshalling of assets and liabilities, receivership, conservatorship, insolvency, bankruptcy, discharge, assignment for the benefit of creditors, reorganization, arrangement, composition or readjustment of, or other similar proceedings affecting, the Borrower or any of its assets or any allegation or contest of the validity of this Guaranty or any Loan Document in any such proceeding;
- **(f)** the surrender or impairment of any Collateral as security for the performance or observance of any of the Guaranteed Obligations;
- **(g)** any failure of the Guarantor or the Borrower to perform and observe any provision, covenant, term or condition, or to discharge any duty or obligation, arising out of or connected with the Guaranty or any Loan Document or the occurrence or pendency of any "Event of Default"

(as defined in any Loan Document) or any proceedings or actions as a result of, or attendant upon, such Event of Default;

- **(h)** the inability of the Guarantor or the Borrower to enforce any provision of any Loan Document for any reason;
- (i) the failure to give notice to the Guarantor or the Borrower of the occurrence of any default under this Guaranty or any of the Loan Documents;
- (j) the disposition by the Borrower of any ownership interest in the Guarantor or any permitted and permissible change, restructuring or termination of the organizational structure, ownership, or existence of the Borrower or of the Guarantor;
- **(k)** any set-off, counterclaim, reduction, or diminution of any Guaranteed Obligations, or any defense or discharge of any kind or nature whatsoever (other than performance), which the Guarantor or the Borrower may have or assert against Lender; or
- (I) any other circumstance (other than full payment and performance of all Guaranteed Obligations) that might otherwise constitute a legal or equitable defense or discharge of the Guarantor under this Guaranty or of the Borrower under any of the Loan Documents.
- 3. Specific Waivers by Guarantor. To the fullest extent permitted by applicable law, the Guarantor hereby waives and agrees not to assert or take advantage of (as a defense or set off or otherwise):
- (a) Any right to require Lender to proceed against the Borrower or any other Person or to proceed against or exhaust any Collateral or other security held by Lender at any time or to pursue any other remedy in Lender's power or under any other agreement before proceeding against the Guarantor hereunder;

- **(b)** The defense of any statute of limitations or statute of repose or of laches in any action hereunder;
- (c) Any defense that may arise by reason of the incapacity, lack of authority, death or disability of any other Person or Persons or the failure of Lender to file or enforce a claim against the estate (in administration, bankruptcy, or any other proceeding) of any other Person or Persons;
- (d) Any failure on the part of Lender to ascertain the extent or nature of the Collateral or any insurance or other rights with respect thereto, or the liability of any party liable for the Loan Documents or the Guaranteed Obligations;
- (e) Demand, presentment for payment, notice of nonpayment, protest, notice of protest and all other notices of any kind, or the lack of any thereof, including, without limiting the generality of the foregoing, notice of the existence, creation or incurring of any new or additional Guaranteed Obligations or of any action or non-action on the part of the Borrower, Lender, any endorser or creditor of the Borrower or of the Guarantor or on the part of any other Person whomsoever under this or any other instrument in connection with any Guaranteed Obligations held by or due to Lender;
 - **(f)** Any defense based upon an election of remedies by Lender;
 - (g) Any right or claim or right to cause a marshaling of the assets of the Guarantor;
- **(h)** Any principle or provision of law, statutory or otherwise, which is or might be in conflict with the terms and provisions of this Guaranty;
- (i) Any duty on the part of Lender to disclose to the Guarantor any facts Lender may now or hereafter know about the Borrower or the Collateral, regardless of whether Lender has reason to believe that any such facts materially increase the risk beyond that which the Guarantor intends to assume or has reason to believe that such facts are unknown to the Guarantor or has a reasonable opportunity to communicate such facts to the Guarantor, it being understood and agreed that the Guarantor is fully responsible for being and keeping informed of the financial condition of the Borrower, of the condition of the Collateral and of any and all circumstances bearing on the risk that liability may be incurred by the Guarantor hereunder;
 - (j) Any lack of notice of disposition or of any manner of disposition of any Collateral:
- **(k)** Failure to properly record any document or any other lack of due diligence by Lender in creating or perfecting a security interest in or collection, protection or realization upon any Collateral or in obtaining reimbursement or performance from any Person now or hereafter liable for the Loan Documents or any obligation secured thereby:
- (I) The inaccuracy of any representation or other provision contained in any Loan Document:
 - (m) Any sale or assignment of the Loan Documents, in whole or in part;
- (n) Any sale or assignment by the Borrower of the Collateral, or any portion thereof or interest therein, whether or not consented to by Lender;
- **(o)** Any invalidity, irregularity, or unenforceability, in whole or in part, of any one or more of the Loan Documents;
- **(p)** Any deficiencies in the Collateral or any deficiency in the ability of Lender to collect or to obtain performance from any Persons now or hereafter liable for the payment and performance of any obligation hereby guaranteed;

- (q) An assertion or claim that the automatic stay provided by 11 U.S.C. § 362 (arising upon the voluntary or involuntary bankruptcy proceeding of the Borrower) or any other stay provided under any other debtor relief law (whether statutory, common law, case law or otherwise) of any jurisdiction whatsoever, now or hereafter in effect, which may be or become applicable, will operate or be interpreted to stay, interdict, condition, reduce or inhibit the ability of Lender to enforce any of its rights, whether now or hereafter required, which Lender may have against the Guarantor or the Collateral;
- (r) Any modifications of the Loan Documents or any obligation of the Borrower relating to the Guaranteed Obligations by operation of law or by action of any court, whether pursuant to the Bankruptcy Reform Act of 1978, as amended, or any other debtor relief law (whether statutory, common law, case law or otherwise) of any jurisdiction whatsoever, now or hereafter in effect, or otherwise;
- (s) The release of the Borrower or of any other Person from performance or observance of any of the agreements, covenants, terms or conditions contained in any of the Loan Documents by operation of law, Lender's voluntary act or otherwise (other than by reason of the full payment and performance of the Guaranteed Obligations); and
- (t) Without limiting the generality of the foregoing, any rights and benefits which might otherwise be available to any guarantor under law.

4. Certain General Provisions.

- **4.1 Fully Recourse**. All of the terms and provisions of this Guaranty are recourse obligations of the Guarantor.
- Condition of Borrower. The Guarantor warrants and represents that the 4.2 Guarantor is fully aware of the financial condition of the Borrower and is executing and delivering this Guaranty based solely upon the Guarantor's own independent investigation of all matters pertinent hereto, and that the Guarantor is not relying in any manner upon any representation or statement of Lender. The Guarantor warrants, represents and agrees that the Guarantor is in a position to obtain, and the Guarantor hereby assumes full responsibility for obtaining, any additional information concerning the financial condition of the Borrower and any other matter pertinent hereto, and that the Guarantor is not relying upon Lender to furnish, and will have no right to require Lender to obtain or disclose, any information with respect to the Guaranteed Obligations, the financial condition or character of the Borrower or the ability of the Borrower to pay or perform the Guaranteed Obligations, the existence of any collateral or security for any or all of such indebtedness or obligations, the existence or nonexistence of any other guaranties of all or any part of such indebtedness or obligations, any actions or non-action on the part of Lender, the Borrower or any other Person, or any other matter, fact or occurrence whatsoever. By executing this Guaranty, the Guarantor acknowledges and knowingly accepts the full range of risks encompassed within a contract of guaranty.
- **4.3 Survival; Security**. This Guaranty will be deemed to be continuing in nature and will remain in full force and effect and will survive the exercise of any remedy by Lender under any of the Loan Documents, including, without limitation, any foreclosure or deed in lieu thereof, even if, as a part of such remedy, the Guaranteed Obligations are paid or satisfied in full; provided, however, this Guaranty shall terminate and be of no further force or effect upon the full and final indefeasible payment and performance in full of the Guaranteed Obligations and the termination of Lender's commitments under the Loan Documents.
- 4.4 Subrogation; No Recourse Against Lender. Unless and until the Guaranteed Obligations are indefeasibly paid in full in cash, the Guarantor will not have any right of subrogation, contribution, reimbursement, or indemnity whatsoever or any right of recourse to or with respect to the assets or property of the Borrower or to any Collateral. In connection with the foregoing, the Guarantor expressly subordinates any and all rights of subrogation to all rights and remedies of Lender against the Borrower, and the Guarantor hereby waives any rights to enforce any remedy which Lender may have against the Borrower and any right to participate in any

Collateral. In addition to and without in any way limiting the foregoing, the Guarantor hereby subordinates any and all indebtedness of the Borrower now or hereafter owed to the Guarantor to all indebtedness of the Borrower to Lender, and will not claim any offset or other reduction of the Guarantor's obligations hereunder because of any such indebtedness and will not take any collateral for or on account of any such indebtedness. Further, the Guarantor will not have any right of recourse against Lender by reason of any action Lender may take or omit to take under the provisions of this Guaranty, other than any action or inaction that would constitute a breach by Lender of the provisions of this Guaranty, or under the provisions of any of the Loan Documents.

- 4.5 **Consents by Guarantor**. The Guarantor hereby consents and agrees that Lender may at any time, and from time to time, without notice to or further consent from the Guarantor, either with or without consideration: release and surrender the Collateral or any portion thereof; substitute for any Collateral held by or on behalf of Lender other collateral of like kind, or of any kind; make other advances or increase the amount of the Guaranteed Obligations; agree to modify the terms of any one or more of the Loan Documents; extend or renew the Guaranteed Obligations for any period; grant releases, compromises and indulgences with respect to any one or more of the Loan Documents and to any Persons now or hereafter liable thereunder or hereunder; release any other guarantor or endorser of or other Person liable upon the Guaranteed Obligations or any of the Loan Documents; or take or fail to take any action of any type whatsoever. No such action which Lender will take or fail to take in connection with the Loan Documents or any Collateral, nor any course of dealing with the Borrower or any other Person, will limit, impair or release the Guarantor's obligations hereunder, affect this Guaranty in any way or afford the Guarantor any recourse against Lender. Nothing contained in this section will be construed to require Lender to take or refrain from taking any action referred to herein. For clarity, the rights of Lender in this Section 4.5 with respect to Collateral may be subject to the terms and conditions of the Collateral Agency Agreement.
- 4.6 Certain Bankruptcy Matters. The Guarantor covenants and agrees that, upon the commencement of a voluntary or involuntary bankruptcy proceeding by or against the Borrower, the Guarantor will not seek or cause the Borrower or any other Person to seek a supplemental stay or other relief, whether injunctive or otherwise, pursuant to 11 U.S.C. § 105 or any other provision of the federal Bankruptcy Code, as amended, or any other debtor relief law (whether statutory, common law, case law or otherwise), of any jurisdiction whatsoever, now or hereafter in effect, which may be or become applicable, to stay, interdict, condition, reduce or inhibit the ability of Lender to enforce any rights of Lender against the Guarantor or the Collateral by virtue of this Guaranty or otherwise.
- **4.7 Representations and Warranties by Guarantor**. To induce Lender to accept this Guaranty, the Guarantor hereby makes the following representations and warranties:
- **4.7.1** This Guaranty has been duly authorized by all necessary corporate or limited liability company action, as the case maybe, on the part of the Guarantor, and constitutes (and will constitute) legal, valid and binding obligations of the Guarantor enforceable against the Guarantor in accordance with its terms, except as such enforceability may be limited by: (i) applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting the enforcement of creditors' rights generally; and (ii) general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law).
- **4.7.2** The execution, delivery and performance by the Guarantor of this Guaranty will not: (i) result in the creation of any lien in respect of any property of the Guarantor under any indenture, mortgage, deed of trust, security agreement, loan, purchase or credit agreement, lease, corporate charter or by-laws, or any other agreement or instrument by which the Guarantor or any of its properties may be bound or affected, other than a lien in favor of Lender; or (ii) (1) contravene, result in any breach of, or constitute a default under, any indenture, mortgage, deed of trust, security agreement, loan, purchase or credit agreement, lease, corporate charter or by-laws, or any other agreement or instrument by which the Guarantor or any of its properties may be bound or affected; (2) conflict with or result in a breach of any of the terms, conditions or

provisions of any order, judgment, decree, or ruling of any court, arbitrator or governmental authority applicable to the Guarantor; or (3) violate any provision of any statute or other rule or regulation of any governmental authority applicable to the Guarantor, in each case to the extent such breach.

- **4.7.3** No consent, approval, or authorization of, or registration, filing or declaration with, any governmental authority is required in connection with the execution, delivery, or performance by the Guarantor of this Guaranty.
- **4.7.4** After executing and delivering this Guaranty, (i) the Guarantor will not be insolvent in that the fair saleable value of its assets will be greater than the amount required to pay its total liabilities; (ii) the Guarantor will not have incurred debts and other obligations in excess of its ability to pay such debts and obligations as they mature, and the cash flow of the Guarantor (after taking into account the anticipated uses of the cash) will be sufficient to pay such debts and obligations when such amounts are required to be paid; and (iii) the capital of the Guarantor will be sufficient and not unreasonably small for the business and transactions in which it is engaged.

5. Reserved.

- **Reservation of Rights**. No waiver under this Guaranty is valid unless it is in writing and signed by Lender. A waiver of a particular matter or remedy does not waive a subsequent or similar matter or remedy. No waiver will excuse the Guarantor from the performance of its other obligations to be performed under this Guaranty. The remedies conferred by this Guaranty are nonexclusive and cumulative of any other remedies now available or existing subsequently. No delay or forbearance by or on behalf of Lender in exercising any right, remedy, power or privilege under this Guaranty ("Lender's Rights") will operate as a waiver of such Lender's Rights, nor will the exercise or non-exercise of any particular Lender's Right preclude any other or further exercise of Lender's Rights pertaining to any current or subsequent default by the Guarantor or the Borrower. Nothing contained in this Guaranty will prevent or in any way diminish or interfere with any rights or remedies, including, without limitation, the right to contribution, which Lender may have against the Borrower, the Guarantor or any other party under the Comprehensive Environmental Response, Compensation and Liability Act of 1980 (codified at Title 42 U.S.C. § 9601 et seq.), as it may be amended from time to time, or any other applicable federal, state or local laws, all such rights being hereby expressly reserved.
- 7. Fees and Costs. In addition to the obligations provided above, in the event Lender incurs any cost and expense in connection with the enforcement of this Guaranty, the Guarantor agrees to reimburse Lender for any and all such costs and expenses, including: (i) its reasonable attorneys' fees; (ii) all reasonable costs and expenses of collection or enforcement; and (iii) all reasonable costs of suit, including all proceedings in courts of original, appellate and bankruptcy jurisdiction.
- **8. Notices and Other Communications.** Unless otherwise expressly provided herein, all notices and other communications provided for hereunder shall be in writing and shall be made to the following:

If to Lender, to:

CoBank, ACB 6340 S. Fiddlers Green Circle Greenwood Village, Colorado 80111-1914 Facsimile: (303) 224-6101 Attention: Credit Information Services If to the Guarantor, to:

c/o Global Water Resources, Inc. Attn: Michael J. Liebman Chief Financial Officer 21410 North 19th Avenue, Suite 220 Phoenix, Arizona 85027-2738

or to such other address, facsimile number, or electronic mail address as shall be designated by such party in a notice to the other party. All such notices and other communications shall be deemed to be given or made upon the earlier to occur of (i) actual receipt by the relevant party hereto and (ii) (A) if delivered by hand or by courier, when signed for by or on behalf of the relevant party hereto; and (B) if delivered by facsimile, when sent and receipt has been confirmed by telephone.

- **9.** Construction. Captions and headings are for convenience and reference only and do not define, limit, or affect the contents of this Guaranty. References to "paragraphs" or "sections" refer to this Guaranty unless stated otherwise. The terms "include" or "including" mean "without limitation by reason of enumeration." All grammatical usage will be deemed to refer to the masculine, feminine, neuter, singular or plural as the context and identity of any persons may require.
- 10. Severability and Interpretation. The invalidity or unenforceability of any provision of this Guaranty does not affect the other remaining provisions. This Guaranty will be construed as if it excluded any invalid or unenforceable provision, which will be severed from this Guaranty. Whenever possible, this Guaranty will be interpreted so as to be valid under applicable law, and will not be construed strictly in favor of or against any particular party, including any party who drafted or prepared this Guaranty, but instead according to its plain meaning to give effect to its intended purposes.
- 11. Governing Law. THIS AGREEMENT AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES HEREUNDER AND THEREUNDER SHALL BE GOVERNED BY AND CONSTRUED AND INTERPRETED IN ACCORDANCE WITH

THE LAWS OF THE STATE OF COLORADO. The Guarantor irrevocably consents to the service of process out of any competent court in any action or proceeding with respect to this Agreement by the mailing of copies thereof by registered or certified mail, postage prepaid, to it at the address for notices pursuant to Section 8, such service to become effective 30 days after such mailing. Nothing herein shall affect the right of Lender to serve process in any other manner permitted by law.

- 12. Waiver of Jury Trial. THE GUARANTOR AND LENDER EACH HEREBY IRREVOCABLY WAIVES ALL RIGHT TO TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM ARISING OUT OF OR RELATING TO THIS GUARANTY OR THE TRANSACTIONS CONTEMPLATED HEREBY AND THEREBY.
- 13. Binding Effect. This Guaranty is binding upon and inures to the benefit of the parties and their respective trustees, debtor(s)-in-possession, receivers, conservators, and the like—if any of the same are ever so appointed, and their successors and assigns.
 - **14. Acceptance**. The Guarantor hereby waives notice of acceptance hereof.

SIGNATURE PAGE FOLLOWS

SIGNATURE PAGE TO GUARANTY AGREEMENT

IN WITNESS WHEREOF, the Guarantor, by its duly authorized officer, has executed this Guaranty.

GLOBAL WATER, LLC

By: /s/ Michael J. Liebman

Name: Michael J. Liebman

Title: Manager, Secretary and Treasurer

GUARANTY AGREEMENT

THIS GUARANTY AGREEMENT (this "Guaranty") dated as of December 10, 2025 is entered into by WEST MARICOPA COMBINE, LLC, an Arizona limited liability company (the "Guarantor"), in favor of COBANK, ACB, a federally-chartered instrumentality of the United States ("Lender").

BACKGROUND

Global Water Resources, Inc., a Delaware corporation (the "Borrower"), and Lender are entering into, among other agreements, a Credit Agreement of even date herewith between the Borrower and Lender (as amended, amended and restated, supplemented or otherwise modified from time to time, the "Credit Agreement") and a Promissory Note of even date herewith by the Borrower in favor of Lender (as amended, amended and restated, supplemented or otherwise modified from time to time, the "Initial Promissory Note"). Pursuant to those documents, Lender has agreed (subject to the terms and conditions therein) to make a \$15,000,000 loan to the Borrower. One of the conditions precedent to Lender's obligation to make the loan is that the Guarantor enter into this Guaranty. The Guarantor will derive substantial direct and indirect benefits from the loan, and therefore the Guarantor has agreed to enter into this Guaranty.

NOW, THEREFORE, in order to induce Lender to extend credit to the Borrower and for good and valuable other consideration, the receipt and sufficiency of which are hereby acknowledged, the Guarantor agrees as follows:

1. Guarantee of Payment. The Guarantor hereby unconditionally and irrevocably guarantees to Lender the punctual payment when due, whether at stated maturity, by acceleration or otherwise, of all indebtedness, obligations and liabilities of the Borrower to Lender, whether now existing or hereafter incurred, arising under the Credit Agreement, the Initial Promissory Note and the other Loan Documents (as hereinafter defined), including, but not limited to, those under or arising out of or in connection with any loans, advances, indemnities, or any other kind of contract or agreement under which the Borrower may be indebted to Lender in any manner under the Loan Documents, whether for principal, interest, fees, surcharges, expenses or otherwise. For ease of reference: (i) all such indebtedness, obligations and liabilities under the Loan Documents shall hereinafter be collectively referred to as the "Guaranteed Obligations"; and (ii) all instruments, documents and agreements evidencing or relating to the Guaranteed Obligations shall hereinafter collectively be referred to as the "Loan Documents." Without limiting the foregoing, the term: (1) "Loan Documents" shall include: (a) the Credit Agreement; (b) all Promissory Notes (as such term is defined in the Credit Agreement) issued under and pursuant to the Credit Agreement, whenever executed, including, without limitation, the Initial Promissory Note; (c) all other loan and loan related documentation executed at any time in connection with the Credit Agreement or any Promissory Note; and (d) all amendments to and restatements of the foregoing; and (2) "Guaranteed Obligations" shall include all indebtedness, obligations and liabilities of the Borrower to Lender arising under the: (a) the Credit Agreement; (b) the Initial Promissory; and (c) all other Loan Documents, including all obligations to pay principal, interest, fees, costs, surcharges, premiums, and other amounts arising thereunder.

2. Nature of Liability.

- This is a continuing guarantee of payment and performance, and not a guaranty of 2.1 collection. The liability of the Guarantor to Lender is separate and independent of the Guaranteed Obligations or of any liability of the Borrower thereunder and any liability of any other guarantors of the Guaranteed Obligations. This is a continuing guarantee, and the Guarantor acknowledges and agrees that the Guarantor's obligations hereunder shall cover Guaranteed Obligations incurred on the date hereof and at any time in the future. If an Event of Default (as defined in the Credit Agreement or any other Loan Document) has occurred and any applicable cure period has expired, then Lender shall have the right to declare the Guaranteed Obligations immediately due and payable in full, regardless of whether Lender has accelerated all or any part of the Borrower's indebtedness and regardless of whether Lender is prohibited from accelerating such indebtedness as a result of a bankruptcy, reorganization or like proceeding involving the Borrower. Without limiting the generality of the foregoing, if the Borrower should at any time (i) become insolvent, (ii) make a general assignment for the benefit of creditors, (iii) petition for or be subject to a receivership proceeding, or (iv) be subject to a petition in bankruptcy or any insolvency or reorganization proceeding, whether voluntary or involuntary, then Lender shall have the right to declare the Guaranteed Obligations hereunder immediately due and payable in full, regardless of whether Lender has accelerated (or is permitted to accelerate) all or any part of Guaranteed Obligations, and Lender shall have the right to demand and to collect from the Guarantor payment in full of the Guaranteed Obligations hereunder, including all principal, interest, fees and charges, whether or not then due and payable by the Borrower. In addition, in the event the Borrower fails to make any payment due to Lender under the Loan Documents, Lender shall also have the right to make a partial demand hereunder in the amount of the defaulted payment or any part thereof and at a later date make one or more additional demands for payment hereunder (whether for all or a portion of the balance of the same defaulted payment, another defaulted payment, or the entire Guaranteed Obligations), all without relieving the Guarantor of its obligations hereunder; it being understood that until all Guaranteed Obligations are actually, finally, indefeasibly and unconditionally paid, performed or satisfied in full in accordance with the applicable terms of the Loan Documents, multiple demands may be made hereunder. In addition, Lender may make a partial demand for payment hereunder with or without terminating Lender's commitment to extend additional Guaranteed Obligations to the Borrower.
- 2.2 The obligations of the Guarantor hereunder will remain in full force and effect and be enforceable until the Guaranteed Obligations are actually, finally, indefeasibly, and unconditionally paid, performed, or satisfied in full in accordance with the applicable terms of the Loan Documents. If, on account of the federal Bankruptcy Code, as amended, or any other debtor relief law (whether statutory, common law, case law or otherwise) of any jurisdiction whatsoever, now or hereafter in effect, that may be or become applicable, the Borrower will be or is relieved of or fails to perform any of the Guaranteed Obligations, the Guarantor will nevertheless be and remain fully liable to pay or perform all Guaranteed Obligations pursuant to this Guaranty. If the Guaranteed Obligations are partially paid or discharged for any reason, including voluntary payment or prepayment accepted by Lender, application of insurance proceeds or condemnation awards, permitted additional financing or refinancing, or sale of any

collateral pledged by the Borrower for the Guaranteed Obligations or a portion thereof (the "Collateral"), with or without the consent or cooperation of Lender, this Guaranty will nevertheless remain in full force and effect, and the Guarantor will remain liable for all remaining Guaranteed Obligations, even though there may be rights that the Guarantor might have had against the Borrower that are destroyed or diminished by the exercise of any remedy or right by Lender, or as the result of any inaction by Lender.

- 2.3 If at any time all or any part of any payment made by the Borrower or the Guarantor or received by Lender from the Borrower or the Guarantor under or with respect to the Guaranteed Obligations or this Guaranty is or must be rescinded or returned for any reason whatsoever (including, but not limited to, the insolvency, bankruptcy or reorganization of the Guarantor or the Borrower), then the obligations of the Guarantor hereunder will, to the extent of the payment rescinded or returned, be deemed to have continued in existence, notwithstanding such previous payment made by the Borrower or the Guarantor, or receipt of payment by Lender, and the obligations of the Guarantor hereunder will continue to be effective or be reinstated, as the case may be, as to such payment, all as though such previous payment had never been made.
- **2.4** Specifically, to the maximum extent permitted by applicable law, this Guaranty and the Guarantor's liability hereunder will in no way be affected or impaired by reason of the happening from time to time of any of the following, whether or not any such event has occurred with or without notice to or consent of the Guarantor:
- (a) the waiver, compromise, settlement, termination or other release of the performance or observance by the Borrower or the Guarantor of any or all of the respective provisions, covenants, terms, or conditions contained in the Loan Documents or this Guaranty;
- **(b)** any failure, omission, delay or lack on the part of Lender to enforce, assert or exercise any right, power or remedy conferred on Lender in the Loan Documents or this Guaranty in accordance with their terms or the inability of Lender to enforce any provision of the Loan Documents or this Guaranty for any reason, or any other act or omission on the part of Lender;
- (c) the transfer, assignment or mortgaging, or the purported transfer, assignment or mortgaging, of all or any part of the Collateral, or the invalidity, unenforceability or termination for any reason whatsoever (other than performance) of any Loan Document or any provision, covenant, term or condition thereof or any right or remedy thereunder or any defect in the title of the Collateral or any part thereof or any loss of possession, use or operation of the Collateral or any part thereof by the Borrower;
- **(d)** the modification or amendment (whether material or otherwise) of any provision, covenant, term, or condition in any Loan Document;
- (e) the voluntary or involuntary liquidation, dissolution, sale of all or substantially all of the assets, marshalling of assets and liabilities, receivership, conservatorship, insolvency, bankruptcy, discharge, assignment for the benefit of creditors, reorganization, arrangement, composition or readjustment of, or other similar proceedings affecting, the Borrower or any of its

assets or any allegation or contest of the validity of this Guaranty or any Loan Document in any such proceeding;

- **(f)** the surrender or impairment of any Collateral as security for the performance or observance of any of the Guaranteed Obligations;
- **(g)** any failure of the Guarantor or the Borrower to perform and observe any provision, covenant, term or condition, or to discharge any duty or obligation, arising out of or connected with the Guaranty or any Loan Document or the occurrence or pendency of any "Event of Default" (as defined in any Loan Document) or any proceedings or actions as a result of, or attendant upon, such Event of Default;
- **(h)** the inability of the Guarantor or the Borrower to enforce any provision of any Loan Document for any reason;
- (i) the failure to give notice to the Guarantor or the Borrower of the occurrence of any default under this Guaranty or any of the Loan Documents;
- (j) the disposition by the Borrower of any ownership interest in the Guarantor or any permitted and permissible change, restructuring or termination of the organizational structure, ownership, or existence of the Borrower or of the Guarantor;
- **(k)** any set-off, counterclaim, reduction, or diminution of any Guaranteed Obligations, or any defense or discharge of any kind or nature whatsoever (other than performance), which the Guarantor or the Borrower may have or assert against Lender; or
- (I) any other circumstance (other than full payment and performance of all Guaranteed Obligations) that might otherwise constitute a legal or equitable defense or discharge of the Guarantor under this Guaranty or of the Borrower under any of the Loan Documents.
- **3. Specific Waivers by Guarantor.** To the fullest extent permitted by applicable law, the Guarantor hereby waives and agrees not to assert or take advantage of (as a defense or set off or otherwise):
- (a) Any right to require Lender to proceed against the Borrower or any other Person or to proceed against or exhaust any Collateral or other security held by Lender at any time or to pursue any other remedy in Lender's power or under any other agreement before proceeding against the Guarantor hereunder;
- **(b)** The defense of any statute of limitations or statute of repose or of laches in any action hereunder;
- **(c)** Any defense that may arise by reason of the incapacity, lack of authority, death or disability of any other Person or Persons or the failure of Lender to file or enforce a claim against the estate (in administration, bankruptcy, or any other proceeding) of any other Person or Persons;

- (d) Any failure on the part of Lender to ascertain the extent or nature of the Collateral or any insurance or other rights with respect thereto, or the liability of any party liable for the Loan Documents or the Guaranteed Obligations;
- (e) Demand, presentment for payment, notice of nonpayment, protest, notice of protest and all other notices of any kind, or the lack of any thereof, including, without limiting the generality of the foregoing, notice of the existence, creation or incurring of any new or additional Guaranteed Obligations or of any action or non-action on the part of the Borrower, Lender, any endorser or creditor of the Borrower or of the Guarantor or on the part of any other Person whomsoever under this or any other instrument in connection with any Guaranteed Obligations held by or due to Lender;
 - **(f)** Any defense based upon an election of remedies by Lender;
 - (g) Any right or claim or right to cause a marshaling of the assets of the Guarantor;
- **(h)** Any principle or provision of law, statutory or otherwise, which is or might be in conflict with the terms and provisions of this Guaranty;
- (i) Any duty on the part of Lender to disclose to the Guarantor any facts Lender may now or hereafter know about the Borrower or the Collateral, regardless of whether Lender has reason to believe that any such facts materially increase the risk beyond that which the Guarantor intends to assume or has reason to believe that such facts are unknown to the Guarantor or has a reasonable opportunity to communicate such facts to the Guarantor, it being understood and agreed that the Guarantor is fully responsible for being and keeping informed of the financial condition of the Borrower, of the condition of the Collateral and of any and all circumstances bearing on the risk that liability may be incurred by the Guarantor hereunder;
- **(j)** Any lack of notice of disposition or of any manner of disposition of any Collateral;
- **(k)** Failure to properly record any document or any other lack of due diligence by Lender in creating or perfecting a security interest in or collection, protection or realization upon any Collateral or in obtaining reimbursement or performance from any Person now or hereafter liable for the Loan Documents or any obligation secured thereby;
- (I) The inaccuracy of any representation or other provision contained in any Loan Document;
 - (m) Any sale or assignment of the Loan Documents, in whole or in part;
- (n) Any sale or assignment by the Borrower of the Collateral, or any portion thereof or interest therein, whether or not consented to by Lender;
- **(o)** Any invalidity, irregularity, or unenforceability, in whole or in part, of any one or more of the Loan Documents;

- **(p)** Any deficiencies in the Collateral or any deficiency in the ability of Lender to collect or to obtain performance from any Persons now or hereafter liable for the payment and performance of any obligation hereby guaranteed;
- (q) An assertion or claim that the automatic stay provided by 11 U.S.C. § 362 (arising upon the voluntary or involuntary bankruptcy proceeding of the Borrower) or any other stay provided under any other debtor relief law (whether statutory, common law, case law or otherwise) of any jurisdiction whatsoever, now or hereafter in effect, which may be or become applicable, will operate or be interpreted to stay, interdict, condition, reduce or inhibit the ability of Lender to enforce any of its rights, whether now or hereafter required, which Lender may have against the Guarantor or the Collateral;
- **(r)** Any modifications of the Loan Documents or any obligation of the Borrower relating to the Guaranteed Obligations by operation of law or by action of any court, whether pursuant to the Bankruptcy Reform Act of 1978, as amended, or any other debtor relief law (whether statutory, common law, case law or otherwise) of any jurisdiction whatsoever, now or hereafter in effect, or otherwise;
- **(s)** The release of the Borrower or of any other Person from performance or observance of any of the agreements, covenants, terms or conditions contained in any of the Loan Documents by operation of law, Lender's voluntary act or otherwise (other than by reason of the full payment and performance of the Guaranteed Obligations); and
- (t) Without limiting the generality of the foregoing, any rights and benefits which might otherwise be available to any guarantor under law.

4. Certain General Provisions.

- **4.1 Fully Recourse**. All of the terms and provisions of this Guaranty are recourse obligations of the Guarantor.
- 4.2 Condition of Borrower. The Guarantor warrants and represents that the Guarantor is fully aware of the financial condition of the Borrower and is executing and delivering this Guaranty based solely upon the Guarantor's own independent investigation of all matters pertinent hereto, and that the Guarantor is not relying in any manner upon any representation or statement of Lender. The Guarantor warrants, represents and agrees that the Guarantor is in a position to obtain, and the Guarantor hereby assumes full responsibility for obtaining, any additional information concerning the financial condition of the Borrower and any other matter pertinent hereto, and that the Guarantor is not relying upon Lender to furnish, and will have no right to require Lender to obtain or disclose, any information with respect to the Guaranteed Obligations, the financial condition or character of the Borrower or the ability of the Borrower to pay or perform the Guaranteed Obligations, the existence of any collateral or security for any or all of such indebtedness or obligations, the existence or nonexistence of any other guaranties of all or any part of such indebtedness or obligations, any actions or non-action on the part of Lender, the Borrower or any other Person, or any other matter, fact or occurrence

whatsoever. By executing this Guaranty, the Guarantor acknowledges and knowingly accepts the full range of risks encompassed within a contract of guaranty.

- **4.3 Survival; Security**. This Guaranty will be deemed to be continuing in nature and will remain in full force and effect and will survive the exercise of any remedy by Lender under any of the Loan Documents, including, without limitation, any foreclosure or deed in lieu thereof, even if, as a part of such remedy, the Guaranteed Obligations are paid or satisfied in full; provided, however, this Guaranty shall terminate and be of no further force or effect upon the full and final indefeasible payment and performance in full of the Guaranteed Obligations and the termination of Lender's commitments under the Loan Documents.
- Subrogation; No Recourse Against Lender. Unless and until the Guaranteed Obligations are indefeasibly paid in full in cash, the Guarantor will not have any right of subrogation, contribution, reimbursement, or indemnity whatsoever or any right of recourse to or with respect to the assets or property of the Borrower or to any Collateral. In connection with the foregoing, the Guarantor expressly subordinates any and all rights of subrogation to all rights and remedies of Lender against the Borrower, and the Guarantor hereby waives any rights to enforce any remedy which Lender may have against the Borrower and any right to participate in any Collateral. In addition to and without in any way limiting the foregoing, the Guarantor hereby subordinates any and all indebtedness of the Borrower now or hereafter owed to the Guarantor to all indebtedness of the Borrower to Lender, and will not claim any offset or other reduction of the Guarantor's obligations hereunder because of any such indebtedness and will not take any collateral for or on account of any such indebtedness. Further, the Guarantor will not have any right of recourse against Lender by reason of any action Lender may take or omit to take under the provisions of this Guaranty, other than any action or inaction that would constitute a breach by Lender of the provisions of this Guaranty, or under the provisions of any of the Loan Documents.
- 4.5 Consents by Guarantor. The Guarantor hereby consents and agrees that Lender may at any time, and from time to time, without notice to or further consent from the Guarantor, either with or without consideration; release and surrender the Collateral or any portion thereof; substitute for any Collateral held by or on behalf of Lender other collateral of like kind, or of any kind; make other advances or increase the amount of the Guaranteed Obligations; agree to modify the terms of any one or more of the Loan Documents; extend or renew the Guaranteed Obligations for any period; grant releases, compromises and indulgences with respect to any one or more of the Loan Documents and to any Persons now or hereafter liable thereunder or hereunder; release any other guarantor or endorser of or other Person liable upon the Guaranteed Obligations or any of the Loan Documents; or take or fail to take any action of any type whatsoever. No such action which Lender will take or fail to take in connection with the Loan Documents or any Collateral, nor any course of dealing with the Borrower or any other Person, will limit, impair or release the Guarantor's obligations hereunder, affect this Guaranty in any way or afford the Guarantor any recourse against Lender. Nothing contained in this section will be construed to require Lender to take or refrain from taking any action referred to herein. For clarity, the rights of Lender in this Section 4.5 with respect to Collateral may be subject to the terms and conditions of the Collateral Agency Agreement.

- 4.6 Certain Bankruptcy Matters. The Guarantor covenants and agrees that, upon the commencement of a voluntary or involuntary bankruptcy proceeding by or against the Borrower, the Guarantor will not seek or cause the Borrower or any other Person to seek a supplemental stay or other relief, whether injunctive or otherwise, pursuant to 11 U.S.C. § 105 or any other provision of the federal Bankruptcy Code, as amended, or any other debtor relief law (whether statutory, common law, case law or otherwise), of any jurisdiction whatsoever, now or hereafter in effect, which may be or become applicable, to stay, interdict, condition, reduce or inhibit the ability of Lender to enforce any rights of Lender against the Guarantor or the Collateral by virtue of this Guaranty or otherwise.
- **4.7 Representations and Warranties by Guarantor**. To induce Lender to accept this Guaranty, the Guarantor hereby makes the following representations and warranties:
- **4.7.1** This Guaranty has been duly authorized by all necessary corporate or limited liability company action, as the case maybe, on the part of the Guarantor, and constitutes (and will constitute) legal, valid and binding obligations of the Guarantor enforceable against the Guarantor in accordance with its terms, except as such enforceability may be limited by: (i) applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting the enforcement of creditors' rights generally; and (ii) general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law).
- 4.7.2 The execution, delivery and performance by the Guarantor of this Guaranty will not: (i) result in the creation of any lien in respect of any property of the Guarantor under any indenture, mortgage, deed of trust, security agreement, loan, purchase or credit agreement, lease, corporate charter or by-laws, or any other agreement or instrument by which the Guarantor or any of its properties may be bound or affected, other than a lien in favor of Lender; or (ii) (1) contravene, result in any breach of, or constitute a default under, any indenture, mortgage, deed of trust, security agreement, loan, purchase or credit agreement, lease, corporate charter or by-laws, or any other agreement or instrument by which the Guarantor or any of its properties may be bound or affected; (2) conflict with or result in a breach of any of the terms, conditions or provisions of any order, judgment, decree, or ruling of any court, arbitrator or governmental authority applicable to the Guarantor; or (3) violate any provision of any statute or other rule or regulation of any governmental authority applicable to the Guarantor, in each case to the extent such breach
- **4.7.3** No consent, approval, or authorization of, or registration, filing or declaration with, any governmental authority is required in connection with the execution, delivery, or performance by the Guarantor of this Guaranty.
- **4.7.4** After executing and delivering this Guaranty, (i) the Guarantor will not be insolvent in that the fair saleable value of its assets will be greater than the amount required to pay its total liabilities; (ii) the Guarantor will not have incurred debts and other obligations in excess of its ability to pay such debts and obligations as they mature, and the cash flow of the Guarantor (after taking into account the anticipated uses of the cash) will be sufficient to pay such debts and obligations when such amounts are required to be paid; and (iii) the capital of the

Guarantor will be sufficient and not unreasonably small for the business and transactions in which it is engaged.

5. Reserved.

- **6.** Reservation of Rights. No waiver under this Guaranty is valid unless it is in writing and signed by Lender. A waiver of a particular matter or remedy does not waive a subsequent or similar matter or remedy. No waiver will excuse the Guarantor from the performance of its other obligations to be performed under this Guaranty. The remedies conferred by this Guaranty are nonexclusive and cumulative of any other remedies now available or existing subsequently. No delay or forbearance by or on behalf of Lender in exercising any right, remedy, power or privilege under this Guaranty ("Lender's Rights") will operate as a waiver of such Lender's Rights, nor will the exercise or non-exercise of any particular Lender's Right preclude any other or further exercise of Lender's Rights pertaining to any current or subsequent default by the Guarantor or the Borrower. Nothing contained in this Guaranty will prevent or in any way diminish or interfere with any rights or remedies, including, without limitation, the right to contribution, which Lender may have against the Borrower, the Guarantor or any other party under the Comprehensive Environmental Response, Compensation and Liability Act of 1980 (codified at Title 42 U.S.C. § 9601 et seq.), as it may be amended from time to time, or any other applicable federal, state or local laws, all such rights being hereby expressly reserved.
- 7. Fees and Costs. In addition to the obligations provided above, in the event Lender incurs any cost and expense in connection with the enforcement of this Guaranty, the Guarantor agrees to reimburse Lender for any and all such costs and expenses, including: (i) its reasonable attorneys' fees; (ii) all reasonable costs and expenses of collection or enforcement; and (iii) all reasonable costs of suit, including all proceedings in courts of original, appellate and bankruptcy jurisdiction.
- **8. Notices and Other Communications.** Unless otherwise expressly provided herein, all notices and other communications provided for hereunder shall be in writing and shall be made to the following:

If to Lender, to:

CoBank, ACB 6340 S. Fiddlers Green Circle Greenwood Village, Colorado 80111-1914

Facsimile: (303) 224-6101

Attention: Credit Information Services

If to the Guarantor, to:

c/o Global Water Resources, Inc. Attn: Michael J. Liebman Chief Financial Officer 21410 North 19th Avenue, Suite 220 Phoenix, Arizona 85027-2738

or to such other address, facsimile number, or electronic mail address as shall be designated by such party in a notice to the other party. All such notices and other communications shall be deemed to be given or made upon the earlier to occur of (i) actual receipt by the relevant party hereto and (ii) (A) if delivered by hand or by courier, when signed for by or on behalf of the relevant party hereto; and (B) if delivered by facsimile, when sent and receipt has been confirmed by telephone.

- **9. Construction**. Captions and headings are for convenience and reference only and do not define, limit, or affect the contents of this Guaranty. References to "paragraphs" or "sections" refer to this Guaranty unless stated otherwise. The terms "include" or "including" mean "without limitation by reason of enumeration." All grammatical usage will be deemed to refer to the masculine, feminine, neuter, singular or plural as the context and identity of any persons may require.
- 10. Severability and Interpretation. The invalidity or unenforceability of any provision of this Guaranty does not affect the other remaining provisions. This Guaranty will be construed as if it excluded any invalid or unenforceable provision, which will be severed from this Guaranty. Whenever possible, this Guaranty will be interpreted so as to be valid under applicable law, and will not be construed strictly in favor of or against any particular party, including any party who drafted or prepared this Guaranty, but instead according to its plain meaning to give effect to its intended purposes.
- 11. Governing Law. THIS AGREEMENT AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES HEREUNDER AND THEREUNDER SHALL BE GOVERNED BY AND CONSTRUED AND INTERPRETED IN ACCORDANCE WITH THE LAWS OF THE STATE OF COLORADO. The Guarantor irrevocably consents to the service of process out of any competent court in any action or proceeding with respect to this Agreement by the mailing of copies thereof by registered or certified mail, postage prepaid, to it at the address for notices pursuant to Section 8, such service to become effective 30 days after such mailing. Nothing herein shall affect the right of Lender to serve process in any other manner permitted by law.
- 12. Waiver of Jury Trial. THE GUARANTOR AND LENDER EACH HEREBY IRREVOCABLY WAIVES ALL RIGHT TO TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM ARISING OUT OF OR RELATING TO THIS GUARANTY OR THE TRANSACTIONS CONTEMPLATED HEREBY AND THEREBY.
- **13. Binding Effect**. This Guaranty is binding upon and inures to the benefit of the parties and their respective trustees, debtor(s)-in-possession, receivers, conservators, and the like —if any of the same are ever so appointed, and their successors and assigns.
 - **14. Acceptance**. The Guarantor hereby waives notice of acceptance hereof.

SIGNATURE PAGE FOLLOWS

SIGNATURE PAGE TO GUARANTY AGREEMENT

IN WITNESS WHEREOF, the Guarantor, by its duly authorized officer, has executed this Guaranty.

WEST MARICOPA COMBINE, LLC

By: /s/ Michael J. Liebman

Name: Michael J. Liebman

Title: Manager

GUARANTY AGREEMENT

THIS GUARANTY AGREEMENT (this "Guaranty") dated as of December 10, 2025 is entered into by GLOBAL WATER HOLDINGS, INC., an Arizona corporation (the "Guarantor"), in favor of COBANK, ACB, a federally-chartered instrumentality of the United States ("Lender").

BACKGROUND

Global Water Resources, Inc., a Delaware corporation (the "Borrower"), and Lender are entering into, among other agreements, a Credit Agreement of even date herewith between the Borrower and Lender (as amended, amended and restated, supplemented or otherwise modified from time to time, the "Credit Agreement") and a Promissory Note of even date herewith by the Borrower in favor of Lender (as amended, amended and restated, supplemented or otherwise modified from time to time, the "Initial Promissory Note"). Pursuant to those documents, Lender has agreed (subject to the terms and conditions therein) to make a \$15,000,000 loan to the Borrower. One of the conditions precedent to Lender's obligation to make the loan is that the Guarantor enter into this Guaranty. The Guarantor will derive substantial direct and indirect benefits from the loan, and therefore the Guarantor has agreed to enter into this Guaranty.

NOW, THEREFORE, in order to induce Lender to extend credit to the Borrower and for good and valuable other consideration, the receipt and sufficiency of which are hereby acknowledged, the Guarantor agrees as follows:

1. Guarantee of Payment. The Guarantor hereby unconditionally and irrevocably guarantees to Lender the punctual payment when due, whether at stated maturity, by acceleration or otherwise, of all indebtedness, obligations and liabilities of the Borrower to Lender, whether now existing or hereafter incurred, arising under the Credit Agreement, the Initial Promissory Note and the other Loan Documents (as hereinafter defined), including, but not limited to, those under or arising out of or in connection with any loans, advances, indemnities, or any other kind of contract or agreement under which the Borrower may be indebted to Lender in any manner under the Loan Documents, whether for principal, interest, fees, surcharges, expenses or otherwise. For ease of reference: (i) all such indebtedness, obligations and liabilities under the Loan Documents shall hereinafter be collectively referred to as the "Guaranteed Obligations"; and (ii) all instruments, documents and agreements evidencing or relating to the Guaranteed Obligations shall hereinafter collectively be referred to as the "Loan Documents." Without limiting the foregoing, the term: (1) "Loan Documents" shall include: (a) the Credit Agreement; (b) all Promissory Notes (as such term is defined in the Credit Agreement) issued under and pursuant to the Credit Agreement, whenever executed, including, without limitation, the Initial Promissory Note; (c) all other loan and loan related documentation executed at any time in connection with the Credit Agreement or any Promissory Note; and (d) all amendments to and restatements of the foregoing; and (2) "Guaranteed Obligations" shall include all indebtedness, obligations and liabilities of the Borrower to Lender arising under the: (a) the Credit Agreement; (b) the Initial Promissory; and (c) all other Loan Documents, including all obligations to pay principal, interest, fees, costs, surcharges, premiums, and other amounts arising thereunder.

2. Nature of Liability.

- This is a continuing guarantee of payment and performance, and not a guaranty of 2.1 collection. The liability of the Guarantor to Lender is separate and independent of the Guaranteed Obligations or of any liability of the Borrower thereunder and any liability of any other guarantors of the Guaranteed Obligations. This is a continuing guarantee, and the Guarantor acknowledges and agrees that the Guarantor's obligations hereunder shall cover Guaranteed Obligations incurred on the date hereof and at any time in the future. If an Event of Default (as defined in the Credit Agreement or any other Loan Document) has occurred and any applicable cure period has expired, then Lender shall have the right to declare the Guaranteed Obligations immediately due and payable in full, regardless of whether Lender has accelerated all or any part of the Borrower's indebtedness and regardless of whether Lender is prohibited from accelerating such indebtedness as a result of a bankruptcy, reorganization or like proceeding involving the Borrower. Without limiting the generality of the foregoing, if the Borrower should at any time (i) become insolvent, (ii) make a general assignment for the benefit of creditors, (iii) petition for or be subject to a receivership proceeding, or (iv) be subject to a petition in bankruptcy or any insolvency or reorganization proceeding, whether voluntary or involuntary, then Lender shall have the right to declare the Guaranteed Obligations hereunder immediately due and payable in full, regardless of whether Lender has accelerated (or is permitted to accelerate) all or any part of Guaranteed Obligations, and Lender shall have the right to demand and to collect from the Guarantor payment in full of the Guaranteed Obligations hereunder, including all principal, interest, fees and charges, whether or not then due and payable by the Borrower. In addition, in the event the Borrower fails to make any payment due to Lender under the Loan Documents, Lender shall also have the right to make a partial demand hereunder in the amount of the defaulted payment or any part thereof and at a later date make one or more additional demands for payment hereunder (whether for all or a portion of the balance of the same defaulted payment, another defaulted payment, or the entire Guaranteed Obligations), all without relieving the Guarantor of its obligations hereunder; it being understood that until all Guaranteed Obligations are actually, finally, indefeasibly and unconditionally paid, performed or satisfied in full in accordance with the applicable terms of the Loan Documents, multiple demands may be made hereunder. In addition, Lender may make a partial demand for payment hereunder with or without terminating Lender's commitment to extend additional Guaranteed Obligations to the Borrower.
- 2.2 The obligations of the Guarantor hereunder will remain in full force and effect and be enforceable until the Guaranteed Obligations are actually, finally, indefeasibly, and unconditionally paid, performed, or satisfied in full in accordance with the applicable terms of the Loan Documents. If, on account of the federal Bankruptcy Code, as amended, or any other debtor relief law (whether statutory, common law, case law or otherwise) of any jurisdiction whatsoever, now or hereafter in effect, that may be or become applicable, the Borrower will be or is relieved of or fails to perform any of the Guaranteed Obligations, the Guarantor will nevertheless be and remain fully liable to pay or perform all Guaranteed Obligations pursuant to this Guaranty. If the Guaranteed Obligations are partially paid or discharged for any reason, including voluntary payment or prepayment accepted by Lender, application of insurance proceeds or condemnation awards, permitted additional financing or refinancing, or sale of any

collateral pledged by the Borrower for the Guaranteed Obligations or a portion thereof (the "Collateral"), with or without the consent or cooperation of Lender, this Guaranty will nevertheless remain in full force and effect, and the Guarantor will remain liable for all remaining Guaranteed Obligations, even though there may be rights that the Guarantor might have had against the Borrower that are destroyed or diminished by the exercise of any remedy or right by Lender, or as the result of any inaction by Lender.

- 2.3 If at any time all or any part of any payment made by the Borrower or the Guarantor or received by Lender from the Borrower or the Guarantor under or with respect to the Guaranteed Obligations or this Guaranty is or must be rescinded or returned for any reason whatsoever (including, but not limited to, the insolvency, bankruptcy or reorganization of the Guarantor or the Borrower), then the obligations of the Guarantor hereunder will, to the extent of the payment rescinded or returned, be deemed to have continued in existence, notwithstanding such previous payment made by the Borrower or the Guarantor, or receipt of payment by Lender, and the obligations of the Guarantor hereunder will continue to be effective or be reinstated, as the case may be, as to such payment, all as though such previous payment had never been made.
- **2.4** Specifically, to the maximum extent permitted by applicable law, this Guaranty and the Guarantor's liability hereunder will in no way be affected or impaired by reason of the happening from time to time of any of the following, whether or not any such event has occurred with or without notice to or consent of the Guarantor:
- (a) the waiver, compromise, settlement, termination or other release of the performance or observance by the Borrower or the Guarantor of any or all of the respective provisions, covenants, terms, or conditions contained in the Loan Documents or this Guaranty;
- **(b)** any failure, omission, delay or lack on the part of Lender to enforce, assert or exercise any right, power or remedy conferred on Lender in the Loan Documents or this Guaranty in accordance with their terms or the inability of Lender to enforce any provision of the Loan Documents or this Guaranty for any reason, or any other act or omission on the part of Lender;
- (c) the transfer, assignment or mortgaging, or the purported transfer, assignment or mortgaging, of all or any part of the Collateral, or the invalidity, unenforceability or termination for any reason whatsoever (other than performance) of any Loan Document or any provision, covenant, term or condition thereof or any right or remedy thereunder or any defect in the title of the Collateral or any part thereof or any loss of possession, use or operation of the Collateral or any part thereof by the Borrower;
- **(d)** the modification or amendment (whether material or otherwise) of any provision, covenant, term, or condition in any Loan Document;
- (e) the voluntary or involuntary liquidation, dissolution, sale of all or substantially all of the assets, marshalling of assets and liabilities, receivership, conservatorship, insolvency, bankruptcy, discharge, assignment for the benefit of creditors, reorganization, arrangement, composition or readjustment of, or other similar proceedings affecting, the Borrower or any of its

assets or any allegation or contest of the validity of this Guaranty or any Loan Document in any such proceeding;

- **(f)** the surrender or impairment of any Collateral as security for the performance or observance of any of the Guaranteed Obligations;
- (g) any failure of the Guarantor or the Borrower to perform and observe any provision, covenant, term or condition, or to discharge any duty or obligation, arising out of or connected with the Guaranty or any Loan Document or the occurrence or pendency of any "Event of Default" (as defined in any Loan Document) or any proceedings or actions as a result of, or attendant upon, such Event of Default;
- **(h)** the inability of the Guarantor or the Borrower to enforce any provision of any Loan Document for any reason;
- (i) the failure to give notice to the Guarantor or the Borrower of the occurrence of any default under this Guaranty or any of the Loan Documents;
- (j) the disposition by the Borrower of any ownership interest in the Guarantor or any permitted and permissible change, restructuring or termination of the organizational structure, ownership, or existence of the Borrower or of the Guarantor;
- **(k)** any set-off, counterclaim, reduction, or diminution of any Guaranteed Obligations, or any defense or discharge of any kind or nature whatsoever (other than performance), which the Guarantor or the Borrower may have or assert against Lender; or
- (I) any other circumstance (other than full payment and performance of all Guaranteed Obligations) that might otherwise constitute a legal or equitable defense or discharge of the Guarantor under this Guaranty or of the Borrower under any of the Loan Documents.
- **3. Specific Waivers by Guarantor.** To the fullest extent permitted by applicable law, the Guarantor hereby waives and agrees not to assert or take advantage of (as a defense or set off or otherwise):
- (a) Any right to require Lender to proceed against the Borrower or any other Person or to proceed against or exhaust any Collateral or other security held by Lender at any time or to pursue any other remedy in Lender's power or under any other agreement before proceeding against the Guarantor hereunder;
- **(b)** The defense of any statute of limitations or statute of repose or of laches in any action hereunder;
- **(c)** Any defense that may arise by reason of the incapacity, lack of authority, death or disability of any other Person or Persons or the failure of Lender to file or enforce a claim against the estate (in administration, bankruptcy, or any other proceeding) of any other Person or Persons;

- (d) Any failure on the part of Lender to ascertain the extent or nature of the Collateral or any insurance or other rights with respect thereto, or the liability of any party liable for the Loan Documents or the Guaranteed Obligations;
- (e) Demand, presentment for payment, notice of nonpayment, protest, notice of protest and all other notices of any kind, or the lack of any thereof, including, without limiting the generality of the foregoing, notice of the existence, creation or incurring of any new or additional Guaranteed Obligations or of any action or non-action on the part of the Borrower, Lender, any endorser or creditor of the Borrower or of the Guarantor or on the part of any other Person whomsoever under this or any other instrument in connection with any Guaranteed Obligations held by or due to Lender;
 - **(f)** Any defense based upon an election of remedies by Lender;
 - (g) Any right or claim or right to cause a marshaling of the assets of the Guarantor;
- **(h)** Any principle or provision of law, statutory or otherwise, which is or might be in conflict with the terms and provisions of this Guaranty;
- (i) Any duty on the part of Lender to disclose to the Guarantor any facts Lender may now or hereafter know about the Borrower or the Collateral, regardless of whether Lender has reason to believe that any such facts materially increase the risk beyond that which the Guarantor intends to assume or has reason to believe that such facts are unknown to the Guarantor or has a reasonable opportunity to communicate such facts to the Guarantor, it being understood and agreed that the Guarantor is fully responsible for being and keeping informed of the financial condition of the Borrower, of the condition of the Collateral and of any and all circumstances bearing on the risk that liability may be incurred by the Guarantor hereunder;
- **(j)** Any lack of notice of disposition or of any manner of disposition of any Collateral;
- **(k)** Failure to properly record any document or any other lack of due diligence by Lender in creating or perfecting a security interest in or collection, protection or realization upon any Collateral or in obtaining reimbursement or performance from any Person now or hereafter liable for the Loan Documents or any obligation secured thereby;
- (I) The inaccuracy of any representation or other provision contained in any Loan Document;
 - (m) Any sale or assignment of the Loan Documents, in whole or in part;
- (n) Any sale or assignment by the Borrower of the Collateral, or any portion thereof or interest therein, whether or not consented to by Lender;
- **(o)** Any invalidity, irregularity, or unenforceability, in whole or in part, of any one or more of the Loan Documents;

- **(p)** Any deficiencies in the Collateral or any deficiency in the ability of Lender to collect or to obtain performance from any Persons now or hereafter liable for the payment and performance of any obligation hereby guaranteed;
- (q) An assertion or claim that the automatic stay provided by 11 U.S.C. § 362 (arising upon the voluntary or involuntary bankruptcy proceeding of the Borrower) or any other stay provided under any other debtor relief law (whether statutory, common law, case law or otherwise) of any jurisdiction whatsoever, now or hereafter in effect, which may be or become applicable, will operate or be interpreted to stay, interdict, condition, reduce or inhibit the ability of Lender to enforce any of its rights, whether now or hereafter required, which Lender may have against the Guarantor or the Collateral;
- **(r)** Any modifications of the Loan Documents or any obligation of the Borrower relating to the Guaranteed Obligations by operation of law or by action of any court, whether pursuant to the Bankruptcy Reform Act of 1978, as amended, or any other debtor relief law (whether statutory, common law, case law or otherwise) of any jurisdiction whatsoever, now or hereafter in effect, or otherwise;
- **(s)** The release of the Borrower or of any other Person from performance or observance of any of the agreements, covenants, terms or conditions contained in any of the Loan Documents by operation of law, Lender's voluntary act or otherwise (other than by reason of the full payment and performance of the Guaranteed Obligations); and
- (t) Without limiting the generality of the foregoing, any rights and benefits which might otherwise be available to any guarantor under law.

4. Certain General Provisions.

- **4.1 Fully Recourse**. All of the terms and provisions of this Guaranty are recourse obligations of the Guarantor.
- 4.2 Condition of Borrower. The Guarantor warrants and represents that the Guarantor is fully aware of the financial condition of the Borrower and is executing and delivering this Guaranty based solely upon the Guarantor's own independent investigation of all matters pertinent hereto, and that the Guarantor is not relying in any manner upon any representation or statement of Lender. The Guarantor warrants, represents and agrees that the Guarantor is in a position to obtain, and the Guarantor hereby assumes full responsibility for obtaining, any additional information concerning the financial condition of the Borrower and any other matter pertinent hereto, and that the Guarantor is not relying upon Lender to furnish, and will have no right to require Lender to obtain or disclose, any information with respect to the Guaranteed Obligations, the financial condition or character of the Borrower or the ability of the Borrower to pay or perform the Guaranteed Obligations, the existence of any collateral or security for any or all of such indebtedness or obligations, the existence or nonexistence of any other guaranties of all or any part of such indebtedness or obligations, any actions or non-action on the part of Lender, the Borrower or any other Person, or any other matter, fact or occurrence

whatsoever. By executing this Guaranty, the Guarantor acknowledges and knowingly accepts the full range of risks encompassed within a contract of guaranty.

- **4.3 Survival; Security**. This Guaranty will be deemed to be continuing in nature and will remain in full force and effect and will survive the exercise of any remedy by Lender under any of the Loan Documents, including, without limitation, any foreclosure or deed in lieu thereof, even if, as a part of such remedy, the Guaranteed Obligations are paid or satisfied in full; provided, however, this Guaranty shall terminate and be of no further force or effect upon the full and final indefeasible payment and performance in full of the Guaranteed Obligations and the termination of Lender's commitments under the Loan Documents.
- Subrogation; No Recourse Against Lender. Unless and until the Guaranteed Obligations are indefeasibly paid in full in cash, the Guarantor will not have any right of subrogation, contribution, reimbursement, or indemnity whatsoever or any right of recourse to or with respect to the assets or property of the Borrower or to any Collateral. In connection with the foregoing, the Guarantor expressly subordinates any and all rights of subrogation to all rights and remedies of Lender against the Borrower, and the Guarantor hereby waives any rights to enforce any remedy which Lender may have against the Borrower and any right to participate in any Collateral. In addition to and without in any way limiting the foregoing, the Guarantor hereby subordinates any and all indebtedness of the Borrower now or hereafter owed to the Guarantor to all indebtedness of the Borrower to Lender, and will not claim any offset or other reduction of the Guarantor's obligations hereunder because of any such indebtedness and will not take any collateral for or on account of any such indebtedness. Further, the Guarantor will not have any right of recourse against Lender by reason of any action Lender may take or omit to take under the provisions of this Guaranty, other than any action or inaction that would constitute a breach by Lender of the provisions of this Guaranty, or under the provisions of any of the Loan Documents.
- 4.5 Consents by Guarantor. The Guarantor hereby consents and agrees that Lender may at any time, and from time to time, without notice to or further consent from the Guarantor, either with or without consideration: release and surrender the Collateral or any portion thereof; substitute for any Collateral held by or on behalf of Lender other collateral of like kind, or of any kind; make other advances or increase the amount of the Guaranteed Obligations; agree to modify the terms of any one or more of the Loan Documents; extend or renew the Guaranteed Obligations for any period; grant releases, compromises and indulgences with respect to any one or more of the Loan Documents and to any Persons now or hereafter liable thereunder or hereunder; release any other guarantor or endorser of or other Person liable upon the Guaranteed Obligations or any of the Loan Documents; or take or fail to take any action of any type whatsoever. No such action which Lender will take or fail to take in connection with the Loan Documents or any Collateral, nor any course of dealing with the Borrower or any other Person, will limit, impair or release the Guarantor's obligations hereunder, affect this Guaranty in any way or afford the Guarantor any recourse against Lender. Nothing contained in this section will be construed to require Lender to take or refrain from taking any action referred to herein. For clarity, the rights of Lender in this Section 4.5 with respect to Collateral may be subject to the terms and conditions of the Collateral Agency Agreement.

- 4.6 Certain Bankruptcy Matters. The Guarantor covenants and agrees that, upon the commencement of a voluntary or involuntary bankruptcy proceeding by or against the Borrower, the Guarantor will not seek or cause the Borrower or any other Person to seek a supplemental stay or other relief, whether injunctive or otherwise, pursuant to 11 U.S.C. § 105 or any other provision of the federal Bankruptcy Code, as amended, or any other debtor relief law (whether statutory, common law, case law or otherwise), of any jurisdiction whatsoever, now or hereafter in effect, which may be or become applicable, to stay, interdict, condition, reduce or inhibit the ability of Lender to enforce any rights of Lender against the Guarantor or the Collateral by virtue of this Guaranty or otherwise.
- **4.7 Representations and Warranties by Guarantor**. To induce Lender to accept this Guaranty, the Guarantor hereby makes the following representations and warranties:
- **4.7.1** This Guaranty has been duly authorized by all necessary corporate or limited liability company action, as the case maybe, on the part of the Guarantor, and constitutes (and will constitute) legal, valid and binding obligations of the Guarantor enforceable against the Guarantor in accordance with its terms, except as such enforceability may be limited by: (i) applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting the enforcement of creditors' rights generally; and (ii) general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law).
- 4.7.2 The execution, delivery and performance by the Guarantor of this Guaranty will not: (i) result in the creation of any lien in respect of any property of the Guarantor under any indenture, mortgage, deed of trust, security agreement, loan, purchase or credit agreement, lease, corporate charter or by-laws, or any other agreement or instrument by which the Guarantor or any of its properties may be bound or affected, other than a lien in favor of Lender; or (ii) (1) contravene, result in any breach of, or constitute a default under, any indenture, mortgage, deed of trust, security agreement, loan, purchase or credit agreement, lease, corporate charter or by-laws, or any other agreement or instrument by which the Guarantor or any of its properties may be bound or affected; (2) conflict with or result in a breach of any of the terms, conditions or provisions of any order, judgment, decree, or ruling of any court, arbitrator or governmental authority applicable to the Guarantor; or (3) violate any provision of any statute or other rule or regulation of any governmental authority applicable to the Guarantor, in each case to the extent such breach
- **4.7.3** No consent, approval, or authorization of, or registration, filing or declaration with, any governmental authority is required in connection with the execution, delivery, or performance by the Guarantor of this Guaranty.
- **4.7.4** After executing and delivering this Guaranty, (i) the Guarantor will not be insolvent in that the fair saleable value of its assets will be greater than the amount required to pay its total liabilities; (ii) the Guarantor will not have incurred debts and other obligations in excess of its ability to pay such debts and obligations as they mature, and the cash flow of the Guarantor (after taking into account the anticipated uses of the cash) will be sufficient to pay such debts and obligations when such amounts are required to be paid; and (iii) the capital of the

Guarantor will be sufficient and not unreasonably small for the business and transactions in which it is engaged.

5. Reserved.

- **6.** Reservation of Rights. No waiver under this Guaranty is valid unless it is in writing and signed by Lender. A waiver of a particular matter or remedy does not waive a subsequent or similar matter or remedy. No waiver will excuse the Guarantor from the performance of its other obligations to be performed under this Guaranty. The remedies conferred by this Guaranty are nonexclusive and cumulative of any other remedies now available or existing subsequently. No delay or forbearance by or on behalf of Lender in exercising any right, remedy, power or privilege under this Guaranty ("Lender's Rights") will operate as a waiver of such Lender's Rights, nor will the exercise or non-exercise of any particular Lender's Right preclude any other or further exercise of Lender's Rights pertaining to any current or subsequent default by the Guarantor or the Borrower. Nothing contained in this Guaranty will prevent or in any way diminish or interfere with any rights or remedies, including, without limitation, the right to contribution, which Lender may have against the Borrower, the Guarantor or any other party under the Comprehensive Environmental Response, Compensation and Liability Act of 1980 (codified at Title 42 U.S.C. § 9601 et seq.), as it may be amended from time to time, or any other applicable federal, state or local laws, all such rights being hereby expressly reserved.
- 7. Fees and Costs. In addition to the obligations provided above, in the event Lender incurs any cost and expense in connection with the enforcement of this Guaranty, the Guarantor agrees to reimburse Lender for any and all such costs and expenses, including: (i) its reasonable attorneys' fees; (ii) all reasonable costs and expenses of collection or enforcement; and (iii) all reasonable costs of suit, including all proceedings in courts of original, appellate and bankruptcy jurisdiction.
- **8. Notices and Other Communications.** Unless otherwise expressly provided herein, all notices and other communications provided for hereunder shall be in writing and shall be made to the following:

If to Lender, to:

CoBank, ACB 6340 S. Fiddlers Green Circle Greenwood Village, Colorado 80111-1914

Facsimile: (303) 224-6101

Attention: Credit Information Services

If to the Guarantor, to:

c/o Global Water Resources, Inc. Attn: Michael J. Liebman Chief Financial Officer 21410 North 19th Avenue, Suite 220 Phoenix, Arizona 85027-2738

or to such other address, facsimile number, or electronic mail address as shall be designated by such party in a notice to the other party. All such notices and other communications shall be deemed to be given or made upon the earlier to occur of (i) actual receipt by the relevant party hereto and (ii) (A) if delivered by hand or by courier, when signed for by or on behalf of the relevant party hereto; and (B) if delivered by facsimile, when sent and receipt has been confirmed by telephone.

- **9. Construction**. Captions and headings are for convenience and reference only and do not define, limit, or affect the contents of this Guaranty. References to "paragraphs" or "sections" refer to this Guaranty unless stated otherwise. The terms "include" or "including" mean "without limitation by reason of enumeration." All grammatical usage will be deemed to refer to the masculine, feminine, neuter, singular or plural as the context and identity of any persons may require.
- 10. Severability and Interpretation. The invalidity or unenforceability of any provision of this Guaranty does not affect the other remaining provisions. This Guaranty will be construed as if it excluded any invalid or unenforceable provision, which will be severed from this Guaranty. Whenever possible, this Guaranty will be interpreted so as to be valid under applicable law, and will not be construed strictly in favor of or against any particular party, including any party who drafted or prepared this Guaranty, but instead according to its plain meaning to give effect to its intended purposes.
- 11. Governing Law. THIS AGREEMENT AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES HEREUNDER AND THEREUNDER SHALL BE GOVERNED BY AND CONSTRUED AND INTERPRETED IN ACCORDANCE WITH THE LAWS OF THE STATE OF COLORADO. The Guarantor irrevocably consents to the service of process out of any competent court in any action or proceeding with respect to this Agreement by the mailing of copies thereof by registered or certified mail, postage prepaid, to it at the address for notices pursuant to Section 8, such service to become effective 30 days after such mailing. Nothing herein shall affect the right of Lender to serve process in any other manner permitted by law.
- 12. Waiver of Jury Trial. THE GUARANTOR AND LENDER EACH HEREBY IRREVOCABLY WAIVES ALL RIGHT TO TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM ARISING OUT OF OR RELATING TO THIS GUARANTY OR THE TRANSACTIONS CONTEMPLATED HEREBY AND THEREBY.
- **13. Binding Effect**. This Guaranty is binding upon and inures to the benefit of the parties and their respective trustees, debtor(s)-in-possession, receivers, conservators, and the like —if any of the same are ever so appointed, and their successors and assigns.
 - **14. Acceptance**. The Guarantor hereby waives notice of acceptance hereof.

SIGNATURE PAGE FOLLOWS

SIGNATURE PAGE TO GUARANTY AGREEMENT

IN WITNESS WHEREOF, the Guarantor, by its duly authorized officer, has executed this Guaranty.

GLOBAL WATER HOLDINGS, INC.

By: /s/ Michael J. Liebman

Name: Michael J. Liebman

Title: First Vice President, Secretary and Treasurer

PLEDGE AND SECURITY AGREEMENT GLOBAL WATER RESOURCES, INC.

THIS PLEDGE AND SECURITY AGREEMENT (this "Agreement") dated as of December 10, 2025 is made by: (a) GLOBAL WATER RESOURCES, INC., a Delaware corporation ("Pledgor" or "Company"); in favor of (b) U.S. BANK TRUST COMPANY, NATIONAL ASSOCIATION, a national banking association (as successor in interest to U.S. Bank National Association), in its capacity as collateral agent (with its successors and permitted assigns in such capacity the "Collateral Agent"); for the benefit of (c) COBANK, ACB, a federally-chartered instrumentality of the United States (the "Bank" and/or any other holder of any Promissory Note at any relevant time the "Holder"), pursuant to the Third Amended and Restated Collateral Agency Agreement dated as of October 26, 2023 (the "Collateral Agency Agreement") among and/or approved by the Collateral Agent, Company, Noteholders, The Northern Trust Company and Bank.

PRELIMINARY STATEMENT

WHEREAS, subject to the terms and conditions set forth in a Credit Agreement of even date herewith between Company and the Bank (as amended, supplemented or restated at any relevant time the "Credit Agreement") and a Promissory Note by Company in favor of the Bank (as amended, supplemented or restated at any relevant time the "Initial Promissory Note"), the Bank has agreed to provide Company a term loan in the initial principal amount of \$15,000,000 (the "Loan"); and

WHEREAS, Pledgor is the owner and holder of all of the Equity Interests of Global Water, LLC, a Delaware limited liability company (the "**Pledged Company**"); and

WHEREAS, pursuant to Guaranty Agreements (the "Guaranties") of even date herewith, the Pledged Company, West Maricopa Combine, LLC, an Arizona limited liability company, and Global Water Holdings, Inc., an Arizona corporation, have guaranteed the Obligations of Company under the Credit Agreement and the Initial Promissory Note; and

WHEREAS, it is a condition precedent to the obligation of Bank to provide the Loan that Pledgor execute and deliver this Agreement to the Collateral Agent for the benefit of the Secured Parties; and

WHEREAS, Pledgor desires to execute this Agreement to satisfy such condition precedent and to secure all Obligations of Obligors under the Loan Documents;

NOW THEREFORE, to induce the Bank to provide the Loan pursuant to the Credit Agreement and the Initial Promissory Note, Pledgor and the Collateral Agent agree as follows:

SECTION 1. DEFINED TERMS.

- (a) Capitalized terms used at any place but not defined herein have the meanings set forth in the Credit Agreement.
- (b) "ACC Regulations" means the applicable regulations, orders and requirements of the Arizona Corporation Commission and applicable statutes administered by the Arizona Corporation Commission.
- (c) "Applicable Law" means all applicable Laws, including all applicable provisions of constitutions, statutes, rules, ordinances, regulations and orders of all Governmental Authorities and all orders, rulings, writs and decrees of all courts, tribunals and arbitrators.

- (d) "Equity Interests" means with respect to any Person, all capital stock of (or other ownership or profit interest in) such Person, all warrants, options or other rights for the purchase or acquisition from such Person of capital stock of (or other ownership or profit interest in) such Person, all securities convertible into or exchangeable for capital stock of (or other ownership or profit interests in) such Person or warrants, rights or options for the purchase or acquisition from such Person of such shares (or other interests), and all other ownership or profit interests in such Person (including partnership, membership and trust interests therein), voting or nonvoting, whether or not such shares, warrants, options, rights or other interests are outstanding on any date.
- "Indemnified Liabilities" means, collectively, all liabilities, obligations, losses, damages, penalties, claims (including under Environmental Laws), actions, judgments, suits, costs (including reasonable costs of investigation, study, sampling, testing, abatement, cleanup, removal, remediation or other response necessary to remove, remediate, clean up or abate Hazardous Materials), expenses and disbursements of any kind (including reasonable fees and disbursements of counsel for Indemnitees in connection with any investigative, administrative or judicial proceeding or hearing commenced or threatened by any Person, whether or not such Indemnitees are designated as a party or potential party thereto, and any reasonable fees or expenses incurred by Indemnitees in enforcing this indemnity), whether direct or indirect and whether based on any Applicable Laws (including Environmental Laws), on common law or equitable cause or on contract or otherwise, imposed on, incurred by, or asserted against any such Indemnitee (regardless of whether any Indemnitee is a party thereof), in any manner relating to or arising out of: (i) this Agreement or any other Loan Document or transactions contemplated hereby or thereby (including the use or intended use of the proceeds of the Loan, or any enforcement of any of the Loan Documents (including any sale of, collection from, or other realization on any Collateral)); or (ii) any claim under Environmental Laws or Hazardous Materials liability relating to or arising from, directly or indirectly, any past or present activity, operation, land ownership or practice of Pledgor or any of its Subsidiaries.
 - (f) "**Obligors**" means the Borrower and the Subsidiary Guarantors.
 - (g) "Pledged Company" has the meaning set forth in the above preliminary statement.
 - (h) "Secured Obligations" has the meaning provided by Section 2(c).
 - (i) "Secured Parties" means all Holders and the Collateral Agent.
- (j) "Specified Account" means the segregated account of Company maintained with The Northern Trust Company, with the last four digits of such account number being 2231, established to receive payments of dividends and distributions on Equity Interests owned by Company or any Subsidiary Guarantor, and all replacements or substitutions for such account established by the Company or any Subsidiary Guarantor, whether in the form of a deposit account or securities account. For the avoidance of doubt, a "Specified Account" shall not include any "collection account," deposit account, securities account or other account in which the revenues of Utility Subsidiaries are remitted or consolidated.
- (k) "UCC" means the Uniform Commercial Code as in effect on the date hereof in the State of New York; provided that if by mandatory provisions of Law, the perfection or the effect of perfection or non-perfection of the security interests granted pursuant to Section 2, as well as all other security interests created or assigned as additional security for the Secured Obligations pursuant to this Agreement, in any Collateral is governed by the UCC as in effect in any jurisdiction other than the State of New York, "UCC" means the UCC as in effect in such other jurisdiction for purposes of the provisions hereof relating to such perfection or effect of perfection or non-perfection.

SECTION 2. GRANT OF SECURITY INTEREST.

- (a) To secure the Secured Obligations, Pledgor grants to the Collateral Agent, for the benefit of the Secured Parties, a security interest in, and acknowledges and agrees that the Collateral Agent has and will continue to have a continuing security interest in, all right, title and interest of Pledgor, whether now owned or existing or hereafter created, acquired or arising, and regardless of where located, in and to all the following (collectively the "Collateral"):
- (i) all Equity Interests, including all Equity Interests in the Pledged Company and all shares, ownership, economic and management interests, membership interests and/or partnership interests in any Person owned or held by Pledgor, all payments and distributions of whatever kind or character, in cash or other property, at any time made, owing or payable to Pledgor in respect of or on account of its present or hereafter acquired Equity Interests, whether due or to become due and whether representing profits, distributions pursuant to complete or partial liquidation or dissolution of the issuer of such Equity Interests, distributions representing the complete or partial redemption of Pledgor's Equity Interests in any Person or complete or partial withdrawal of Pledgor from any Person, repayment of capital contributions made to or with respect to any Person in respect of Equity Interests in such Person held by Pledgor and the right to receive, receipt for, use, and enjoy all such payments and distributions, and all other rights and privileges incident to Pledgor's interest in such Equity Interests, provided that prior to the occurrence of an Event of Default, Pledgor shall retain certain rights pursuant to Section 7;

(ii) the Specified Account;

- (iii) all interest, dividends, cash, instruments, investment property, general intangibles and other property from time to time received, receivable or otherwise payable in respect of, or in exchange for, any of the foregoing;
- (iv) supporting evidence and documentation relating to any of the above-described property, including computer programs, disks, tapes, electronic archives, clouds and related data processing media, and all rights of Pledgor to retrieve the same from third parties, written applications, credit information, account cards, payment records, correspondence, delivery and installation certificates, invoice copies, delivery receipts, notes, and other evidences of indebtedness, insurance certificates and the like, together with all books of account, ledgers and cabinets in which the same are reflected or maintained; and
- (v) to the extent not covered by Sections 2(a)(i) through 2(a)(iv), all proceeds (as defined in the UCC) of any or all of the foregoing.
- (b) The Collateral Agent shall have with respect to the Collateral, in addition to the rights and remedies set forth herein and in the other Loan Documents, all rights and remedies of a secured party under the UCC as if fully set forth herein.
- (c) The security interest herein granted is made and given to secure, and shall secure, the payment and performance of: (i) all Obligations of any of Obligors to the Secured Parties (whether arising before or after the filing of a petition in bankruptcy), whether direct or indirect, absolute or contingent, due or to become due, now existing or hereafter arising, however held, evidenced or acquired, and whether several, joint, or joint and several; and (ii) all expenses and charges, legal or otherwise, suffered or incurred by the Secured Parties in collecting or enforcing any such Obligations or realizing on or protecting or preserving any Collateral or other security therefor, including the lien and security interest granted hereby (all of the foregoing being the "Secured Obligations").
- (d) For the avoidance of doubt and notwithstanding anything in any Loan Document to the contrary: (i) no Subsidiary of Pledgor that is now a regulated utility is a borrower or

guarantor under any Loan Document, nor is any such Subsidiary pledging any of its property as collateral for the Secured Obligations; and (ii) no regulated utility may declare distributions or dividends to its equity holders except in accordance with Applicable Law (including ACC Regulations), and subject to each regulated utility's obligations to maintain revenues and funds sufficient to fund direct and indirect operating and maintenance expenses (including general and administrative expenses and reasonable and necessary costs, fees and expenses for operation and maintenance of system utilities).

SECTION 3. CERTIFICATES OR INSTRUMENTS. Any certificates or instruments representing or evidencing Collateral shall be delivered to and held by or on behalf of the Collateral Agent for the benefit of the Secured Parties pursuant hereto and to the Collateral Agency Agreement and be in suitable form for transfer by delivery or accompanied by duly executed instruments of transfer or assignment in blank, all in form reasonably satisfactory to the Collateral Agent and in form and substance reasonably satisfactory to the Bank. Upon the occurrence and during the continuance of any Event of Default, the Collateral Agent will have the right at any time in its sole discretion and without notice to Pledgor to transfer to or to register in the name of the Collateral Agent or its nominees any or all of the Collateral, subject only to the revocable rights specified in Section 7(a) and compliance with ACC Regulations. In addition, on the occurrence and during the continuance of any Event of Default, the Collateral Agent will have the right to exchange certificates or instruments representing or evidencing Collateral for certificates or instruments of smaller or larger denominations.

SECTION 4. SPECIFIED ACCOUNT COLLATERAL.

- The Company has established the Specified Account as a deposit account (within the meaning of Section 9-102 of the UCC), with The Northern Trust Company. Pledgor shall promptly notify the Collateral Agent of any other Specified Account opened or maintained by the Company after the date hereof. To the extent requested by the Collateral Agent following the occurrence and continuation of any Default or the occurrence of any Event of Default. Pledgor shall, and shall cause the applicable depository or other institution to, execute and deliver to the Collateral Agent an account control agreement in form reasonably satisfactory to the Collateral Agent and in form and substance reasonably satisfactory to the Holder which provides among other things for the depository or other institution's agreement that it will comply with: (i) instructions originated by the Collateral Agent directing the disposition of the funds in each such Specified Account that is a deposit account; or (ii) entitlement orders originated by the Collateral Agent with respect to each such Specified Account that is a securities account, in each case without further consent by Pledgor. Notwithstanding the foregoing, any account control agreement which requires the Collateral Agent in its individual capacity to indemnify the depository or other institution other than out of the Collateral will not be reasonably satisfactory to the Collateral Agent.
- (b) Subject to Section 2(d), Pledgor covenants and agrees that upon declaration and payment of any dividend or distribution by any Subsidiary, it will cause such dividend or distribution to be paid and segregated into the Specified Account. Prior to the occurrence and continuance of an Event of Default, but subject to the limitation on Restricted Payments by Pledgor in the Credit Agreement, amounts so deposited in the Specified Account may be withdrawn and used for corporate purposes, including investments in or loans to Subsidiaries permitted under the Credit Agreement.

SECTION 5. REPRESENTATIONS, WARRANTIES AND CERTAIN COVENANTS. Pledgor represents, warrants and covenants as follows:

(a) The Equity Interests have been duly authorized and validly issued and are fully paid and non-assessable. No Equity Interests constituting interests in limited liability companies

constitute or are evidenced by certificated securities, unless such certificates have been delivered to the Collateral Agent.

- (b) Pledgor is the legal and beneficial owner of the Collateral free and clear of any Lien or other encumbrance except for the Lien created by this Agreement and any other Liens created in favor of the Collateral Agent and described in the Collateral Agency Agreement. There is no existing agreement, option, right or privilege capable of becoming an agreement or option pursuant to which Pledgor would be required to sell or otherwise dispose of any Equity Interest, except as described in the Collateral Agency Agreement.
- (c) Except for the delivery of any certificates or instruments representing Collateral to the Collateral Agent pursuant to this Agreement, filing of an appropriate financing statement with the Delaware Secretary of State, and any control agreement contemplated by Section 4(a), no other action is required to create or maintain the Lien of the Collateral Agent as a valid and perfected first priority Lien (pari passu with the other Secured Parties (as defined in the Collateral Agency Agreement)) in the Collateral to secure the Secured Obligations.
- (d) No authorization, approval or other action by, and no notice to or filing with, any Governmental Authority or regulatory body (except as set forth in Section 5(c)) is required either: (i) for the pledge by Pledgor of the Collateral pursuant to this Agreement or for the execution, delivery or performance of this Agreement by the Pledgor; or (ii) for the exercise by the Collateral Agent of its rights and the rights of any other Secured Parties provided for in this Agreement and the Collateral Agency Agreement or the remedies in respect of the Collateral pursuant to this Agreement and the Collateral Agency Agreement (except as may be required in connection with such disposition by Laws affecting the offering and sale of securities generally and except for compliance with requirements of the ACC Regulations as set forth in Sections 7(b) and 8).
- (e) The execution, delivery and performance of this Agreement does not and will not: (i) violate any provision of any Law (including Regulations T, U and X of the Board of Governors of the Federal Reserve System), order, writ, judgment, injunction, decree, determination or award applicable to Pledgor; (ii) result in breach of, or constitute a default under, any indenture, credit or loan or note agreement or any other agreement, lease or instrument to which Pledgor presently is a party, or by which it or its properties are bound or affected; or (iii) result in or require (other than pursuant to this Agreement) the creation or imposition of any Lien or other share or encumbrance upon or with respect to any properties now owned or hereafter acquired by Pledgor. Pledgor is not in violation of or default under any such Law or material provision of any such indenture, agreement, lease or instrument.
- (f) All Equity Interests in the Pledged Company are owned by Pledgor and pledged by this Agreement.
- (g) Except for any such agreements in favor of the Collateral Agent for the benefit of the Secured Parties and the other Secured Parties (as defined in the Collateral Agency Agreement), Pledgor shall not enter into any agreement providing any Person with "control" (within the meaning of Sections 9-104 or 9-106 of the applicable UCC) of any Specified Account.
- (h) Pledgor is a corporation duly formed under the laws of Delaware, and is validly existing and in good standing under the laws of such jurisdiction and the State of Arizona. Pledgor has its chief executive office in the State of Arizona. Pledgor agrees to give the Collateral Agent at least 30 days' prior written notice before changing the state in which its chief executive office is located or the state in which it is organized, and prior to the effectiveness of any such change shall take all steps necessary to maintain the security interest provided for herein as a first-priority (pari passu with the other Secured Parties (as defined in the Collateral

Agency Agreement)) perfected security interest, including filing additional UCC financing statements or amendments as may be necessary or requested by (and subject to the rights of) the Collateral Agent.

- (i) Pledgor shall not: (i) sell, assign (by agreement, operation of law or otherwise) or otherwise dispose of, or grant any option with respect to, any Collateral; or (ii) create or permit to exist any Lien or security interest on or with respect to any Collateral, except for the Lien created by this Agreement and any other Lien in favor of the Collateral Agent for the benefit of the Secured Parties.
- (j) Pledgor shall not permit or cause to be issued any Equity Interests: (i) in substitution for any existing Equity Interests; and (ii) in addition to the existing Equity Interests, except following notice to the Collateral Agent and provided that immediately upon its acquisition (directly or indirectly) of any such substitute or additional securities, Pledgor will execute such documentation as is necessary to pledge or evidence the pledge or as may be requested by Collateral Agent or the Holder pledging, and evidencing the pledge hereunder of, such Equity Interests.
- (k) Pledgor shall, at its expense, protect and defend this Agreement, all rights of the Collateral Agent hereunder, and the Collateral against all claims and demands of other parties. Pledgor shall pay all claims and charges that in the reasonable opinion of the Collateral Agent or the Holder might prejudice, imperil or otherwise affect Collateral or the security interest therein. Pledgor shall promptly notify the Collateral Agent of any levy, distraint or other seizure by legal process or otherwise of any Collateral and of any threatened or filed claims or proceedings that might affect or impair this Agreement.
- SECTION 6. FURTHER ASSURANCES. Pledgor authorizes the Collateral Agent to file any financing statements covering the Collateral or any part thereof as the Collateral Agent may desire. Pledgor agrees that from time to time, at the expense of Pledgor, Pledgor will promptly execute and deliver all further documentation and take all further actions that the Collateral Agent or Holder may reasonably request, in order to perfect and protect any security interest granted or purported to be granted hereby or to enable the Collateral Agent to exercise and enforce its rights and remedies hereunder with respect to any Collateral. Pledgor will furnish to the Collateral Agent from time to time statements and schedules further identifying and describing the Collateral and such other reports in connection with the Collateral as the Collateral Agent or Holder may reasonably request, all in reasonable detail, Pledgor agrees, and agrees to cause each issuer of Equity Interests included in the Collateral, to mark its books and records to reflect the Lien of the Collateral Agent in the Collateral.

SECTION 7. VOTING AND DIVIDENDS.

- (a) So long as no Event of Default has occurred and is continuing:
- (i) Pledgor will be entitled to exercise any voting and other consensual rights pertaining to any Collateral for any purpose not inconsistent with this Agreement or any other Loan Documents.
- (ii) Pledgor will be entitled to receive and retain any dividends or distributions paid in respect of the Collateral; provided however, except as expressly permitted by the Credit Agreement, any (A) dividends or distributions paid or payable other than in cash in respect of, and instruments and other property received, receivable or otherwise distributed in respect of, or in exchange for, any Collateral, (B) dividends and other distributions paid or payable in cash in respect of any Collateral in connection with a partial or total liquidation or dissolution, and (C) cash paid, payable or otherwise distributed in redemption of, or in exchange for, any Collateral, shall be forthwith delivered to the Collateral Agent to hold as Collateral and shall, if received by

Pledgor, be received in trust for the benefit of the Collateral Agent, be segregated from the other property or funds of Pledgor, and be forthwith delivered to the Collateral Agent as Collateral in the same form as so received (with any necessary endorsement).

- (iii) The Collateral Agent shall execute and deliver (or cause to be executed and delivered) to Pledgor all such proxies and other instruments as Pledgor may reasonably request for the purpose of enabling Pledgor to exercise the voting and other rights it is entitled to exercise pursuant to Section7(a)(i) and to receive dividends it is authorized to receive and retain pursuant to Section7(a)(ii).
- (b) Subject to any requirements of ACC Regulations, on the occurrence and during the continuation of an Event of Default:
- (i) All rights of Pledgor to exercise voting and other consensual rights it would otherwise be entitled to exercise pursuant to Section 7(a)(i) will automatically cease, and the Collateral Agent will thereupon have the sole right to exercise such rights.
- (ii) All rights of Pledgor to receive the distributions and dividends it would otherwise be entitled to receive and retain pursuant to Section 7(a)(ii) will automatically cease, and the Collateral Agent will thereupon have the sole right to receive and hold as Collateral such dividends, distributions and interest.
- (iii) Any distributions and dividends received by Pledgor contrary to the provisions of Section 7(b)(ii) will be received in trust for the benefit of the Collateral Agent on behalf of the Secured Parties, shall be segregated from other funds of Pledgor and shall be forthwith paid over to the Collateral Agent on behalf of the Secured Parties as Collateral in the same form as received (with any necessary endorsement).

SECTION 8. REMEDIES. Subject to any requirements of ACC Regulations:

- (a) The Collateral Agent may exercise in respect of any Collateral, in addition to other rights and remedies provided for herein or otherwise available to it, all rights and remedies of a secured party on default under the UCC (whether or not the UCC applies to the affected Collateral), and the Collateral Agent may also, without notice except as specified below, sell the Collateral or any part thereof in one or more parcels at public or private sale, at any exchange, broker's board or at any of the Collateral Agent's offices or elsewhere, for cash, on credit or for future delivery, and on such other terms as the Collateral Agent may claim commercially reasonable. Pledgor agrees that at least ten days' notice to Pledgor of the time and place of any public sale or time after which any private sale is to be made will constitute reasonable notification. The Collateral Agent will not be obligated to make any sale of Collateral regardless of notice of sale having been given. The Collateral Agent may adjourn any public or private sale from time to time by announcement at the time and place fixed therefor, and such sale may, without further notice, be made at the time and place to which it was so adjourned.
- (b) Any cash held by the Collateral Agent as Collateral and all cash proceeds received by the Collateral Agent in respect of any sale of, collection from, or other realization on any Collateral shall be applied by the Collateral Agent against the Secured Obligations in such order as the Collateral Agent may elect, subject to the requirements of the Credit Agreement and Collateral Agency Agreement. Any surplus of such cash or cash proceeds held by the Collateral Agent and remaining after payment in full of all Secured Obligations and all other Secured Obligations (as defined in the Collateral Agency Agreement) shall be paid over to the Pledgor or to whomsoever may be lawfully entitled to receive such surplus.

- (c) All rights and remedies of the Collateral Agent expressed herein are in addition to all other rights and remedies possessed by the Collateral Agent or the Holder in the Loan Documents and any other agreement or instrument relating to the Secured Obligations.
- (d) In connection with a public or private sale of any Collateral, the Collateral Agent may disclose to prospective purchasers any non-public information available to the Collateral Agent which pertains to: (i) the issuer of any Collateral; or (ii) Pledgor, provided in the case of Pledgor such non-public information is material to said issuer, its financial condition or the Collateral.
- (e) Without in any way limiting the foregoing, on the occurrence and during the continuation of any Event of Default, the Collateral Agent shall have the right, in addition to all other rights provided herein or by Law, to direct the disposition of the funds in any Specified Account that is a deposit account, or entitlement orders originated by the secured party with respect to each such Specified Account that is a securities account, in each case without further consent by Pledgor.
- (f) If the Collateral Agent exercises its right to take possession of any Collateral, Pledgor shall also at its expense perform any other steps requested by the Collateral Agent or Holder to preserve and protect the security interest hereby granted in the Collateral, such as maintaining Collateral records and filing UCC financing and continuation statements.

SECTION 9. WAIVERS; PRIVATE SALES.

- Pledgor waives any right to require the Collateral Agent or Holder to: (i) proceed against any Person, including any other Obligor; or (ii) proceed against or exhaust any Collateral, or (iii) pursue any other remedy in Collateral Agent's or the Holder's power; and waives any defense arising by reason of any disability of Pledgor or any other Person. Until the Secured Obligations have been paid in full, Pledgor waives any right of subrogation, reimbursement, indemnification, and contribution (contractual, statutory or otherwise), including any right of subrogation under the Bankruptcy Code (Title 11, United States Code) or any successor statute, arising from the existence or performance of this Agreement, and Pledgor waives any right to enforce any remedy which the Collateral Agent now has or may hereafter have against Pledgor or any other Person, and waives any benefit of, and any right to participate in, any security now or hereafter held by the Collateral Agent. Pledgor authorizes the Collateral Agent, without notice or demand and without affecting Pledgor's liability hereunder, from time to time to: (a) renew, compromise, extend, accelerate or otherwise change the time for payment of, or otherwise change the terms of, any Secured Obligations, including increase or decrease of any rate of interest thereon; (b) receive and hold security, other than the Collateral herein described, for the payment of any Secured Obligations, and exchange, enforce, waive, release, fail to perfect, sell, or otherwise dispose of any Collateral or other security; and (c) release or substitute the Company, or any other guarantors of such Secured Obligations or any part thereof, or any other parties thereto.
- (b) Pledgor recognizes that the Collateral Agent may be unable to effect a public sale of any of the Collateral by reason of ACC Regulations and/or certain prohibitions contained in the Laws of any jurisdiction outside the United States or in the Securities Act and applicable state securities Laws, but may instead be compelled to resort to one or more private sales thereof to a restricted group of purchasers who will be obliged to agree, among other things, to acquire such Collateral for their own account for investment and not with a view to the distribution or resale thereof, or otherwise in accordance with the ACC Regulations. Pledgor agrees that any private sale may result in prices and other terms less favorable to the seller than if such sale were a public sale and, notwithstanding such circumstances, agrees that any such private sale shall, to the extent permitted by Law, be made in a commercially reasonable manner. The Collateral Agent will not be under any obligation to delay a sale of any Collateral for the period of time

necessary to permit the issuer of such securities to register such securities under the Laws of any jurisdiction outside the United States, under the Securities Act or under any applicable state securities Laws, even if the issuer would agree to do so. If the Collateral Agent is able to lawfully effect a public sale without registration of the Collateral under the laws of any jurisdiction outside the United States, under the Securities Act or under any applicable state securities Laws, then, subject to any applicable ACC Regulations, the Collateral Agent may, but will not be required to, conduct a public sale of the Collateral, rather than a private sale, if the Collateral Agent reasonably believes it would realize a higher sales price in a public sale.

SECTION 10. COLLATERAL AGENT'S DUTIES.

- (a) The powers conferred on the Collateral Agent hereunder are solely to protect the interests of the Secured Parties in the Collateral and shall not impose any duty on the Collateral Agent to exercise any such powers. Except for reasonable care in the custody of any Collateral in its possession and accounting for moneys actually received by it hereunder, the Collateral Agent will have no duty as to any Collateral or as to the taking of any necessary steps to preserve rights against prior parties or any other rights pertaining to any Collateral. The Collateral Agent will have exercised reasonable care in the custody and preservation of Collateral in its possession if the Collateral is accorded treatment substantially equal to that which the Collateral Agent accords its own property and collateral held for others in its capacity as a collateral agent, it being understood that the Collateral Agent will not have any responsibility for: (a) ascertaining or taking action with respect to calls, conversions, exchanges, maturities, tenders or other matters relative to any Collateral, whether or not the Collateral Agent has or is deemed to have knowledge of such matters; or (b) taking any necessary steps to preserve rights against any parties with respect to any Collateral.
- (b) Upon the appointment of a replacement collateral agent pursuant to the Collateral Agency Agreement, the Collateral Agent may transfer all of its interest and Liens in or any of the Collateral and be fully discharged thereafter from all liability and responsibility with respect to such Collateral so transferred, and the transferee shall be vested with all the rights and powers of the Collateral Agent hereunder with respect to such Collateral so transferred; but with respect to any of its interest and liens in Collateral not so transferred, the Collateral Agent will retain all rights and powers hereby given.

SECTION 11. OTHER RIGHTS.

- (a) The rights, powers and remedies given to the Collateral Agent by this Agreement shall be in addition to all rights, powers and remedies given to the Collateral Agent by virtue of any statute or rule of Law. Any forbearance or failure or delay by the Collateral Agent in exercising any right, power or remedy hereunder shall not be a waiver of such right, power or remedy, and any single or partial exercise of any right, power or remedy hereunder shall not preclude the further exercise thereof; and every right, power and remedy of the Collateral Agent shall continue in full force and effect until such right, power or remedy is specifically waived by a writing executed by the Collateral Agent. The rights and remedies of the Collateral Agent under this Agreement shall be cumulative and not exclusive of any other right or remedy which the Collateral Agent or any Holder may have.
- (b) This Agreement constitutes an assignment of rights only and not an assignment of any duties or obligations of Pledgor in any way related to the Collateral, and the Collateral Agent will have no duty or obligation to discharge any such duty or obligation. The Collateral Agent will have no responsibility for taking any necessary steps to preserve rights against any parties with respect to any Collateral or initiating any action to protect any Collateral against the possibility of a decline in market value. Neither the Collateral Agent nor any party acting as attorney for the Collateral Agent will be liable for any acts or omissions or for any error of

judgment or mistake of fact or Law other than its gross negligence or willful misconduct or negligence in the handling of funds.

In addition to any other powers of attorney contained herein, Pledgor hereby appoints the Collateral Agent, its nominee, and any other Person whom the Collateral Agent may designate, as Pledgor's attorney-in-fact, with full power and authority to sign Pledgor's name on verifications of Collateral; to send requests for verification of Collateral to obligors; to endorse Pledgor's name on any checks, notes, acceptances, money orders, drafts and other forms of payment or security that may come into the Collateral Agent's possession or on any assignments, stock powers or other instruments of transfer relating to any Collateral; to sign Pledgor's name on claims to enforce collection of any Collateral, on notices to and drafts against obligors, on schedules and assignments of Collateral, on notices of assignment and on public records; on the occurrence and during the continuance of an Event of Default to notify post office authorities to change the address for delivery of Pledgor's mail to an address designated by the Collateral Agent; upon the occurrence and during the continuance of an Event of Default to receive, open and dispose of all mail addressed to Pledgor; and to do all things necessary to carry out this Agreement. Pledgor hereby ratifies and approves all acts of any such attorney and agrees that neither the Collateral Agent nor any such attorney will be liable for any acts or omissions nor for any error of judgment or mistake of fact or Law other than such Person's gross negligence or willful misconduct or negligence in the handling of funds. The foregoing powers of attorney, being coupled with an interest, are irrevocable until the Secured Obligations have been fully paid and satisfied and all agreements of any Secured Party to extend credit to or for the account of Pledgor have expired or otherwise been terminated.

SECTION 12. INDEMNITY; WAIVER.

- Pledgor agrees to indemnify, pay and hold harmless, the Collateral Agent and each other Secured Party and any of their Related Parties (each an "Indemnitee"), from and against any and all Indemnified Liabilities, IN ALL CASES, WHETHER OR NOT CAUSED BY OR ARISING, IN WHOLE OR IN PART, OUT OF THE COMPARATIVE OR CONTRIBUTORY NEGLIGENCE OF SUCH INDEMNITEE; provided, Pledgor shall not have any obligation to any Indemnitee hereunder with respect to any Indemnified Liabilities to the extent such Indemnified Liabilities (x) arise from the gross negligence or willful misconduct, as determined by a court of competent jurisdiction by final and nonappealable judgment, of that Indemnitee, or (y) result from a claim brought by Pledgor against an Indemnitee for breach in bad faith of such Indemnitee's obligations hereunder or under any other Loan Document, as determined by a court of competent jurisdiction by final and nonappealable judgment. To the extent the undertakings to indemnify, pay and hold harmless in this Section 12 may be unenforceable in whole or in part because they are violative of any Law or public policy, Pledgor shall contribute the maximum portion it is permitted to pay and satisfy under Applicable Laws to the payment and satisfaction of all Indemnified Liabilities incurred by any Indemnitees. All amounts due under this Section 12(a) shall be payable promptly after demand therefor. For purposes hereof, "Related Parties" means, with respect to any Person, such Person's Affiliates and the partners, managers, directors, trustees, officers, employees or other personnel, counsel, agents and advisors of such Person and of such Person's Affiliates.
- (b) To the extent not prohibited by Applicable Law, Pledgor shall not assert, and Pledgor waives, any claim against the Collateral Agent and its respective Related Parties, on any theory of liability, for special, indirect, consequential or punitive damages (as opposed to direct or actual damages) (whether or not the claim therefor is based on contract, tort or duty imposed by any applicable legal requirement) arising out of, in connection with, as a result of, or in any way related to, this Agreement or any other Loan Document or any agreement or instrument contemplated hereby or thereby or referred to herein or therein, the transactions contemplated hereby or thereby, or the use of the proceeds thereof or any act or omission or event occurring in

10

connection therewith, and Pledgor waives, releases and agrees not to sue on any such claim or any such damages, whether or not accrued and whether or not known or suspected to exist in its favor. No Indemnitee shall be liable for any damages arising from the use by unintended recipients of any information or other materials distributed to it through telecommunications, electronic or other information transmission systems in connection with this Agreement or any other Loan Documents or the transactions contemplated hereby or thereby. In no event shall the Collateral Agent be responsible or liable for any failure or delay in the performance of its obligations hereunder arising out of or caused by; directly or indirectly, forces beyond its control, including strikes, work stoppages, accidents, acts of war or terrorism, civil or military disturbances, nuclear or natural catastrophes or acts of God, and interruptions, loss or malfunctions of utilities, communications or computer (software and hardware) services, it being understood that the Collateral Agent shall use reasonable best efforts which are consistent with accepted practices in the banking industry to resume performance as soon as practicable under the circumstances.

SECTION 13. INTERPRETATION.

- (a) In this Agreement, unless a clear contrary intention appears:
 - (i) the singular number includes the plural number and vice versa;
 - (ii) reference to any gender includes each other gender;
- (iii) the words "herein," "hereof" and "hereunder" and words of similar import refer to this Agreement as a whole and not to any particular Article, Section or other subdivision;
- (iv) reference to any Person includes such Person's successors and assigns but, if applicable, only if such successors and assigns are permitted by this Agreement, and reference to a Person in a particular capacity excludes such Person in any other capacity or individually, provided that nothing in this Section 13(a)(iv) is intended to authorize any assignment not otherwise permitted by this Agreement;
- (v) reference to any agreement, document or instrument means such agreement, document or instrument as amended, supplemented or modified and in effect from time to time in accordance with the terms thereof and, if applicable, the terms hereof, and reference to any note includes any note issued pursuant to any Loan Document in extension or renewal thereof and in substitution or replacement therefor;
- (vi) unless the context indicates otherwise, reference to any Article, Section or Schedule means such Article or Section hereof or Schedule hereto;
- (vii) "including" (with its correlative meaning "include") means including, without limiting the generality of any description preceding such term;
- (viii) with respect to the determination of any period of time, the word "from" means "from and including" and the word "to" means "to but excluding;"
- (ix) reference to any Law means such as modified, codified or reenacted, in whole or in part, and in effect from time to time; and
- (x) references to the Collateral Agent are to it in its capacity as Collateral Agent for the Secured Parties and the other Secured Parties (as defined in the Collateral Agency Agreement) under and as provided in the Collateral Agency Agreement.
- (b) Article and Section headings herein are for convenience only and shall not affect the construction hereof.

- (c) No provision of this Agreement shall be interpreted or construed against any Person because that Person or its legal representative drafted such provision.
- SECTION 14. AMENDMENTS. No amendment or waiver of any provision of this Agreement, nor consent to any departure by Pledgor herefrom, will be effective unless in writing and signed by Pledgor and the Collateral Agent, and then only in the specific instance and for the specific purpose for which given.
- SECTION 15. NOTICES. All notices and other communications provided for hereunder shall be in writing and sent: (a) by telecopy, facsimile or electronic mail if the sender on the same day sends a confirming copy of such notice by a nationally recognized overnight delivery service (charges prepaid); (b) by registered or certified mail with return receipt requested (postage prepaid); or (c) by a nationally recognized overnight delivery service (with charges prepaid). Any such notice must be sent:
- (i) if to the Bank or its nominee, to the Bank or nominee at the address specified for such communications in the Credit Agreement, or at such other address as the Bank or nominee shall have specified to the Company in writing;
- (ii) if to any other Holder, to such Holder at such address as such Holder shall have specified to the Company in writing;
- (iii) if to Pledgor, to Pledgor at the following address: 21410 North 19th Avenue, Suite 220, Phoenix, Arizona 85027 to the attention of: Michael J. Liebman, or at such other address as Pledgor shall have specified to the Collateral Agent and each Holder in writing; or
- (iv) if to the Collateral Agent, at 333 Commerce Street, Suite 900, Nashville, Tennessee 37201, Attention: W. Jones (Global Water Resources, Inc.), wally.jones@usbank.com, or at such other address as the Collateral Agent shall have specified to Pledgor in writing.

Notices under this Section 15 will be effective only when actually received.

SECTION 16. SEPARABILITY. If any clause, sentence, paragraph, subsection or Section of this Agreement is judicially declared to be invalid, unenforceable or void, such decision will not have the effect of invalidating or voiding the remainder of this Agreement, and the part or parts of this Agreement so held to be invalid, unenforceable or void will be deemed to have been stricken herefrom by the parties hereto, and the remainder of this Agreement will have the same force and effect as if such stricken part or parts had never been included herein

SECTION 17. COUNTERPARTS. This Agreement may be executed in any number of counterparts and by different parties in separate counterparts, each of which when so executed will be an original and all of which taken together will constitute one and the same Agreement. Signatures of parties transmitted by facsimile or electronic transmission will be their original signatures for all purposes.

SECTION 18. CONTINUING LIEN. This Agreement shall create a continuing security interest in the Collateral and shall: (a) remain in effect until payment in full of the Secured Obligations; (b) be binding on Pledgor, its successors and assigns; and (c) inure to the benefit of the Collateral Agent and other Secured Parties and their successors, transferees and assigns. Without limiting the foregoing clause (c), any Holder may assign or otherwise transfer all or a portion of its interests, rights and obligations under any Promissory Note in accordance with the Loan Documents. Upon the payment in full of the Secured Obligations and the other Secured Obligations (as defined in the Collateral Agency Agreement), Pledgor will be entitled to the return, on its request and at its expense, of such Collateral as shall not have been disposed of.

SECTION 19. SURVIVAL. All representations and warranties in this Agreement or made in writing by Pledgor in connection herewith will survive the execution and delivery of this Agreement and repayment of the Secured Obligations. Any investigation by any Secured Party will not diminish in any respect its right to rely on such representations and warranties.

SECTION 20. LAW. THIS AGREEMENT AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES HEREUNDER SHALL BE GOVERNED BY, AND SHALL BE CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK WITHOUT REGARD TO CONFLICT OF LAWS PRINCIPLES THEREOF THAT WOULD REQUIRE OR PERMIT THE APPLICATION OF THE LAWS OF ANOTHER JURISDICTION EXCEPT TO THE EXTENT THE PERFECTION OF THE SECURITY INTEREST HEREUNDER, OR REMEDIES HEREUNDER, IN RESPECT OF ANY PARTICULAR COLLATERAL ARE GOVERNED BY THE LAWS OF A JURISDICTION OTHER THAN THE STATE OF NEW YORK.

SECTION 21. JURISDICTION; WAIVER OF JURY TRIAL.

- (a) Pledgor irrevocably submits to the non-exclusive jurisdiction of any New York State or federal court sitting in the Borough of Manhattan, The City of New York, over any suit, action or proceeding arising out of or relating to the Loan Documents. To the fullest extent permitted by Applicable Law, Pledgor irrevocably waives and agrees not to assert, by way of motion, as a defense or otherwise, any claim that it is not subject to the jurisdiction of any such court, any objection that it may now or hereafter have to the laying of the venue of any such suit, action or proceeding brought in any such court and any claim that any such suit, action or proceeding brought in any such court has been brought in an inconvenient forum.
- (b) Pledgor consents to process being served by or on behalf of the Collateral Agent or any Holder in any suit, action or proceeding of the nature referred to in Section 21(a) by hand delivery, reputable overnight commercial delivery service or by mailing a copy thereof by registered or certified or express mail (or any substantially similar form of mail), postage prepaid; return receipt requested, in each case to it at its address described in Section 15 or at such other address of which the Collateral Agent or such Holder shall then have been notified pursuant to Section 15. Pledgor agrees that such service upon receipt: (i) will be in every respect effective service of process on it in any such suit, action or proceeding; and (ii) will, to the fullest extent permitted by Applicable Law, be valid personal service on and personal delivery to it. Notices hereunder will be conclusively presumed received as evidenced by a delivery receipt furnished by the United States Postal Service or any reputable commercial delivery service.
- (c) Nothing in this Section 21 shall affect the right of the Collateral Agent or any Holder to serve process in any manner permitted by Law, or limit any right the Collateral Agent or Holder may have to bring proceedings against Pledgor in the courts of any appropriate jurisdiction or to enforce in any lawful manner a judgment obtained in one jurisdiction in any other jurisdiction.
- (d) THE PARTIES HERETO HEREBY WAIVE TRIAL BY JURY IN ANY ACTION BROUGHT ON OR WITH RESPECT TO OR RELATED TO ANY LOAN DOCUMENT.

SECTION 22. FINAL AGREEMENT. THIS AGREEMENT AND ANY OTHER LOAN DOCUMENTS EXECUTED IN CONNECTION HEREWITH REPRESENT (WITH THE COLLATERAL AGENCY AGREEMENT) THE FINAL AGREEMENT BETWEEN THE PARTIES AND MAY NOT BE CONTRADICTED BY EVIDENCE OF ANY PRIOR, CONTEMPORANEOUS, OR SUBSEQUENT ORAL AGREEMENTS OF THE PARTIES. THERE ARE NO UNWRITTEN ORAL AGREEMENTS BETWEEN THE PARTIES.

SECTION 23. LLC INTERESTS. Pledgor represents and warrants that: (a) any Pledged Company that is a limited liability company has not issued any certificate or other instrument to evidence any of Pledgor's limited liability company interests in such Pledged Company; (b) the limited liability company interest in any such Pledged Company have not been dealt in or traded on any securities exchange or in securities markets; and (c) the operating agreement of any such Pledged Company does not provide that any limited liability company interest in any such Pledged Company shall be a security governed by Article 8 of the Uniform Commercial Code. Pledgor will not: (A) permit any Pledged Company that is a limited liability company to issue any certificate or other instrument to evidence any of Pledgor's limited liability company interests in such Pledged Company; (B) permit the limited liability company interest in any such Pledged Company to be dealt in or traded on any securities exchange or in securities markets; or (C) allow the operating agreement of any such Pledged Company to provide that any limited liability company interest in any such Pledged Company shall be a security governed by Article 8 of the Uniform Commercial Code.

[SIGNATURE PAGE FOLLOWS]

SIGNATURE PAGE TO PLEDGE AND SECURITY AGREEMENT

IN WITNESS WHEREOF, the Pledger has executed and delivered this Agreement effective as of the date first above written.

PLEDGOR:

GLOBAL WATER RESOURCES, INC.

By: /s/ Michael J. Liebman

Name: Michael J. Liebman

Title: Senior Vice President, Chief Financial Officer and Secretary

COLLATERAL AGENT:

U.S. BANK TRUST COMPANY, NATIONAL ASSOCCIATION, not in its individual capacity, but solely as Collateral Agent

By: /s/ Gregory M. Jackson

Name: Gregory M. Jackson

Title: Vice President

PLEDGE AND SECURITY AGREEMENT GLOBAL WATER, LLC

THIS PLEDGE AND SECURITY AGREEMENT (this "Agreement") dated as of December 10, 2025 is made by: (a) GLOBAL WATER, LLC, a Delaware limited liability company ("Pledgor"); in favor of (b) U.S. BANK TRUST COMPANY, NATIONAL ASSOCIATION, a national banking association (as successor in interest to U.S. Bank National Association), in its capacity as collateral agent (with its successors and permitted assigns in such capacity the "Collateral Agent"); for the benefit of (c) COBANK, ACB, a federally-chartered instrumentality of the United States (the "Bank" and/or any other holder of any Promissory Note at any relevant time the "Holder"), pursuant to the Third Amended and Restated Collateral Agency Agreement dated as of October 26, 2023 (the "Collateral Agency Agreement") among and/or approved by the Collateral Agent, Global Water Resources, Inc., a Delaware corporation ("Company"), Noteholders, The Northern Trust Company and Bank.

PRELIMINARY STATEMENT

WHEREAS, subject to the terms and conditions set forth in a Credit Agreement of even date herewith between Company and the Bank (as amended, supplemented or restated at any relevant time the "Credit Agreement") and a Promissory Note by Company in favor of the Bank (as amended, supplemented or restated at any relevant time the "Initial Promissory Note"), the Bank has agreed to provide Company a term loan in the initial principal amount of \$15,000,000 (the "Loan"); and

WHEREAS, Pledgor is the owner and holder of all of the Equity Interests of the Persons described on Schedule I hereto (the "**Pledged Companies**"); and

WHEREAS, pursuant to Guaranty Agreements (the "Guaranties") of even date herewith, Pledgor, West Maricopa Combine, LLC, an Arizona limited liability company, and Global Water Holdings, Inc., an Arizona corporation, have guaranteed the Obligations of Company under the Credit Agreement and the Initial Promissory Note; and

WHEREAS, it is a condition precedent to the obligation of Bank to provide the Loan that Pledgor execute and deliver this Agreement to the Collateral Agent for the benefit of the Secured Parties; and

WHEREAS, Pledgor desires to execute this Agreement to satisfy such condition precedent and to secure its Obligations under its Guaranty;

NOW THEREFORE, to induce the Bank to provide the Loan pursuant to the Credit Agreement and the Initial Promissory Note, Pledgor and the Collateral Agent agree as follows:

SECTION 1. DEFINED TERMS.

- (a) Capitalized terms used at any place but not defined herein have the meanings set forth in the Credit Agreement.
- (b) "ACC Regulations" means the applicable regulations, orders and requirements of the Arizona Corporation Commission and applicable statutes administered by the Arizona Corporation Commission.
- (c) "Applicable Law" means all applicable Laws, including all applicable provisions of constitutions, statutes, rules, ordinances, regulations and orders of all Governmental Authorities and all orders, rulings, writs and decrees of all courts, tribunals and arbitrators.

- (d) "Equity Interests" means with respect to any Person, all capital stock of (or other ownership or profit interest in) such Person, all warrants, options or other rights for the purchase or acquisition from such Person of capital stock of (or other ownership or profit interest in) such Person, all securities convertible into or exchangeable for capital stock of (or other ownership or profit interests in) such Person or warrants, rights or options for the purchase or acquisition from such Person of such shares (or other interests), and all other ownership or profit interests in such Person (including partnership, membership and trust interests therein), voting or nonvoting, whether or not such shares, warrants, options, rights or other interests are outstanding on any date.
- "Indemnified Liabilities" means, collectively, all liabilities, obligations, losses, damages, penalties, claims (including under Environmental Laws), actions, judgments, suits, costs (including reasonable costs of investigation, study, sampling, testing, abatement, cleanup, removal, remediation or other response necessary to remove, remediate, clean up or abate Hazardous Materials), expenses and disbursements of any kind (including reasonable fees and disbursements of counsel for Indemnitees in connection with any investigative, administrative or judicial proceeding or hearing commenced or threatened by any Person, whether or not such Indemnitees are designated as a party or potential party thereto, and any reasonable fees or expenses incurred by Indemnitees in enforcing this indemnity), whether direct or indirect and whether based on any Applicable Laws (including Environmental Laws), on common law or equitable cause or on contract or otherwise, imposed on, incurred by, or asserted against any such Indemnitee (regardless of whether any Indemnitee is a party thereof), in any manner relating to or arising out of: (i) this Agreement or any other Loan Document or transactions contemplated hereby or thereby (including the use or intended use of the proceeds of the Loan, or any enforcement of any of the Loan Documents (including any sale of, collection from, or other realization on any Collateral)); or (ii) any claim under Environmental Laws or Hazardous Materials liability relating to or arising from, directly or indirectly, any past or present activity, operation, land ownership or practice of Pledgor or any of its Subsidiaries.
 - (f) "**Obligors**" means the Borrower and the Subsidiary Guarantors.
- (g) "Pledged Companies" has the meaning set forth in the above preliminary statement.
 - (h) "Secured Obligations" has the meaning provided by Section 2(c).
 - (i) "Secured Parties" means all Holders and the Collateral Agent.
- (j) "Specified Account" means the segregated account of Company maintained with The Northern Trust Company, with the last four digits of such account number being 2231, established to receive payments of dividends and distributions on Equity Interests owned by Company or any Subsidiary Guarantor, and all replacements or substitutions for such account established by the Company or any Subsidiary Guarantor, whether in the form of a deposit account or securities account. For the avoidance of doubt, a "Specified Account" shall not include any "collection account," deposit account, securities account or other account in which the revenues of Utility Subsidiaries are remitted or consolidated.
- (k) "UCC" means the Uniform Commercial Code as in effect on the date hereof in the State of New York; provided that if by mandatory provisions of Law, the perfection or the effect of perfection or non-perfection of the security interests granted pursuant to Section 2, as well as all other security interests created or assigned as additional security for the Secured Obligations pursuant to this Agreement, in any Collateral is governed by the UCC as in effect in any jurisdiction other than the State of New York, "UCC" means the UCC as in effect in such other jurisdiction for purposes of the provisions hereof relating to such perfection or effect of perfection or non-perfection.

SECTION 2. GRANT OF SECURITY INTEREST.

- (a) To secure the Secured Obligations, Pledgor grants to the Collateral Agent, for the benefit of the Secured Parties, a security interest in, and acknowledges and agrees that the Collateral Agent has and will continue to have a continuing security interest in, all right, title and interest of Pledgor, whether now owned or existing or hereafter created, acquired or arising, and regardless of where located, in and to all the following (collectively the "Collateral"):
- (i) all Equity Interests, including all Equity Interests in the Pledged Companies and all shares, ownership, economic and management interests, membership interests and/or partnership interests in any Person owned or held by Pledgor, all payments and distributions of whatever kind or character, in cash or other property, at any time made, owing or payable to Pledgor in respect of or on account of its present or hereafter acquired Equity Interests, whether due or to become due and whether representing profits, distributions pursuant to complete or partial liquidation or dissolution of the issuer of such Equity Interests, distributions representing the complete or partial redemption of Pledgor's Equity Interests in any Person or complete or partial withdrawal of Pledgor from any Person, repayment of capital contributions made to or with respect to any Person in respect of Equity Interests in such Person held by Pledgor and the right to receive, receipt for, use, and enjoy all such payments and distributions, and all other rights and privileges incident to Pledgor's interest in such Equity Interests, provided that prior to the occurrence of an Event of Default, Pledgor shall retain certain rights pursuant to Section 7;

(ii) the Specified Account;

- (iii) all interest, dividends, cash, instruments, investment property, general intangibles and other property from time to time received, receivable or otherwise payable in respect of, or in exchange for, any of the foregoing;
- (iv) supporting evidence and documentation relating to any of the above-described property, including computer programs, disks, tapes, electronic archives, clouds and related data processing media, and all rights of Pledgor to retrieve the same from third parties, written applications, credit information, account cards, payment records, correspondence, delivery and installation certificates, invoice copies, delivery receipts, notes, and other evidences of indebtedness, insurance certificates and the like, together with all books of account, ledgers and cabinets in which the same are reflected or maintained; and
- (v) to the extent not covered by Sections 2(a)(i) through 2(a)(iv), all proceeds (as defined in the UCC) of any or all of the foregoing.
- (b) The Collateral Agent shall have with respect to the Collateral, in addition to the rights and remedies set forth herein and in the other Loan Documents, all rights and remedies of a secured party under the UCC as if fully set forth herein.
- (c) The security interest herein granted is made and given to secure, and shall secure, the payment and performance of: (i) all Obligations of Pledgor to the Secured Parties (whether arising before or after the filing of a petition in bankruptcy), whether direct or indirect, absolute or contingent, due or to become due, now existing or hereafter arising, however held, evidenced or acquired, and whether several, joint, or joint and several; and (ii) all expenses and charges, legal or otherwise, suffered or incurred by the Secured Parties in collecting or enforcing any such Obligations or realizing on or protecting or preserving any Collateral or other security therefor, including the lien and security interest granted hereby (all of the foregoing being the "Secured Obligations").

- (d) For the avoidance of doubt and notwithstanding anything in any Loan Document to the contrary: (i) no Subsidiary of Pledgor that is now a regulated utility is a borrower or guarantor under any Loan Document, nor is any such Subsidiary pledging any of its property as collateral for the Secured Obligations; and (ii) no regulated utility may declare distributions or dividends to its equity holders except in accordance with Applicable Law (including ACC Regulations), and subject to each regulated utility's obligations to maintain revenues and funds sufficient to fund direct and indirect operating and maintenance expenses (including general and administrative expenses and reasonable and necessary costs, fees and expenses for operation and maintenance of system utilities).
- SECTION 3. CERTIFICATES OR INSTRUMENTS. Any certificates or instruments representing or evidencing Collateral shall be delivered to and held by or on behalf of the Collateral Agent for the benefit of the Secured Parties pursuant hereto and to the Collateral Agency Agreement and be in suitable form for transfer by delivery or accompanied by duly executed instruments of transfer or assignment in blank, all in form reasonably satisfactory to the Collateral Agent and in form and substance reasonably satisfactory to the Bank. Upon the occurrence and during the continuance of any Event of Default, the Collateral Agent will have the right at any time in its sole discretion and without notice to Pledgor to transfer to or to register in the name of the Collateral Agent or its nominees any or all of the Collateral, subject only to the revocable rights specified in Section 7(a) and compliance with ACC Regulations. In addition, on the occurrence and during the continuance of any Event of Default, the Collateral Agent will have the right to exchange certificates or instruments representing or evidencing Collateral for certificates or instruments of smaller or larger denominations.

SECTION 4. SPECIFIED ACCOUNT COLLATERAL.

- The Company has established the Specified Account as a deposit account (within the meaning of Section 9-102 of the UCC), with The Northern Trust Company. Pledgor shall promptly notify the Collateral Agent of any other Specified Account opened or maintained by the Company or Pledgor after the date hereof. To the extent requested by the Collateral Agent following the occurrence and continuation of any Default or the occurrence of any Event of Default, Pledgor shall, and shall cause the applicable depository or other institution to, execute and deliver to the Collateral Agent an account control agreement in form reasonably satisfactory to the Collateral Agent and in form and substance reasonably satisfactory to the Holder which provides among other things for the depository or other institution's agreement that it will comply with: (i) instructions originated by the Collateral Agent directing the disposition of the funds in each such Specified Account that is a deposit account; or (ii) entitlement orders originated by the Collateral Agent with respect to each such Specified Account that is a securities account, in each case without further consent by Pledgor. Notwithstanding the foregoing, any account control agreement which requires the Collateral Agent in its individual capacity to indemnify the depository or other institution other than out of the Collateral will not be reasonably satisfactory to the Collateral Agent.
- (b) Subject to Section 2(d), Pledgor covenants and agrees that upon declaration and payment of any dividend or distribution by any Subsidiary, it will cause such dividend or distribution to be paid and segregated into the Specified Account. Prior to the occurrence and continuance of an Event of Default, amounts so deposited in the Specified Account may be withdrawn and used for corporate purposes, including investments in or loans to Subsidiaries permitted under the Credit Agreement.

SECTION 5. REPRESENTATIONS, WARRANTIES AND CERTAIN COVENANTS. Pledgor represents, warrants and covenants as follows:

(a) The Equity Interests have been duly authorized and validly issued and are fully paid and non-assessable. No Equity Interests constituting interests in limited liability companies

constitute or are evidenced by certificated securities, unless such certificates have been delivered to the Collateral Agent.

- (b) Pledgor is the legal and beneficial owner of the Collateral free and clear of any Lien or other encumbrance except for the Lien created by this Agreement and any other Liens created in favor of the Collateral Agent and described in the Collateral Agency Agreement. There is no existing agreement, option, right or privilege capable of becoming an agreement or option pursuant to which Pledgor would be required to sell or otherwise dispose of any Equity Interest, except as described in the Collateral Agency Agreement.
- (c) Except for the delivery of any certificates or instruments representing Collateral to the Collateral Agent pursuant to this Agreement, filing of an appropriate financing statement with the Delaware Secretary of State, and any control agreement contemplated by Section 4(a), no other action is required to create or maintain the Lien of the Collateral Agent as a valid and perfected first priority Lien (pari passu with the other Secured Parties (as defined in the Collateral Agency Agreement)) in the Collateral to secure the Secured Obligations.
- (d) No authorization, approval or other action by, and no notice to or filing with, any Governmental Authority or regulatory body (except as set forth in Section 5(c)) is required either: (i) for the pledge by Pledgor of the Collateral pursuant to this Agreement or for the execution, delivery or performance of this Agreement by the Pledgor; or (ii) for the exercise by the Collateral Agent of its rights and the rights of any other Secured Parties provided for in this Agreement and the Collateral Agency Agreement or the remedies in respect of the Collateral pursuant to this Agreement and the Collateral Agency Agreement (except as may be required in connection with such disposition by Laws affecting the offering and sale of securities generally and except for compliance with requirements of the ACC Regulations as set forth in Sections 7(b) and 8).
- (e) The execution, delivery and performance of this Agreement does not and will not: (i) violate any provision of any Law (including Regulations T, U and X of the Board of Governors of the Federal Reserve System), order, writ, judgment, injunction, decree, determination or award applicable to Pledgor; (ii) result in breach of, or constitute a default under, any indenture, credit or loan or note agreement or any other agreement, lease or instrument to which Pledgor presently is a party, or by which it or its properties are bound or affected; or (iii) result in or require (other than pursuant to this Agreement) the creation or imposition of any Lien or other share or encumbrance upon or with respect to any properties now owned or hereafter acquired by Pledgor. Pledgor is not in violation of or default under any such Law or material provision of any such indenture, agreement, lease or instrument.
- (f) Schedule I correct sets forth the names of the Pledged Companies and the percentage of Equity Interests in the Pledged Companies owned by Pledgor and pledged by this Agreement.
- (g) Except for any such agreements in favor of the Collateral Agent for the benefit of the Secured Parties and the other Secured Parties (as defined in the Collateral Agency Agreement), Pledgor shall not enter into any agreement providing any Person with "control" (within the meaning of Sections 9-104 or 9-106 of the applicable UCC) of any Specified Account.
- (h) Pledgor is a limited liability company duly formed under the laws of Delaware, and is validly existing and in good standing under the laws of such jurisdiction and the State of Arizona. Pledgor has its chief executive office in the State of Arizona. Pledgor agrees to give the Collateral Agent at least 30 days' prior written notice before changing the state in which its chief executive office is located or the state in which it is organized, and prior to the effectiveness of any such change shall take all steps necessary to maintain the security interest provided for

herein as a first-priority (pari passu with the other Secured Parties (as defined in the Collateral Agency Agreement)) perfected security interest, including filing additional UCC financing statements or amendments as may be necessary or requested by (and subject to the rights of) the Collateral Agent.

- (i) Pledgor shall not: (i) sell, assign (by agreement, operation of law or otherwise) or otherwise dispose of, or grant any option with respect to, any Collateral; or (ii) create or permit to exist any Lien or security interest on or with respect to any Collateral, except for the Lien created by this Agreement and any other Lien in favor of the Collateral Agent for the benefit of the Secured Parties.
- (j) Pledgor shall not permit or cause to be issued any Equity Interests: (i) in substitution for any existing Equity Interests; and (ii) in addition to the existing Equity Interests, except following notice to the Collateral Agent and provided that immediately upon its acquisition (directly or indirectly) of any such substitute or additional securities, Pledgor will execute such documentation as is necessary to pledge or evidence the pledge or as may be requested by Collateral Agent or the Holder pledging, and evidencing the pledge hereunder of, such Equity Interests.
- (k) Pledgor shall, at its expense, protect and defend this Agreement, all rights of the Collateral Agent hereunder, and the Collateral against all claims and demands of other parties. Pledgor shall pay all claims and charges that in the reasonable opinion of the Collateral Agent or the Holder might prejudice, imperil or otherwise affect Collateral or the security interest therein. Pledgor shall promptly notify the Collateral Agent of any levy, distraint or other seizure by legal process or otherwise of any Collateral and of any threatened or filed claims or proceedings that might affect or impair this Agreement.
- SECTION 6. FURTHER ASSURANCES. Pledgor authorizes the Collateral Agent to file any financing statements covering the Collateral or any part thereof as the Collateral Agent may desire. Pledgor agrees that from time to time, at the expense of Pledgor, Pledgor will promptly execute and deliver all further documentation and take all further actions that the Collateral Agent or Holder may reasonably request, in order to perfect and protect any security interest granted or purported to be granted hereby or to enable the Collateral Agent to exercise and enforce its rights and remedies hereunder with respect to any Collateral. Pledgor will furnish to the Collateral Agent from time to time statements and schedules further identifying and describing the Collateral and such other reports in connection with the Collateral as the Collateral Agent or Holder may reasonably request, all in reasonable detail, Pledgor agrees, and agrees to cause each issuer of Equity Interests included in the Collateral, to mark its books and records to reflect the Lien of the Collateral Agent in the Collateral.

SECTION 7. VOTING AND DIVIDENDS.

- (a) So long as no Event of Default has occurred and is continuing:
- (i) Pledgor will be entitled to exercise any voting and other consensual rights pertaining to any Collateral for any purpose not inconsistent with this Agreement or any other Loan Documents.
- (ii) Pledgor will be entitled to receive and retain any dividends or distributions paid in respect of the Collateral; provided however, except as expressly permitted by the Credit Agreement, any (A) dividends or distributions paid or payable other than in cash in respect of, and instruments and other property received, receivable or otherwise distributed in respect of, or in exchange for, any Collateral, (B) dividends and other distributions paid or payable in cash in respect of any Collateral in connection with a partial or total liquidation or dissolution, and (C) cash paid, payable or otherwise distributed in redemption of, or in exchange for, any Collateral,

shall be forthwith delivered to the Collateral Agent to hold as Collateral and shall, if received by Pledgor, be received in trust for the benefit of the Collateral Agent, be segregated from the other property or funds of Pledgor, and be forthwith delivered to the Collateral Agent as Collateral in the same form as so received (with any necessary endorsement).

- (iii) The Collateral Agent shall execute and deliver (or cause to be executed and delivered) to Pledgor all such proxies and other instruments as Pledgor may reasonably request for the purpose of enabling Pledgor to exercise the voting and other rights it is entitled to exercise pursuant to Section7(a)(i) and to receive dividends it is authorized to receive and retain pursuant to Section7(a)(ii).
- (b) Subject to any requirements of ACC Regulations, on the occurrence and during the continuation of an Event of Default:
- (i) All rights of Pledgor to exercise voting and other consensual rights it would otherwise be entitled to exercise pursuant to Section 7(a)(i) will automatically cease, and the Collateral Agent will thereupon have the sole right to exercise such rights.
- (ii) All rights of Pledgor to receive the distributions and dividends it would otherwise be entitled to receive and retain pursuant to Section 7(a)(ii) will automatically cease, and the Collateral Agent will thereupon have the sole right to receive and hold as Collateral such dividends, distributions and interest.
- (iii) Any distributions and dividends received by Pledgor contrary to the provisions of Section 7(b)(ii) will be received in trust for the benefit of the Collateral Agent on behalf of the Secured Parties, shall be segregated from other funds of Pledgor and shall be forthwith paid over to the Collateral Agent on behalf of the Secured Parties as Collateral in the same form as received (with any necessary endorsement).

SECTION 8. REMEDIES. Subject to any requirements of ACC Regulations:

- (a) The Collateral Agent may exercise in respect of any Collateral, in addition to other rights and remedies provided for herein or otherwise available to it, all rights and remedies of a secured party on default under the UCC (whether or not the UCC applies to the affected Collateral), and the Collateral Agent may also, without notice except as specified below, sell the Collateral or any part thereof in one or more parcels at public or private sale, at any exchange, broker's board or at any of the Collateral Agent's offices or elsewhere, for cash, on credit or for future delivery, and on such other terms as the Collateral Agent may claim commercially reasonable. Pledgor agrees that at least ten days' notice to Pledgor of the time and place of any public sale or time after which any private sale is to be made will constitute reasonable notification. The Collateral Agent will not be obligated to make any sale of Collateral regardless of notice of sale having been given. The Collateral Agent may adjourn any public or private sale from time to time by announcement at the time and place fixed therefor, and such sale may, without further notice, be made at the time and place to which it was so adjourned.
- (b) Any cash held by the Collateral Agent as Collateral and all cash proceeds received by the Collateral Agent in respect of any sale of, collection from, or other realization on any Collateral shall be applied by the Collateral Agent against the Secured Obligations in such order as the Collateral Agent may elect, subject to the requirements of the Credit Agreement and Collateral Agency Agreement. Any surplus of such cash or cash proceeds held by the Collateral Agent and remaining after payment in full of all Secured Obligations and all other Secured Obligations (as defined in the Collateral Agency Agreement) shall be paid over to the Pledgor or to whomsoever may be lawfully entitled to receive such surplus.

- (c) All rights and remedies of the Collateral Agent expressed herein are in addition to all other rights and remedies possessed by the Collateral Agent or the Holder in the Loan Documents and any other agreement or instrument relating to the Secured Obligations.
- (d) In connection with a public or private sale of any Collateral, the Collateral Agent may disclose to prospective purchasers any non-public information available to the Collateral Agent which pertains to: (i) the issuer of any Collateral; or (ii) Pledgor, provided in the case of Pledgor such non-public information is material to said issuer, its financial condition or the Collateral.
- (e) Without in any way limiting the foregoing, on the occurrence and during the continuation of any Event of Default, the Collateral Agent shall have the right, in addition to all other rights provided herein or by Law, to direct the disposition of the funds in any Specified Account that is a deposit account, or entitlement orders originated by the secured party with respect to each such Specified Account that is a securities account, in each case without further consent by Pledgor.
- (f) If the Collateral Agent exercises its right to take possession of any Collateral, Pledgor shall also at its expense perform any other steps requested by the Collateral Agent or Holder to preserve and protect the security interest hereby granted in the Collateral, such as maintaining Collateral records and filing UCC financing and continuation statements.

SECTION 9. WAIVERS; PRIVATE SALES.

- Pledgor waives any right to require the Collateral Agent or Holder to: (i) proceed against any Person, including any other Obligor; or (ii) proceed against or exhaust any Collateral, or (iii) pursue any other remedy in Collateral Agent's or the Holder's power; and waives any defense arising by reason of any disability of Pledgor or any other Person. Until the Secured Obligations have been paid in full, Pledgor waives any right of subrogation, reimbursement, indemnification, and contribution (contractual, statutory or otherwise), including any right of subrogation under the Bankruptcy Code (Title 11, United States Code) or any successor statute, arising from the existence or performance of this Agreement, and Pledgor waives any right to enforce any remedy which the Collateral Agent now has or may hereafter have against Pledgor or any other Person, and waives any benefit of, and any right to participate in, any security now or hereafter held by the Collateral Agent. Pledgor authorizes the Collateral Agent, without notice or demand and without affecting Pledgor's liability hereunder, from time to time to: (a) renew, compromise, extend, accelerate or otherwise change the time for payment of, or otherwise change the terms of, any Secured Obligations, including increase or decrease of any rate of interest thereon; (b) receive and hold security, other than the Collateral herein described, for the payment of any Secured Obligations, and exchange, enforce, waive, release, fail to perfect, sell, or otherwise dispose of any Collateral or other security; and (c) release or substitute the Company, or any other guarantors of such Secured Obligations or any part thereof, or any other parties thereto.
- (b) Pledgor recognizes that the Collateral Agent may be unable to effect a public sale of any of the Collateral by reason of ACC Regulations and/or certain prohibitions contained in the Laws of any jurisdiction outside the United States or in the Securities Act and applicable state securities Laws, but may instead be compelled to resort to one or more private sales thereof to a restricted group of purchasers who will be obliged to agree, among other things, to acquire such Collateral for their own account for investment and not with a view to the distribution or resale thereof, or otherwise in accordance with the ACC Regulations. Pledgor agrees that any private sale may result in prices and other terms less favorable to the seller than if such sale were a public sale and, notwithstanding such circumstances, agrees that any such private sale shall, to the extent permitted by Law, be made in a commercially reasonable manner. The Collateral Agent will not be under any obligation to delay a sale of any Collateral for the period of time

necessary to permit the issuer of such securities to register such securities under the Laws of any jurisdiction outside the United States, under the Securities Act or under any applicable state securities Laws, even if the issuer would agree to do so. If the Collateral Agent is able to lawfully effect a public sale without registration of the Collateral under the laws of any jurisdiction outside the United States, under the Securities Act or under any applicable state securities Laws, then, subject to any applicable ACC Regulations, the Collateral Agent may, but will not be required to, conduct a public sale of the Collateral, rather than a private sale, if the Collateral Agent reasonably believes it would realize a higher sales price in a public sale.

SECTION 10. COLLATERAL AGENT'S DUTIES.

- (a) The powers conferred on the Collateral Agent hereunder are solely to protect the interests of the Secured Parties in the Collateral and shall not impose any duty on the Collateral Agent to exercise any such powers. Except for reasonable care in the custody of any Collateral in its possession and accounting for moneys actually received by it hereunder, the Collateral Agent will have no duty as to any Collateral or as to the taking of any necessary steps to preserve rights against prior parties or any other rights pertaining to any Collateral. The Collateral Agent will have exercised reasonable care in the custody and preservation of Collateral in its possession if the Collateral is accorded treatment substantially equal to that which the Collateral Agent accords its own property and collateral held for others in its capacity as a collateral agent, it being understood that the Collateral Agent will not have any responsibility for: (a) ascertaining or taking action with respect to calls, conversions, exchanges, maturities, tenders or other matters relative to any Collateral, whether or not the Collateral Agent has or is deemed to have knowledge of such matters; or (b) taking any necessary steps to preserve rights against any parties with respect to any Collateral.
- (b) Upon the appointment of a replacement collateral agent pursuant to the Collateral Agency Agreement, the Collateral Agent may transfer all of its interest and Liens in or any of the Collateral and be fully discharged thereafter from all liability and responsibility with respect to such Collateral so transferred, and the transferee shall be vested with all the rights and powers of the Collateral Agent hereunder with respect to such Collateral so transferred; but with respect to any of its interest and liens in Collateral not so transferred, the Collateral Agent will retain all rights and powers hereby given.

SECTION 11. OTHER RIGHTS.

- (a) The rights, powers and remedies given to the Collateral Agent by this Agreement shall be in addition to all rights, powers and remedies given to the Collateral Agent by virtue of any statute or rule of Law. Any forbearance or failure or delay by the Collateral Agent in exercising any right, power or remedy hereunder shall not be a waiver of such right, power or remedy, and any single or partial exercise of any right, power or remedy hereunder shall not preclude the further exercise thereof; and every right, power and remedy of the Collateral Agent shall continue in full force and effect until such right, power or remedy is specifically waived by a writing executed by the Collateral Agent. The rights and remedies of the Collateral Agent under this Agreement shall be cumulative and not exclusive of any other right or remedy which the Collateral Agent or any Holder may have.
- (b) This Agreement constitutes an assignment of rights only and not an assignment of any duties or obligations of Pledgor in any way related to the Collateral, and the Collateral Agent will have no duty or obligation to discharge any such duty or obligation. The Collateral Agent will have no responsibility for taking any necessary steps to preserve rights against any parties with respect to any Collateral or initiating any action to protect any Collateral against the possibility of a decline in market value. Neither the Collateral Agent nor any party acting as attorney for the Collateral Agent will be liable for any acts or omissions or for any error of

judgment or mistake of fact or Law other than its gross negligence or willful misconduct or negligence in the handling of funds.

In addition to any other powers of attorney contained herein, Pledgor hereby appoints the Collateral Agent, its nominee, and any other Person whom the Collateral Agent may designate, as Pledgor's attorney-in-fact, with full power and authority to sign Pledgor's name on verifications of Collateral; to send requests for verification of Collateral to obligors; to endorse Pledgor's name on any checks, notes, acceptances, money orders, drafts and other forms of payment or security that may come into the Collateral Agent's possession or on any assignments, stock powers or other instruments of transfer relating to any Collateral; to sign Pledgor's name on claims to enforce collection of any Collateral, on notices to and drafts against obligors, on schedules and assignments of Collateral, on notices of assignment and on public records; on the occurrence and during the continuance of an Event of Default to notify post office authorities to change the address for delivery of Pledgor's mail to an address designated by the Collateral Agent; upon the occurrence and during the continuance of an Event of Default to receive, open and dispose of all mail addressed to Pledgor; and to do all things necessary to carry out this Agreement. Pledgor hereby ratifies and approves all acts of any such attorney and agrees that neither the Collateral Agent nor any such attorney will be liable for any acts or omissions nor for any error of judgment or mistake of fact or Law other than such Person's gross negligence or willful misconduct or negligence in the handling of funds. The foregoing powers of attorney, being coupled with an interest, are irrevocable until the Secured Obligations have been fully paid and satisfied and all agreements of any Secured Party to extend credit to or for the account of Pledgor have expired or otherwise been terminated.

SECTION 12. INDEMNITY; WAIVER.

- Pledgor agrees to indemnify, pay and hold harmless, the Collateral Agent and each other Secured Party and any of their Related Parties (each an "Indemnitee"), from and against any and all Indemnified Liabilities, IN ALL CASES, WHETHER OR NOT CAUSED BY OR ARISING, IN WHOLE OR IN PART, OUT OF THE COMPARATIVE OR CONTRIBUTORY NEGLIGENCE OF SUCH INDEMNITEE; provided, Pledgor shall not have any obligation to any Indemnitee hereunder with respect to any Indemnified Liabilities to the extent such Indemnified Liabilities (x) arise from the gross negligence or willful misconduct, as determined by a court of competent jurisdiction by final and nonappealable judgment, of that Indemnitee, or (y) result from a claim brought by Pledgor against an Indemnitee for breach in bad faith of such Indemnitee's obligations hereunder or under any other Loan Document, as determined by a court of competent jurisdiction by final and nonappealable judgment. To the extent the undertakings to indemnify, pay and hold harmless in this Section 12 may be unenforceable in whole or in part because they are violative of any Law or public policy, Pledgor shall contribute the maximum portion it is permitted to pay and satisfy under Applicable Laws to the payment and satisfaction of all Indemnified Liabilities incurred by any Indemnitees. All amounts due under this Section 12(a) shall be payable promptly after demand therefor. For purposes hereof, "Related Parties" means, with respect to any Person, such Person's Affiliates and the partners, managers, directors, trustees, officers, employees or other personnel, counsel, agents and advisors of such Person and of such Person's Affiliates.
- (b) To the extent not prohibited by Applicable Law, Pledgor shall not assert, and Pledgor waives, any claim against the Collateral Agent and its respective Related Parties, on any theory of liability, for special, indirect, consequential or punitive damages (as opposed to direct or actual damages) (whether or not the claim therefor is based on contract, tort or duty imposed by any applicable legal requirement) arising out of, in connection with, as a result of, or in any way related to, this Agreement or any other Loan Document or any agreement or instrument contemplated hereby or thereby or referred to herein or therein, the transactions contemplated hereby or thereby, or the use of the proceeds thereof or any act or omission or event occurring in

10

connection therewith, and Pledgor waives, releases and agrees not to sue on any such claim or any such damages, whether or not accrued and whether or not known or suspected to exist in its favor. No Indemnitee shall be liable for any damages arising from the use by unintended recipients of any information or other materials distributed to it through telecommunications, electronic or other information transmission systems in connection with this Agreement or any other Loan Documents or the transactions contemplated hereby or thereby. In no event shall the Collateral Agent be responsible or liable for any failure or delay in the performance of its obligations hereunder arising out of or caused by; directly or indirectly, forces beyond its control, including strikes, work stoppages, accidents, acts of war or terrorism, civil or military disturbances, nuclear or natural catastrophes or acts of God, and interruptions, loss or malfunctions of utilities, communications or computer (software and hardware) services, it being understood that the Collateral Agent shall use reasonable best efforts which are consistent with accepted practices in the banking industry to resume performance as soon as practicable under the circumstances.

SECTION 13. INTERPRETATION.

- (a) In this Agreement, unless a clear contrary intention appears:
 - (i) the singular number includes the plural number and vice versa;
 - (ii) reference to any gender includes each other gender;
- (iii) the words "herein," "hereof" and "hereunder" and words of similar import refer to this Agreement as a whole and not to any particular Article, Section or other subdivision;
- (iv) reference to any Person includes such Person's successors and assigns but, if applicable, only if such successors and assigns are permitted by this Agreement, and reference to a Person in a particular capacity excludes such Person in any other capacity or individually, provided that nothing in this Section 13(a)(iv) is intended to authorize any assignment not otherwise permitted by this Agreement;
- (v) reference to any agreement, document or instrument means such agreement, document or instrument as amended, supplemented or modified and in effect from time to time in accordance with the terms thereof and, if applicable, the terms hereof, and reference to any note includes any note issued pursuant to any Loan Document in extension or renewal thereof and in substitution or replacement therefor;
- (vi) unless the context indicates otherwise, reference to any Article, Section or Schedule means such Article or Section hereof or Schedule hereto;
- (vii) "including" (with its correlative meaning "include") means including, without limiting the generality of any description preceding such term;
- (viii) with respect to the determination of any period of time, the word "from" means "from and including" and the word "to" means "to but excluding;"
- (ix) reference to any Law means such as modified, codified or reenacted, in whole or in part, and in effect from time to time; and
- (x) references to the Collateral Agent are to it in its capacity as Collateral Agent for the Secured Parties and the other Secured Parties (as defined in the Collateral Agency Agreement) under and as provided in the Collateral Agency Agreement.
- (b) Article and Section headings herein are for convenience only and shall not affect the construction hereof.

- (c) No provision of this Agreement shall be interpreted or construed against any Person because that Person or its legal representative drafted such provision.
- SECTION 14. AMENDMENTS. No amendment or waiver of any provision of this Agreement, nor consent to any departure by Pledgor herefrom, will be effective unless in writing and signed by Pledgor and the Collateral Agent, and then only in the specific instance and for the specific purpose for which given.
- SECTION 15. NOTICES. All notices and other communications provided for hereunder shall be in writing and sent: (a) by telecopy, facsimile or electronic mail if the sender on the same day sends a confirming copy of such notice by a nationally recognized overnight delivery service (charges prepaid); (b) by registered or certified mail with return receipt requested (postage prepaid); or (c) by a nationally recognized overnight delivery service (with charges prepaid). Any such notice must be sent:
- (i) if to the Bank or its nominee, to the Bank or nominee at the address specified for such communications in the Credit Agreement, or at such other address as the Bank or nominee shall have specified to the Company in writing;
- (ii) if to any other Holder, to such Holder at such address as such Holder shall have specified to the Company in writing;
- (iii) if to Pledgor, to Pledgor at the following address: 21410 North 19th Avenue, Suite 220, Phoenix, Arizona 85027 to the attention of: Michael J. Liebman, or at such other address as Pledgor shall have specified to the Collateral Agent and each Holder in writing; or
- (iv) if to the Collateral Agent, at 333 Commerce Street, Suite 900, Nashville, Tennessee 37201, Attention: W. Jones (Global Water Resources, Inc.), wally.jones@usbank.com, or at such other address as the Collateral Agent shall have specified to Pledgor in writing.

Notices under this Section 15 will be effective only when actually received.

SECTION 16. SEPARABILITY. If any clause, sentence, paragraph, subsection or Section of this Agreement is judicially declared to be invalid, unenforceable or void, such decision will not have the effect of invalidating or voiding the remainder of this Agreement, and the part or parts of this Agreement so held to be invalid, unenforceable or void will be deemed to have been stricken herefrom by the parties hereto, and the remainder of this Agreement will have the same force and effect as if such stricken part or parts had never been included herein

SECTION 17. COUNTERPARTS. This Agreement may be executed in any number of counterparts and by different parties in separate counterparts, each of which when so executed will be an original and all of which taken together will constitute one and the same Agreement. Signatures of parties transmitted by facsimile or electronic transmission will be their original signatures for all purposes.

SECTION 18. CONTINUING LIEN. This Agreement shall create a continuing security interest in the Collateral and shall: (a) remain in effect until payment in full of the Secured Obligations; (b) be binding on Pledgor, its successors and assigns; and (c) inure to the benefit of the Collateral Agent and other Secured Parties and their successors, transferees and assigns. Without limiting the foregoing clause (c), any Holder may assign or otherwise transfer all or a portion of its interests, rights and obligations under any Promissory Note in accordance with the Loan Documents. Upon the payment in full of the Secured Obligations and the other Secured Obligations (as defined in the Collateral Agency Agreement), Pledgor will be entitled to the return, on its request and at its expense, of such Collateral as shall not have been disposed of.

SECTION 19. SURVIVAL. All representations and warranties in this Agreement or made in writing by Pledgor in connection herewith will survive the execution and delivery of this Agreement and repayment of the Secured Obligations. Any investigation by any Secured Party will not diminish in any respect its right to rely on such representations and warranties.

SECTION 20. LAW. THIS AGREEMENT AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES HEREUNDER SHALL BE GOVERNED BY, AND SHALL BE CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK WITHOUT REGARD TO CONFLICT OF LAWS PRINCIPLES THEREOF THAT WOULD REQUIRE OR PERMIT THE APPLICATION OF THE LAWS OF ANOTHER JURISDICTION EXCEPT TO THE EXTENT THE PERFECTION OF THE SECURITY INTEREST HEREUNDER, OR REMEDIES HEREUNDER, IN RESPECT OF ANY PARTICULAR COLLATERAL ARE GOVERNED BY THE LAWS OF A JURISDICTION OTHER THAN THE STATE OF NEW YORK.

SECTION 21. JURISDICTION; WAIVER OF JURY TRIAL.

- (a) Pledgor irrevocably submits to the non-exclusive jurisdiction of any New York State or federal court sitting in the Borough of Manhattan, The City of New York, over any suit, action or proceeding arising out of or relating to the Loan Documents. To the fullest extent permitted by Applicable Law, Pledgor irrevocably waives and agrees not to assert, by way of motion, as a defense or otherwise, any claim that it is not subject to the jurisdiction of any such court, any objection that it may now or hereafter have to the laying of the venue of any such suit, action or proceeding brought in any such court and any claim that any such suit, action or proceeding brought in any such court has been brought in an inconvenient forum.
- (b) Pledgor consents to process being served by or on behalf of the Collateral Agent or any Holder in any suit, action or proceeding of the nature referred to in Section 21(a) by hand delivery, reputable overnight commercial delivery service or by mailing a copy thereof by registered or certified or express mail (or any substantially similar form of mail), postage prepaid; return receipt requested, in each case to it at its address described in Section 15 or at such other address of which the Collateral Agent or such Holder shall then have been notified pursuant to Section 15. Pledgor agrees that such service upon receipt: (i) will be in every respect effective service of process on it in any such suit, action or proceeding; and (ii) will, to the fullest extent permitted by Applicable Law, be valid personal service on and personal delivery to it. Notices hereunder will be conclusively presumed received as evidenced by a delivery receipt furnished by the United States Postal Service or any reputable commercial delivery service.
- (c) Nothing in this Section 21 shall affect the right of the Collateral Agent or any Holder to serve process in any manner permitted by Law, or limit any right the Collateral Agent or Holder may have to bring proceedings against Pledgor in the courts of any appropriate jurisdiction or to enforce in any lawful manner a judgment obtained in one jurisdiction in any other jurisdiction.
- (d) THE PARTIES HERETO HEREBY WAIVE TRIAL BY JURY IN ANY ACTION BROUGHT ON OR WITH RESPECT TO OR RELATED TO ANY LOAN DOCUMENT.

SECTION 22. FINAL AGREEMENT. THIS AGREEMENT AND ANY OTHER LOAN DOCUMENTS EXECUTED IN CONNECTION HEREWITH REPRESENT (WITH THE COLLATERAL AGENCY AGREEMENT) THE FINAL AGREEMENT BETWEEN THE PARTIES AND MAY NOT BE CONTRADICTED BY EVIDENCE OF ANY PRIOR, CONTEMPORANEOUS, OR SUBSEQUENT ORAL AGREEMENTS OF THE PARTIES. THERE ARE NO UNWRITTEN ORAL AGREEMENTS BETWEEN THE PARTIES.

SECTION 23. LLC INTERESTS. Pledgor represents and warrants that: (a) any Pledged Company that is a limited liability company has not issued any certificate or other instrument to evidence any of Pledgor's limited liability company interests in such Pledged Company; (b) the limited liability company interest in any such Pledged Company have not been dealt in or traded on any securities exchange or in securities markets; and (c) the operating agreement of any such Pledged Company does not provide that any limited liability company interest in any such Pledged Company shall be a security governed by Article 8 of the Uniform Commercial Code. Pledgor will not: (A) permit any Pledged Company that is a limited liability company to issue any certificate or other instrument to evidence any of Pledgor's limited liability company interests in such Pledged Company; (B) permit the limited liability company interest in any such Pledged Company to be dealt in or traded on any securities exchange or in securities markets; or (C) allow the operating agreement of any such Pledged Company to provide that any limited liability company interest in any such Pledged Company shall be a security governed by Article 8 of the Uniform Commercial Code.

[SIGNATURE PAGE FOLLOWS]

SIGNATURE PAGE TO PLEDGE AND SECURITY AGREEMENT

IN WITNESS WHEREOF, the Pledgor has executed and delivered this Agreement effective as of the date first above written.

PLEDGOR:

GLOBAL WATER RESOURCES, LLC

By: /s/ Michael J. Liebman

Name: Michael J. Liebman

Title: Manager, Secretary and Treasurer

COLLATERAL AGENT:

U.S. BANK TRUST COMPANY, NATIONAL ASSOCCIATION, not in its individual capacity, but solely as Collateral Agent

By: /s/ Gregory M. Jackson

Name: Gregory M. Jackson

Title: Vice President

Schedule I
PLEDGED COMPANIES

Issuer	Class of Shares/ Membership Interests	Number of Shares	Percentage of Outstanding Shares/ Membership Interests
Global Water – 303 Utilities Company, Inc.	Common	1,000	100%
Global Water – CP Water Company, Inc.	Common	1,000	100%
Global Water – Hassayampa Utilities Company, Inc.	Common	1,000	100%
Global Water – Palo Verde Utilities Company, Inc.	Common	1,000	100%
Global Water – Santa Cruz Water Company, Inc.	Common	1,000	100%
Global Water – Turner Ranches Irrigation, Inc.	Common	1,000	100%
Global Water Holdings, Inc.	Common	1,000	100%
West Maricopa Combine, LLC	Common	N/A	100%

PLEDGE AND SECURITY AGREEMENT WEST MARICOPA COMBINE, LLC

THIS PLEDGE AND SECURITY AGREEMENT (this "Agreement") dated as of December 10, 2025 is made by: (a) WEST MARICOPA COMBINE, LLC, an Arizona limited liability company ("Pledgor"); in favor of (b) U.S. BANK TRUST COMPANY, NATIONAL ASSOCIATION, a national banking association (as successor in interest to U.S. Bank National Association), in its capacity as collateral agent (with its successors and permitted assigns in such capacity the "Collateral Agent"); for the benefit of (c) COBANK, ACB, a federally-chartered instrumentality of the United States (the "Bank" and/or any other holder of any Promissory Note at any relevant time the "Holder"), pursuant to the Third Amended and Restated Collateral Agency Agreement dated as of October 26, 2023 (the "Collateral Agency Agreement") among and/or approved by the Collateral Agent, Global Water Resources, Inc., a Delaware corporation ("Company"), Noteholders, The Northern Trust Company and Bank.

PRELIMINARY STATEMENT

WHEREAS, subject to the terms and conditions set forth in a Credit Agreement of even date herewith between Company and the Bank (as amended, supplemented or restated at any relevant time the "Credit Agreement") and a Promissory Note by Company in favor of the Bank (as amended, supplemented or restated at any relevant time the "Initial Promissory Note"), the Bank has agreed to provide Company a term loan in the initial principal amount of \$15,000,000 (the "Loan"); and

WHEREAS, Pledgor is the owner and holder of all of the Equity Interests of the Persons described on Schedule I hereto (the "**Pledged Companies**"); and

WHEREAS, pursuant to Guaranty Agreements (the "Guaranties") of even date herewith, Global Water, LLC, a Delaware limited liability company, Pledgor, and Global Water Holdings, Inc., an Arizona corporation, have guaranteed the Obligations of Company under the Credit Agreement and the Initial Promissory Note; and

WHEREAS, it is a condition precedent to the obligation of Bank to provide the Loan that Pledgor execute and deliver this Agreement to the Collateral Agent for the benefit of the Secured Parties; and

WHEREAS, Pledgor desires to execute this Agreement to satisfy such condition precedent and to secure its Obligations under its Guaranty;

NOW THEREFORE, to induce the Bank to provide the Loan pursuant to the Credit Agreement and the Initial Promissory Note, Pledgor and the Collateral Agent agree as follows:

SECTION 1. DEFINED TERMS.

- (a) Capitalized terms used at any place but not defined herein have the meanings set forth in the Credit Agreement.
- (b) "ACC Regulations" means the applicable regulations, orders and requirements of the Arizona Corporation Commission and applicable statutes administered by the Arizona Corporation Commission.
- (c) "Applicable Law" means all applicable Laws, including all applicable provisions of constitutions, statutes, rules, ordinances, regulations and orders of all Governmental Authorities and all orders, rulings, writs and decrees of all courts, tribunals and arbitrators.

- (d) "Equity Interests" means with respect to any Person, all capital stock of (or other ownership or profit interest in) such Person, all warrants, options or other rights for the purchase or acquisition from such Person of capital stock of (or other ownership or profit interest in) such Person, all securities convertible into or exchangeable for capital stock of (or other ownership or profit interests in) such Person or warrants, rights or options for the purchase or acquisition from such Person of such shares (or other interests), and all other ownership or profit interests in such Person (including partnership, membership and trust interests therein), voting or nonvoting, whether or not such shares, warrants, options, rights or other interests are outstanding on any date.
- "Indemnified Liabilities" means, collectively, all liabilities, obligations, losses, damages, penalties, claims (including under Environmental Laws), actions, judgments, suits, costs (including reasonable costs of investigation, study, sampling, testing, abatement, cleanup, removal, remediation or other response necessary to remove, remediate, clean up or abate Hazardous Materials), expenses and disbursements of any kind (including reasonable fees and disbursements of counsel for Indemnitees in connection with any investigative, administrative or judicial proceeding or hearing commenced or threatened by any Person, whether or not such Indemnitees are designated as a party or potential party thereto, and any reasonable fees or expenses incurred by Indemnitees in enforcing this indemnity), whether direct or indirect and whether based on any Applicable Laws (including Environmental Laws), on common law or equitable cause or on contract or otherwise, imposed on, incurred by, or asserted against any such Indemnitee (regardless of whether any Indemnitee is a party thereof), in any manner relating to or arising out of: (i) this Agreement or any other Loan Document or transactions contemplated hereby or thereby (including the use or intended use of the proceeds of the Loan, or any enforcement of any of the Loan Documents (including any sale of, collection from, or other realization on any Collateral)); or (ii) any claim under Environmental Laws or Hazardous Materials liability relating to or arising from, directly or indirectly, any past or present activity, operation, land ownership or practice of Pledgor or any of its Subsidiaries.
 - (f) "**Obligors**" means the Borrower and the Subsidiary Guarantors.
- (g) "Pledged Companies" has the meaning set forth in the above preliminary statement.
 - (h) "Secured Obligations" has the meaning provided by Section 2(c).
 - (i) "Secured Parties" means all Holders and the Collateral Agent.
- (j) "Specified Account" means the segregated account of Company maintained with The Northern Trust Company, with the last four digits of such account number being 2231, established to receive payments of dividends and distributions on Equity Interests owned by Company or any Subsidiary Guarantor, and all replacements or substitutions for such account established by the Company or any Subsidiary Guarantor, whether in the form of a deposit account or securities account. For the avoidance of doubt, a "Specified Account" shall not include any "collection account," deposit account, securities account or other account in which the revenues of Utility Subsidiaries are remitted or consolidated.
- (k) "UCC" means the Uniform Commercial Code as in effect on the date hereof in the State of New York; provided that if by mandatory provisions of Law, the perfection or the effect of perfection or non-perfection of the security interests granted pursuant to Section 2, as well as all other security interests created or assigned as additional security for the Secured Obligations pursuant to this Agreement, in any Collateral is governed by the UCC as in effect in any jurisdiction other than the State of New York, "UCC" means the UCC as in effect in such other jurisdiction for purposes of the provisions hereof relating to such perfection or effect of perfection or non-perfection.

SECTION 2. GRANT OF SECURITY INTEREST.

- (a) To secure the Secured Obligations, Pledgor grants to the Collateral Agent, for the benefit of the Secured Parties, a security interest in, and acknowledges and agrees that the Collateral Agent has and will continue to have a continuing security interest in, all right, title and interest of Pledgor, whether now owned or existing or hereafter created, acquired or arising, and regardless of where located, in and to all the following (collectively the "Collateral"):
- (i) all Equity Interests, including all Equity Interests in the Pledged Companies and all shares, ownership, economic and management interests, membership interests and/or partnership interests in any Person owned or held by Pledgor, all payments and distributions of whatever kind or character, in cash or other property, at any time made, owing or payable to Pledgor in respect of or on account of its present or hereafter acquired Equity Interests, whether due or to become due and whether representing profits, distributions pursuant to complete or partial liquidation or dissolution of the issuer of such Equity Interests, distributions representing the complete or partial redemption of Pledgor's Equity Interests in any Person or complete or partial withdrawal of Pledgor from any Person, repayment of capital contributions made to or with respect to any Person in respect of Equity Interests in such Person held by Pledgor and the right to receive, receipt for, use, and enjoy all such payments and distributions, and all other rights and privileges incident to Pledgor's interest in such Equity Interests, provided that prior to the occurrence of an Event of Default, Pledgor shall retain certain rights pursuant to Section 7;

(ii) the Specified Account;

- (iii) all interest, dividends, cash, instruments, investment property, general intangibles and other property from time to time received, receivable or otherwise payable in respect of, or in exchange for, any of the foregoing;
- (iv) supporting evidence and documentation relating to any of the above-described property, including computer programs, disks, tapes, electronic archives, clouds and related data processing media, and all rights of Pledgor to retrieve the same from third parties, written applications, credit information, account cards, payment records, correspondence, delivery and installation certificates, invoice copies, delivery receipts, notes, and other evidences of indebtedness, insurance certificates and the like, together with all books of account, ledgers and cabinets in which the same are reflected or maintained; and
- (v) to the extent not covered by Sections 2(a)(i) through 2(a)(iv), all proceeds (as defined in the UCC) of any or all of the foregoing.
- (b) The Collateral Agent shall have with respect to the Collateral, in addition to the rights and remedies set forth herein and in the other Loan Documents, all rights and remedies of a secured party under the UCC as if fully set forth herein.
- (c) The security interest herein granted is made and given to secure, and shall secure, the payment and performance of: (i) all Obligations of Pledgor to the Secured Parties (whether arising before or after the filing of a petition in bankruptcy), whether direct or indirect, absolute or contingent, due or to become due, now existing or hereafter arising, however held, evidenced or acquired, and whether several, joint, or joint and several; and (ii) all expenses and charges, legal or otherwise, suffered or incurred by the Secured Parties in collecting or enforcing any such Obligations or realizing on or protecting or preserving any Collateral or other security therefor, including the lien and security interest granted hereby (all of the foregoing being the "Secured Obligations").

- (d) For the avoidance of doubt and notwithstanding anything in any Loan Document to the contrary: (i) no Subsidiary of Pledgor that is now a regulated utility is a borrower or guarantor under any Loan Document, nor is any such Subsidiary pledging any of its property as collateral for the Secured Obligations; and (ii) no regulated utility may declare distributions or dividends to its equity holders except in accordance with Applicable Law (including ACC Regulations), and subject to each regulated utility's obligations to maintain revenues and funds sufficient to fund direct and indirect operating and maintenance expenses (including general and administrative expenses and reasonable and necessary costs, fees and expenses for operation and maintenance of system utilities).
- SECTION 3. CERTIFICATES OR INSTRUMENTS. Any certificates or instruments representing or evidencing Collateral shall be delivered to and held by or on behalf of the Collateral Agent for the benefit of the Secured Parties pursuant hereto and to the Collateral Agency Agreement and be in suitable form for transfer by delivery or accompanied by duly executed instruments of transfer or assignment in blank, all in form reasonably satisfactory to the Collateral Agent and in form and substance reasonably satisfactory to the Bank. Upon the occurrence and during the continuance of any Event of Default, the Collateral Agent will have the right at any time in its sole discretion and without notice to Pledgor to transfer to or to register in the name of the Collateral Agent or its nominees any or all of the Collateral, subject only to the revocable rights specified in Section 7(a) and compliance with ACC Regulations. In addition, on the occurrence and during the continuance of any Event of Default, the Collateral Agent will have the right to exchange certificates or instruments representing or evidencing Collateral for certificates or instruments of smaller or larger denominations.

SECTION 4. SPECIFIED ACCOUNT COLLATERAL.

- The Company has established the Specified Account as a deposit account (within the meaning of Section 9-102 of the UCC), with The Northern Trust Company. Pledgor shall promptly notify the Collateral Agent of any other Specified Account opened or maintained by the Company or Pledgor after the date hereof. To the extent requested by the Collateral Agent following the occurrence and continuation of any Default or the occurrence of any Event of Default, Pledgor shall, and shall cause the applicable depository or other institution to, execute and deliver to the Collateral Agent an account control agreement in form reasonably satisfactory to the Collateral Agent and in form and substance reasonably satisfactory to the Holder which provides among other things for the depository or other institution's agreement that it will comply with: (i) instructions originated by the Collateral Agent directing the disposition of the funds in each such Specified Account that is a deposit account; or (ii) entitlement orders originated by the Collateral Agent with respect to each such Specified Account that is a securities account, in each case without further consent by Pledgor. Notwithstanding the foregoing, any account control agreement which requires the Collateral Agent in its individual capacity to indemnify the depository or other institution other than out of the Collateral will not be reasonably satisfactory to the Collateral Agent.
- (b) Subject to Section 2(d), Pledgor covenants and agrees that upon declaration and payment of any dividend or distribution by any Subsidiary, it will cause such dividend or distribution to be paid and segregated into the Specified Account. Prior to the occurrence and continuance of an Event of Default, amounts so deposited in the Specified Account may be withdrawn and used for corporate purposes, including investments in or loans to Subsidiaries permitted under the Credit Agreement.

SECTION 5. REPRESENTATIONS, WARRANTIES AND CERTAIN COVENANTS. Pledgor represents, warrants and covenants as follows:

(a) The Equity Interests have been duly authorized and validly issued and are fully paid and non-assessable. No Equity Interests constituting interests in limited liability companies

constitute or are evidenced by certificated securities, unless such certificates have been delivered to the Collateral Agent.

- (b) Pledgor is the legal and beneficial owner of the Collateral free and clear of any Lien or other encumbrance except for the Lien created by this Agreement and any other Liens created in favor of the Collateral Agent and described in the Collateral Agency Agreement. There is no existing agreement, option, right or privilege capable of becoming an agreement or option pursuant to which Pledgor would be required to sell or otherwise dispose of any Equity Interest, except as described in the Collateral Agency Agreement.
- (c) Except for the delivery of any certificates or instruments representing Collateral to the Collateral Agent pursuant to this Agreement, filing of an appropriate financing statement with the Arizona Secretary of State, and any control agreement contemplated by Section 4(a), no other action is required to create or maintain the Lien of the Collateral Agent as a valid and perfected first priority Lien (pari passu with the other Secured Parties (as defined in the Collateral Agency Agreement)) in the Collateral to secure the Secured Obligations.
- (d) No authorization, approval or other action by, and no notice to or filing with, any Governmental Authority or regulatory body (except as set forth in Section 5(c)) is required either: (i) for the pledge by Pledgor of the Collateral pursuant to this Agreement or for the execution, delivery or performance of this Agreement by the Pledgor; or (ii) for the exercise by the Collateral Agent of its rights and the rights of any other Secured Parties provided for in this Agreement and the Collateral Agency Agreement or the remedies in respect of the Collateral pursuant to this Agreement and the Collateral Agency Agreement (except as may be required in connection with such disposition by Laws affecting the offering and sale of securities generally and except for compliance with requirements of the ACC Regulations as set forth in Sections 7(b) and 8).
- (e) The execution, delivery and performance of this Agreement does not and will not: (i) violate any provision of any Law (including Regulations T, U and X of the Board of Governors of the Federal Reserve System), order, writ, judgment, injunction, decree, determination or award applicable to Pledgor; (ii) result in breach of, or constitute a default under, any indenture, credit or loan or note agreement or any other agreement, lease or instrument to which Pledgor presently is a party, or by which it or its properties are bound or affected; or (iii) result in or require (other than pursuant to this Agreement) the creation or imposition of any Lien or other share or encumbrance upon or with respect to any properties now owned or hereafter acquired by Pledgor. Pledgor is not in violation of or default under any such Law or material provision of any such indenture, agreement, lease or instrument.
- (f) Schedule I correct sets forth the names of the Pledged Companies and the percentage of Equity Interests in the Pledged Companies owned by Pledgor and pledged by this Agreement.
- (g) Except for any such agreements in favor of the Collateral Agent for the benefit of the Secured Parties and the other Secured Parties (as defined in the Collateral Agency Agreement), Pledgor shall not enter into any agreement providing any Person with "control" (within the meaning of Sections 9-104 or 9-106 of the applicable UCC) of any Specified Account.
- (h) Pledgor is a limited liability company duly formed under the laws of Arizona, and is validly existing and in good standing under the laws of such jurisdiction. Pledgor has its chief executive office in the State of Arizona. Pledgor agrees to give the Collateral Agent at least 30 days' prior written notice before changing the state in which its chief executive office is located or the state in which it is organized, and prior to the effectiveness of any such change shall take all steps necessary to maintain the security interest provided for herein as a first-priority (pari

passu with the other Secured Parties (as defined in the Collateral Agency Agreement)) perfected security interest, including filing additional UCC financing statements or amendments as may be necessary or requested by (and subject to the rights of) the Collateral Agent.

- (i) Pledgor shall not: (i) sell, assign (by agreement, operation of law or otherwise) or otherwise dispose of, or grant any option with respect to, any Collateral; or (ii) create or permit to exist any Lien or security interest on or with respect to any Collateral, except for the Lien created by this Agreement and any other Lien in favor of the Collateral Agent for the benefit of the Secured Parties.
- (j) Pledgor shall not permit or cause to be issued any Equity Interests: (i) in substitution for any existing Equity Interests; and (ii) in addition to the existing Equity Interests, except following notice to the Collateral Agent and provided that immediately upon its acquisition (directly or indirectly) of any such substitute or additional securities, Pledgor will execute such documentation as is necessary to pledge or evidence the pledge or as may be requested by Collateral Agent or the Holder pledging, and evidencing the pledge hereunder of, such Equity Interests.
- (k) Pledgor shall, at its expense, protect and defend this Agreement, all rights of the Collateral Agent hereunder, and the Collateral against all claims and demands of other parties. Pledgor shall pay all claims and charges that in the reasonable opinion of the Collateral Agent or the Holder might prejudice, imperil or otherwise affect Collateral or the security interest therein. Pledgor shall promptly notify the Collateral Agent of any levy, distraint or other seizure by legal process or otherwise of any Collateral and of any threatened or filed claims or proceedings that might affect or impair this Agreement.
- SECTION 6. FURTHER ASSURANCES. Pledgor authorizes the Collateral Agent to file any financing statements covering the Collateral or any part thereof as the Collateral Agent may desire. Pledgor agrees that from time to time, at the expense of Pledgor, Pledgor will promptly execute and deliver all further documentation and take all further actions that the Collateral Agent or Holder may reasonably request, in order to perfect and protect any security interest granted or purported to be granted hereby or to enable the Collateral Agent to exercise and enforce its rights and remedies hereunder with respect to any Collateral. Pledgor will furnish to the Collateral Agent from time to time statements and schedules further identifying and describing the Collateral and such other reports in connection with the Collateral as the Collateral Agent or Holder may reasonably request, all in reasonable detail, Pledgor agrees, and agrees to cause each issuer of Equity Interests included in the Collateral, to mark its books and records to reflect the Lien of the Collateral Agent in the Collateral.

SECTION 7. VOTING AND DIVIDENDS.

- (a) So long as no Event of Default has occurred and is continuing:
- (i) Pledgor will be entitled to exercise any voting and other consensual rights pertaining to any Collateral for any purpose not inconsistent with this Agreement or any other Loan Documents.
- (ii) Pledgor will be entitled to receive and retain any dividends or distributions paid in respect of the Collateral; provided however, except as expressly permitted by the Credit Agreement, any (A) dividends or distributions paid or payable other than in cash in respect of, and instruments and other property received, receivable or otherwise distributed in respect of, or in exchange for, any Collateral, (B) dividends and other distributions paid or payable in cash in respect of any Collateral in connection with a partial or total liquidation or dissolution, and (C) cash paid, payable or otherwise distributed in redemption of, or in exchange for, any Collateral, shall be forthwith delivered to the Collateral Agent to hold as Collateral and shall, if received by

Pledgor, be received in trust for the benefit of the Collateral Agent, be segregated from the other property or funds of Pledgor, and be forthwith delivered to the Collateral Agent as Collateral in the same form as so received (with any necessary endorsement).

- (iii) The Collateral Agent shall execute and deliver (or cause to be executed and delivered) to Pledgor all such proxies and other instruments as Pledgor may reasonably request for the purpose of enabling Pledgor to exercise the voting and other rights it is entitled to exercise pursuant to Section7(a)(i) and to receive dividends it is authorized to receive and retain pursuant to Section7(a)(ii).
- (b) Subject to any requirements of ACC Regulations, on the occurrence and during the continuation of an Event of Default:
- (i) All rights of Pledgor to exercise voting and other consensual rights it would otherwise be entitled to exercise pursuant to Section 7(a)(i) will automatically cease, and the Collateral Agent will thereupon have the sole right to exercise such rights.
- (ii) All rights of Pledgor to receive the distributions and dividends it would otherwise be entitled to receive and retain pursuant to Section 7(a)(ii) will automatically cease, and the Collateral Agent will thereupon have the sole right to receive and hold as Collateral such dividends, distributions and interest.
- (iii) Any distributions and dividends received by Pledgor contrary to the provisions of Section 7(b)(ii) will be received in trust for the benefit of the Collateral Agent on behalf of the Secured Parties, shall be segregated from other funds of Pledgor and shall be forthwith paid over to the Collateral Agent on behalf of the Secured Parties as Collateral in the same form as received (with any necessary endorsement).

SECTION 8. REMEDIES. Subject to any requirements of ACC Regulations:

- (a) The Collateral Agent may exercise in respect of any Collateral, in addition to other rights and remedies provided for herein or otherwise available to it, all rights and remedies of a secured party on default under the UCC (whether or not the UCC applies to the affected Collateral), and the Collateral Agent may also, without notice except as specified below, sell the Collateral or any part thereof in one or more parcels at public or private sale, at any exchange, broker's board or at any of the Collateral Agent's offices or elsewhere, for cash, on credit or for future delivery, and on such other terms as the Collateral Agent may claim commercially reasonable. Pledgor agrees that at least ten days' notice to Pledgor of the time and place of any public sale or time after which any private sale is to be made will constitute reasonable notification. The Collateral Agent will not be obligated to make any sale of Collateral regardless of notice of sale having been given. The Collateral Agent may adjourn any public or private sale from time to time by announcement at the time and place fixed therefor, and such sale may, without further notice, be made at the time and place to which it was so adjourned.
- (b) Any cash held by the Collateral Agent as Collateral and all cash proceeds received by the Collateral Agent in respect of any sale of, collection from, or other realization on any Collateral shall be applied by the Collateral Agent against the Secured Obligations in such order as the Collateral Agent may elect, subject to the requirements of the Credit Agreement and Collateral Agency Agreement. Any surplus of such cash or cash proceeds held by the Collateral Agent and remaining after payment in full of all Secured Obligations and all other Secured Obligations (as defined in the Collateral Agency Agreement) shall be paid over to the Pledgor or to whomsoever may be lawfully entitled to receive such surplus.

- (c) All rights and remedies of the Collateral Agent expressed herein are in addition to all other rights and remedies possessed by the Collateral Agent or the Holder in the Loan Documents and any other agreement or instrument relating to the Secured Obligations.
- (d) In connection with a public or private sale of any Collateral, the Collateral Agent may disclose to prospective purchasers any non-public information available to the Collateral Agent which pertains to: (i) the issuer of any Collateral; or (ii) Pledgor, provided in the case of Pledgor such non-public information is material to said issuer, its financial condition or the Collateral.
- (e) Without in any way limiting the foregoing, on the occurrence and during the continuation of any Event of Default, the Collateral Agent shall have the right, in addition to all other rights provided herein or by Law, to direct the disposition of the funds in any Specified Account that is a deposit account, or entitlement orders originated by the secured party with respect to each such Specified Account that is a securities account, in each case without further consent by Pledgor.
- (f) If the Collateral Agent exercises its right to take possession of any Collateral, Pledgor shall also at its expense perform any other steps requested by the Collateral Agent or Holder to preserve and protect the security interest hereby granted in the Collateral, such as maintaining Collateral records and filing UCC financing and continuation statements.

SECTION 9. WAIVERS; PRIVATE SALES.

- Pledgor waives any right to require the Collateral Agent or Holder to: (i) proceed against any Person, including any other Obligor; or (ii) proceed against or exhaust any Collateral, or (iii) pursue any other remedy in Collateral Agent's or the Holder's power; and waives any defense arising by reason of any disability of Pledgor or any other Person. Until the Secured Obligations have been paid in full, Pledgor waives any right of subrogation, reimbursement, indemnification, and contribution (contractual, statutory or otherwise), including any right of subrogation under the Bankruptcy Code (Title 11, United States Code) or any successor statute, arising from the existence or performance of this Agreement, and Pledgor waives any right to enforce any remedy which the Collateral Agent now has or may hereafter have against Pledgor or any other Person, and waives any benefit of, and any right to participate in, any security now or hereafter held by the Collateral Agent. Pledgor authorizes the Collateral Agent, without notice or demand and without affecting Pledgor's liability hereunder, from time to time to: (a) renew, compromise, extend, accelerate or otherwise change the time for payment of, or otherwise change the terms of, any Secured Obligations, including increase or decrease of any rate of interest thereon; (b) receive and hold security, other than the Collateral herein described, for the payment of any Secured Obligations, and exchange, enforce, waive, release, fail to perfect, sell, or otherwise dispose of any Collateral or other security; and (c) release or substitute the Company, or any other guarantors of such Secured Obligations or any part thereof, or any other parties thereto.
- (b) Pledgor recognizes that the Collateral Agent may be unable to effect a public sale of any of the Collateral by reason of ACC Regulations and/or certain prohibitions contained in the Laws of any jurisdiction outside the United States or in the Securities Act and applicable state securities Laws, but may instead be compelled to resort to one or more private sales thereof to a restricted group of purchasers who will be obliged to agree, among other things, to acquire such Collateral for their own account for investment and not with a view to the distribution or resale thereof, or otherwise in accordance with the ACC Regulations. Pledgor agrees that any private sale may result in prices and other terms less favorable to the seller than if such sale were a public sale and, notwithstanding such circumstances, agrees that any such private sale shall, to the extent permitted by Law, be made in a commercially reasonable manner. The Collateral Agent will not be under any obligation to delay a sale of any Collateral for the period of time

necessary to permit the issuer of such securities to register such securities under the Laws of any jurisdiction outside the United States, under the Securities Act or under any applicable state securities Laws, even if the issuer would agree to do so. If the Collateral Agent is able to lawfully effect a public sale without registration of the Collateral under the laws of any jurisdiction outside the United States, under the Securities Act or under any applicable state securities Laws, then, subject to any applicable ACC Regulations, the Collateral Agent may, but will not be required to, conduct a public sale of the Collateral, rather than a private sale, if the Collateral Agent reasonably believes it would realize a higher sales price in a public sale.

SECTION 10. COLLATERAL AGENT'S DUTIES.

- (a) The powers conferred on the Collateral Agent hereunder are solely to protect the interests of the Secured Parties in the Collateral and shall not impose any duty on the Collateral Agent to exercise any such powers. Except for reasonable care in the custody of any Collateral in its possession and accounting for moneys actually received by it hereunder, the Collateral Agent will have no duty as to any Collateral or as to the taking of any necessary steps to preserve rights against prior parties or any other rights pertaining to any Collateral. The Collateral Agent will have exercised reasonable care in the custody and preservation of Collateral in its possession if the Collateral is accorded treatment substantially equal to that which the Collateral Agent accords its own property and collateral held for others in its capacity as a collateral agent, it being understood that the Collateral Agent will not have any responsibility for: (a) ascertaining or taking action with respect to calls, conversions, exchanges, maturities, tenders or other matters relative to any Collateral, whether or not the Collateral Agent has or is deemed to have knowledge of such matters; or (b) taking any necessary steps to preserve rights against any parties with respect to any Collateral.
- (b) Upon the appointment of a replacement collateral agent pursuant to the Collateral Agency Agreement, the Collateral Agent may transfer all of its interest and Liens in or any of the Collateral and be fully discharged thereafter from all liability and responsibility with respect to such Collateral so transferred, and the transferee shall be vested with all the rights and powers of the Collateral Agent hereunder with respect to such Collateral so transferred; but with respect to any of its interest and liens in Collateral not so transferred, the Collateral Agent will retain all rights and powers hereby given.

SECTION 11. OTHER RIGHTS.

- (a) The rights, powers and remedies given to the Collateral Agent by this Agreement shall be in addition to all rights, powers and remedies given to the Collateral Agent by virtue of any statute or rule of Law. Any forbearance or failure or delay by the Collateral Agent in exercising any right, power or remedy hereunder shall not be a waiver of such right, power or remedy, and any single or partial exercise of any right, power or remedy hereunder shall not preclude the further exercise thereof; and every right, power and remedy of the Collateral Agent shall continue in full force and effect until such right, power or remedy is specifically waived by a writing executed by the Collateral Agent. The rights and remedies of the Collateral Agent under this Agreement shall be cumulative and not exclusive of any other right or remedy which the Collateral Agent or any Holder may have.
- (b) This Agreement constitutes an assignment of rights only and not an assignment of any duties or obligations of Pledgor in any way related to the Collateral, and the Collateral Agent will have no duty or obligation to discharge any such duty or obligation. The Collateral Agent will have no responsibility for taking any necessary steps to preserve rights against any parties with respect to any Collateral or initiating any action to protect any Collateral against the possibility of a decline in market value. Neither the Collateral Agent nor any party acting as attorney for the Collateral Agent will be liable for any acts or omissions or for any error of

judgment or mistake of fact or Law other than its gross negligence or willful misconduct or negligence in the handling of funds.

In addition to any other powers of attorney contained herein, Pledgor hereby appoints the Collateral Agent, its nominee, and any other Person whom the Collateral Agent may designate, as Pledgor's attorney-in-fact, with full power and authority to sign Pledgor's name on verifications of Collateral; to send requests for verification of Collateral to obligors; to endorse Pledgor's name on any checks, notes, acceptances, money orders, drafts and other forms of payment or security that may come into the Collateral Agent's possession or on any assignments, stock powers or other instruments of transfer relating to any Collateral; to sign Pledgor's name on claims to enforce collection of any Collateral, on notices to and drafts against obligors, on schedules and assignments of Collateral, on notices of assignment and on public records; on the occurrence and during the continuance of an Event of Default to notify post office authorities to change the address for delivery of Pledgor's mail to an address designated by the Collateral Agent; upon the occurrence and during the continuance of an Event of Default to receive, open and dispose of all mail addressed to Pledgor; and to do all things necessary to carry out this Agreement. Pledgor hereby ratifies and approves all acts of any such attorney and agrees that neither the Collateral Agent nor any such attorney will be liable for any acts or omissions nor for any error of judgment or mistake of fact or Law other than such Person's gross negligence or willful misconduct or negligence in the handling of funds. The foregoing powers of attorney, being coupled with an interest, are irrevocable until the Secured Obligations have been fully paid and satisfied and all agreements of any Secured Party to extend credit to or for the account of Pledgor have expired or otherwise been terminated.

SECTION 12. INDEMNITY; WAIVER.

- Pledgor agrees to indemnify, pay and hold harmless, the Collateral Agent and each other Secured Party and any of their Related Parties (each an "Indemnitee"), from and against any and all Indemnified Liabilities, IN ALL CASES, WHETHER OR NOT CAUSED BY OR ARISING, IN WHOLE OR IN PART, OUT OF THE COMPARATIVE OR CONTRIBUTORY NEGLIGENCE OF SUCH INDEMNITEE; provided, Pledgor shall not have any obligation to any Indemnitee hereunder with respect to any Indemnified Liabilities to the extent such Indemnified Liabilities (x) arise from the gross negligence or willful misconduct, as determined by a court of competent jurisdiction by final and nonappealable judgment, of that Indemnitee, or (y) result from a claim brought by Pledgor against an Indemnitee for breach in bad faith of such Indemnitee's obligations hereunder or under any other Loan Document, as determined by a court of competent jurisdiction by final and nonappealable judgment. To the extent the undertakings to indemnify, pay and hold harmless in this Section 12 may be unenforceable in whole or in part because they are violative of any Law or public policy, Pledgor shall contribute the maximum portion it is permitted to pay and satisfy under Applicable Laws to the payment and satisfaction of all Indemnified Liabilities incurred by any Indemnitees. All amounts due under this Section 12(a) shall be payable promptly after demand therefor. For purposes hereof, "Related Parties" means, with respect to any Person, such Person's Affiliates and the partners, managers, directors, trustees, officers, employees or other personnel, counsel, agents and advisors of such Person and of such Person's Affiliates.
- (b) To the extent not prohibited by Applicable Law, Pledgor shall not assert, and Pledgor waives, any claim against the Collateral Agent and its respective Related Parties, on any theory of liability, for special, indirect, consequential or punitive damages (as opposed to direct or actual damages) (whether or not the claim therefor is based on contract, tort or duty imposed by any applicable legal requirement) arising out of, in connection with, as a result of, or in any way related to, this Agreement or any other Loan Document or any agreement or instrument contemplated hereby or thereby or referred to herein or therein, the transactions contemplated hereby or thereby, or the use of the proceeds thereof or any act or omission or event occurring in

10

connection therewith, and Pledgor waives, releases and agrees not to sue on any such claim or any such damages, whether or not accrued and whether or not known or suspected to exist in its favor. No Indemnitee shall be liable for any damages arising from the use by unintended recipients of any information or other materials distributed to it through telecommunications, electronic or other information transmission systems in connection with this Agreement or any other Loan Documents or the transactions contemplated hereby or thereby. In no event shall the Collateral Agent be responsible or liable for any failure or delay in the performance of its obligations hereunder arising out of or caused by; directly or indirectly, forces beyond its control, including strikes, work stoppages, accidents, acts of war or terrorism, civil or military disturbances, nuclear or natural catastrophes or acts of God, and interruptions, loss or malfunctions of utilities, communications or computer (software and hardware) services, it being understood that the Collateral Agent shall use reasonable best efforts which are consistent with accepted practices in the banking industry to resume performance as soon as practicable under the circumstances.

SECTION 13. INTERPRETATION.

- (a) In this Agreement, unless a clear contrary intention appears:
 - (i) the singular number includes the plural number and vice versa;
 - (ii) reference to any gender includes each other gender;
- (iii) the words "herein," "hereof" and "hereunder" and words of similar import refer to this Agreement as a whole and not to any particular Article, Section or other subdivision;
- (iv) reference to any Person includes such Person's successors and assigns but, if applicable, only if such successors and assigns are permitted by this Agreement, and reference to a Person in a particular capacity excludes such Person in any other capacity or individually, provided that nothing in this Section 13(a)(iv) is intended to authorize any assignment not otherwise permitted by this Agreement;
- (v) reference to any agreement, document or instrument means such agreement, document or instrument as amended, supplemented or modified and in effect from time to time in accordance with the terms thereof and, if applicable, the terms hereof, and reference to any note includes any note issued pursuant to any Loan Document in extension or renewal thereof and in substitution or replacement therefor;
- (vi) unless the context indicates otherwise, reference to any Article, Section or Schedule means such Article or Section hereof or Schedule hereto;
- (vii) "including" (with its correlative meaning "include") means including, without limiting the generality of any description preceding such term;
- (viii) with respect to the determination of any period of time, the word "from" means "from and including" and the word "to" means "to but excluding;"
- (ix) reference to any Law means such as modified, codified or reenacted, in whole or in part, and in effect from time to time; and
- (x) references to the Collateral Agent are to it in its capacity as Collateral Agent for the Secured Parties and the other Secured Parties (as defined in the Collateral Agency Agreement) under and as provided in the Collateral Agency Agreement.
- (b) Article and Section headings herein are for convenience only and shall not affect the construction hereof.

- (c) No provision of this Agreement shall be interpreted or construed against any Person because that Person or its legal representative drafted such provision.
- SECTION 14. AMENDMENTS. No amendment or waiver of any provision of this Agreement, nor consent to any departure by Pledgor herefrom, will be effective unless in writing and signed by Pledgor and the Collateral Agent, and then only in the specific instance and for the specific purpose for which given.
- SECTION 15. NOTICES. All notices and other communications provided for hereunder shall be in writing and sent: (a) by telecopy, facsimile or electronic mail if the sender on the same day sends a confirming copy of such notice by a nationally recognized overnight delivery service (charges prepaid); (b) by registered or certified mail with return receipt requested (postage prepaid); or (c) by a nationally recognized overnight delivery service (with charges prepaid). Any such notice must be sent:
- (i) if to the Bank or its nominee, to the Bank or nominee at the address specified for such communications in the Credit Agreement, or at such other address as the Bank or nominee shall have specified to the Company in writing;
- (ii) if to any other Holder, to such Holder at such address as such Holder shall have specified to the Company in writing;
- (iii) if to Pledgor, to Pledgor at the following address: 21410 North 19th Avenue, Suite 220, Phoenix, Arizona 85027 to the attention of: Michael J. Liebman, or at such other address as Pledgor shall have specified to the Collateral Agent and each Holder in writing; or
- (iv) if to the Collateral Agent, at 333 Commerce Street, Suite 900, Nashville, Tennessee 37201, Attention: W. Jones (Global Water Resources, Inc.), wally.jones@usbank.com, or at such other address as the Collateral Agent shall have specified to Pledgor in writing.

Notices under this Section 15 will be effective only when actually received.

SECTION 16. SEPARABILITY. If any clause, sentence, paragraph, subsection or Section of this Agreement is judicially declared to be invalid, unenforceable or void, such decision will not have the effect of invalidating or voiding the remainder of this Agreement, and the part or parts of this Agreement so held to be invalid, unenforceable or void will be deemed to have been stricken herefrom by the parties hereto, and the remainder of this Agreement will have the same force and effect as if such stricken part or parts had never been included herein

SECTION 17. COUNTERPARTS. This Agreement may be executed in any number of counterparts and by different parties in separate counterparts, each of which when so executed will be an original and all of which taken together will constitute one and the same Agreement. Signatures of parties transmitted by facsimile or electronic transmission will be their original signatures for all purposes.

SECTION 18. CONTINUING LIEN. This Agreement shall create a continuing security interest in the Collateral and shall: (a) remain in effect until payment in full of the Secured Obligations; (b) be binding on Pledgor, its successors and assigns; and (c) inure to the benefit of the Collateral Agent and other Secured Parties and their successors, transferees and assigns. Without limiting the foregoing clause (c), any Holder may assign or otherwise transfer all or a portion of its interests, rights and obligations under any Promissory Note in accordance with the Loan Documents. Upon the payment in full of the Secured Obligations and the other Secured Obligations (as defined in the Collateral Agency Agreement), Pledgor will be entitled to the return, on its request and at its expense, of such Collateral as shall not have been disposed of.

SECTION 19. SURVIVAL. All representations and warranties in this Agreement or made in writing by Pledgor in connection herewith will survive the execution and delivery of this Agreement and repayment of the Secured Obligations. Any investigation by any Secured Party will not diminish in any respect its right to rely on such representations and warranties.

SECTION 20. LAW. THIS AGREEMENT AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES HEREUNDER SHALL BE GOVERNED BY, AND SHALL BE CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK WITHOUT REGARD TO CONFLICT OF LAWS PRINCIPLES THEREOF THAT WOULD REQUIRE OR PERMIT THE APPLICATION OF THE LAWS OF ANOTHER JURISDICTION EXCEPT TO THE EXTENT THE PERFECTION OF THE SECURITY INTEREST HEREUNDER, OR REMEDIES HEREUNDER, IN RESPECT OF ANY PARTICULAR COLLATERAL ARE GOVERNED BY THE LAWS OF A JURISDICTION OTHER THAN THE STATE OF NEW YORK.

SECTION 21. JURISDICTION; WAIVER OF JURY TRIAL.

- (a) Pledgor irrevocably submits to the non-exclusive jurisdiction of any New York State or federal court sitting in the Borough of Manhattan, The City of New York, over any suit, action or proceeding arising out of or relating to the Loan Documents. To the fullest extent permitted by Applicable Law, Pledgor irrevocably waives and agrees not to assert, by way of motion, as a defense or otherwise, any claim that it is not subject to the jurisdiction of any such court, any objection that it may now or hereafter have to the laying of the venue of any such suit, action or proceeding brought in any such court and any claim that any such suit, action or proceeding brought in any such court has been brought in an inconvenient forum.
- (b) Pledgor consents to process being served by or on behalf of the Collateral Agent or any Holder in any suit, action or proceeding of the nature referred to in Section 21(a) by hand delivery, reputable overnight commercial delivery service or by mailing a copy thereof by registered or certified or express mail (or any substantially similar form of mail), postage prepaid; return receipt requested, in each case to it at its address described in Section 15 or at such other address of which the Collateral Agent or such Holder shall then have been notified pursuant to Section 15. Pledgor agrees that such service upon receipt: (i) will be in every respect effective service of process on it in any such suit, action or proceeding; and (ii) will, to the fullest extent permitted by Applicable Law, be valid personal service on and personal delivery to it. Notices hereunder will be conclusively presumed received as evidenced by a delivery receipt furnished by the United States Postal Service or any reputable commercial delivery service.
- (c) Nothing in this Section 21 shall affect the right of the Collateral Agent or any Holder to serve process in any manner permitted by Law, or limit any right the Collateral Agent or Holder may have to bring proceedings against Pledgor in the courts of any appropriate jurisdiction or to enforce in any lawful manner a judgment obtained in one jurisdiction in any other jurisdiction.
- (d) THE PARTIES HERETO HEREBY WAIVE TRIAL BY JURY IN ANY ACTION BROUGHT ON OR WITH RESPECT TO OR RELATED TO ANY LOAN DOCUMENT.

SECTION 22. FINAL AGREEMENT. THIS AGREEMENT AND ANY OTHER LOAN DOCUMENTS EXECUTED IN CONNECTION HEREWITH REPRESENT (WITH THE COLLATERAL AGENCY AGREEMENT) THE FINAL AGREEMENT BETWEEN THE PARTIES AND MAY NOT BE CONTRADICTED BY EVIDENCE OF ANY PRIOR, CONTEMPORANEOUS, OR SUBSEQUENT ORAL AGREEMENTS OF THE PARTIES. THERE ARE NO UNWRITTEN ORAL AGREEMENTS BETWEEN THE PARTIES.

SECTION 23. LLC INTERESTS. Pledgor represents and warrants that: (a) any Pledged Company that is a limited liability company has not issued any certificate or other instrument to evidence any of Pledgor's limited liability company interests in such Pledged Company; (b) the limited liability company interest in any such Pledged Company have not been dealt in or traded on any securities exchange or in securities markets; and (c) the operating agreement of any such Pledged Company does not provide that any limited liability company interest in any such Pledged Company shall be a security governed by Article 8 of the Uniform Commercial Code. Pledgor will not: (A) permit any Pledged Company that is a limited liability company to issue any certificate or other instrument to evidence any of Pledgor's limited liability company interest in any such Pledged Company to be dealt in or traded on any securities exchange or in securities markets; or (C) allow the operating agreement of any such Pledged Company to provide that any limited liability company interest in any such Pledged Company shall be a security governed by Article 8 of the Uniform Commercial Code.

[SIGNATURE PAGE FOLLOWS]

SIGNATURE PAGE TO PLEDGE AND SECURITY AGREEMENT

IN WITNESS WHEREOF, the Pledgor has executed and delivered this Agreement effective as of the date first above written.

PLEDGOR:

WEST MARICOPA COMBINE, LLC

By: /s/ Michael J. Liebman

Name: Michael J. Liebman

Title: Manager

COLLATERAL AGENT:

U.S. BANK TRUST COMPANY, NATIONAL ASSOCCIATION, not in its individual capacity, but solely as Collateral Agent

By: /s/ Gregory M. Jackson

Name: Gregory M. Jackson

Title: Vice President

Schedule I PLEDGED COMPANIES

Issuer	Class of Shares	Number of Shares	Percentage of Outstanding Shares
Global Water – Belmont Water Company, Inc.	Common	1,000	100%
Valencia Water Company, Inc.	Common	1,000	100%
Water Utility of Greater Buckeye, Inc.	Common	1,000	100%
Willow Valley Water Company, Inc.	Common	1,000	100%

PLEDGE AND SECURITY AGREEMENT GLOBAL WATER HOLDINGS, INC.

THIS PLEDGE AND SECURITY AGREEMENT (this "Agreement") dated as of December 10, 2025 is made by: (a) GLOBAL WATER HOLDINGS, INC., an Arizona corporation ("Pledgor"); in favor of (b) U.S. BANK TRUST COMPANY, NATIONAL ASSOCIATION, a national banking association (as successor in interest to U.S. Bank National Association), in its capacity as collateral agent (with its successors and permitted assigns in such capacity the "Collateral Agent"); for the benefit of (c) COBANK, ACB, a federally-chartered instrumentality of the United States (the "Bank" and/or any other holder of any Promissory Note at any relevant time the "Holder"), pursuant to the Third Amended and Restated Collateral Agency Agreement dated as of October 26, 2023 (the "Collateral Agency Agreement") among and/or approved by the Collateral Agent, Global Water Resources, Inc., a Delaware corporation ("Company"), Noteholders, The Northern Trust Company and Bank.

PRELIMINARY STATEMENT

WHEREAS, subject to the terms and conditions set forth in a Credit Agreement of even date herewith between Company and the Bank (as amended, supplemented or restated at any relevant time the "Credit Agreement") and a Promissory Note by Company in favor of the Bank (as amended, supplemented or restated at any relevant time the "Initial Promissory Note"), the Bank has agreed to provide Company a term loan in the initial principal amount of \$15,000,000 (the "Loan"); and

WHEREAS, Pledgor is the owner and holder of all of the Equity Interests of the Persons described on Schedule I hereto (the "**Pledged Companies**"); and

WHEREAS, pursuant to Guaranty Agreements (the "Guaranties") of even date herewith, Global Water, LLC, a Delaware limited liability company, West Maricopa Combine, LLC, an Arizona limited liability company, and Pledgor have guaranteed the Obligations of Company under the Credit Agreement and the Initial Promissory Note; and

WHEREAS, it is a condition precedent to the obligation of Bank to provide the Loan that Pledgor execute and deliver this Agreement to the Collateral Agent for the benefit of the Secured Parties; and

WHEREAS, Pledgor desires to execute this Agreement to satisfy such condition precedent and to secure its Obligations under its Guaranty;

NOW THEREFORE, to induce the Bank to provide the Loan pursuant to the Credit Agreement and the Initial Promissory Note, Pledgor and the Collateral Agent agree as follows:

SECTION 1. DEFINED TERMS.

- (a) Capitalized terms used at any place but not defined herein have the meanings set forth in the Credit Agreement.
- (b) "ACC Regulations" means the applicable regulations, orders and requirements of the Arizona Corporation Commission and applicable statutes administered by the Arizona Corporation Commission.
- (c) "Applicable Law" means all applicable Laws, including all applicable provisions of constitutions, statutes, rules, ordinances, regulations and orders of all Governmental Authorities and all orders, rulings, writs and decrees of all courts, tribunals and arbitrators.

- (d) "Equity Interests" means with respect to any Person, all capital stock of (or other ownership or profit interest in) such Person, all warrants, options or other rights for the purchase or acquisition from such Person of capital stock of (or other ownership or profit interest in) such Person, all securities convertible into or exchangeable for capital stock of (or other ownership or profit interests in) such Person or warrants, rights or options for the purchase or acquisition from such Person of such shares (or other interests), and all other ownership or profit interests in such Person (including partnership, membership and trust interests therein), voting or nonvoting, whether or not such shares, warrants, options, rights or other interests are outstanding on any date.
- "Indemnified Liabilities" means, collectively, all liabilities, obligations, losses, damages, penalties, claims (including under Environmental Laws), actions, judgments, suits, costs (including reasonable costs of investigation, study, sampling, testing, abatement, cleanup, removal, remediation or other response necessary to remove, remediate, clean up or abate Hazardous Materials), expenses and disbursements of any kind (including reasonable fees and disbursements of counsel for Indemnitees in connection with any investigative, administrative or judicial proceeding or hearing commenced or threatened by any Person, whether or not such Indemnitees are designated as a party or potential party thereto, and any reasonable fees or expenses incurred by Indemnitees in enforcing this indemnity), whether direct or indirect and whether based on any Applicable Laws (including Environmental Laws), on common law or equitable cause or on contract or otherwise, imposed on, incurred by, or asserted against any such Indemnitee (regardless of whether any Indemnitee is a party thereof), in any manner relating to or arising out of: (i) this Agreement or any other Loan Document or transactions contemplated hereby or thereby (including the use or intended use of the proceeds of the Loan, or any enforcement of any of the Loan Documents (including any sale of, collection from, or other realization on any Collateral)); or (ii) any claim under Environmental Laws or Hazardous Materials liability relating to or arising from, directly or indirectly, any past or present activity, operation, land ownership or practice of Pledgor or any of its Subsidiaries.
 - (f) "**Obligors**" means the Borrower and the Subsidiary Guarantors.
- (g) "Pledged Companies" has the meaning set forth in the above preliminary statement.
 - (h) "Secured Obligations" has the meaning provided by Section 2(c).
 - (i) "Secured Parties" means all Holders and the Collateral Agent.
- (j) "Specified Account" means the segregated account of Company maintained with The Northern Trust Company, with the last four digits of such account number being 2231, established to receive payments of dividends and distributions on Equity Interests owned by Company or any Subsidiary Guarantor, and all replacements or substitutions for such account established by the Company or any Subsidiary Guarantor, whether in the form of a deposit account or securities account. For the avoidance of doubt, a "Specified Account" shall not include any "collection account," deposit account, securities account or other account in which the revenues of Utility Subsidiaries are remitted or consolidated.
- (k) "UCC" means the Uniform Commercial Code as in effect on the date hereof in the State of New York; provided that if by mandatory provisions of Law, the perfection or the effect of perfection or non-perfection of the security interests granted pursuant to Section 2, as well as all other security interests created or assigned as additional security for the Secured Obligations pursuant to this Agreement, in any Collateral is governed by the UCC as in effect in any jurisdiction other than the State of New York, "UCC" means the UCC as in effect in such other jurisdiction for purposes of the provisions hereof relating to such perfection or effect of perfection or non-perfection.

SECTION 2. GRANT OF SECURITY INTEREST.

- (a) To secure the Secured Obligations, Pledgor grants to the Collateral Agent, for the benefit of the Secured Parties, a security interest in, and acknowledges and agrees that the Collateral Agent has and will continue to have a continuing security interest in, all right, title and interest of Pledgor, whether now owned or existing or hereafter created, acquired or arising, and regardless of where located, in and to all the following (collectively the "Collateral"):
- (i) all Equity Interests, including all Equity Interests in the Pledged Companies and all shares, ownership, economic and management interests, membership interests and/or partnership interests in any Person owned or held by Pledgor, all payments and distributions of whatever kind or character, in cash or other property, at any time made, owing or payable to Pledgor in respect of or on account of its present or hereafter acquired Equity Interests, whether due or to become due and whether representing profits, distributions pursuant to complete or partial liquidation or dissolution of the issuer of such Equity Interests, distributions representing the complete or partial redemption of Pledgor's Equity Interests in any Person or complete or partial withdrawal of Pledgor from any Person, repayment of capital contributions made to or with respect to any Person in respect of Equity Interests in such Person held by Pledgor and the right to receive, receipt for, use, and enjoy all such payments and distributions, and all other rights and privileges incident to Pledgor's interest in such Equity Interests, provided that prior to the occurrence of an Event of Default, Pledgor shall retain certain rights pursuant to Section 7;

(ii) the Specified Account;

- (iii) all interest, dividends, cash, instruments, investment property, general intangibles and other property from time to time received, receivable or otherwise payable in respect of, or in exchange for, any of the foregoing;
- (iv) supporting evidence and documentation relating to any of the above-described property, including computer programs, disks, tapes, electronic archives, clouds and related data processing media, and all rights of Pledgor to retrieve the same from third parties, written applications, credit information, account cards, payment records, correspondence, delivery and installation certificates, invoice copies, delivery receipts, notes, and other evidences of indebtedness, insurance certificates and the like, together with all books of account, ledgers and cabinets in which the same are reflected or maintained; and
- (v) to the extent not covered by Sections 2(a)(i) through 2(a)(iv), all proceeds (as defined in the UCC) of any or all of the foregoing.
- (b) The Collateral Agent shall have with respect to the Collateral, in addition to the rights and remedies set forth herein and in the other Loan Documents, all rights and remedies of a secured party under the UCC as if fully set forth herein.
- (c) The security interest herein granted is made and given to secure, and shall secure, the payment and performance of: (i) all Obligations of Pledgor to the Secured Parties (whether arising before or after the filing of a petition in bankruptcy), whether direct or indirect, absolute or contingent, due or to become due, now existing or hereafter arising, however held, evidenced or acquired, and whether several, joint, or joint and several; and (ii) all expenses and charges, legal or otherwise, suffered or incurred by the Secured Parties in collecting or enforcing any such Obligations or realizing on or protecting or preserving any Collateral or other security therefor, including the lien and security interest granted hereby (all of the foregoing being the "Secured Obligations").

- (d) For the avoidance of doubt and notwithstanding anything in any Loan Document to the contrary: (i) no Subsidiary of Pledgor that is now a regulated utility is a borrower or guarantor under any Loan Document, nor is any such Subsidiary pledging any of its property as collateral for the Secured Obligations; and (ii) no regulated utility may declare distributions or dividends to its equity holders except in accordance with Applicable Law (including ACC Regulations), and subject to each regulated utility's obligations to maintain revenues and funds sufficient to fund direct and indirect operating and maintenance expenses (including general and administrative expenses and reasonable and necessary costs, fees and expenses for operation and maintenance of system utilities).
- SECTION 3. CERTIFICATES OR INSTRUMENTS. Any certificates or instruments representing or evidencing Collateral shall be delivered to and held by or on behalf of the Collateral Agent for the benefit of the Secured Parties pursuant hereto and to the Collateral Agency Agreement and be in suitable form for transfer by delivery or accompanied by duly executed instruments of transfer or assignment in blank, all in form reasonably satisfactory to the Collateral Agent and in form and substance reasonably satisfactory to the Bank. Upon the occurrence and during the continuance of any Event of Default, the Collateral Agent will have the right at any time in its sole discretion and without notice to Pledgor to transfer to or to register in the name of the Collateral Agent or its nominees any or all of the Collateral, subject only to the revocable rights specified in Section 7(a) and compliance with ACC Regulations. In addition, on the occurrence and during the continuance of any Event of Default, the Collateral Agent will have the right to exchange certificates or instruments representing or evidencing Collateral for certificates or instruments of smaller or larger denominations.

SECTION 4. SPECIFIED ACCOUNT COLLATERAL.

- The Company has established the Specified Account as a deposit account (within the meaning of Section 9-102 of the UCC), with The Northern Trust Company. Pledgor shall promptly notify the Collateral Agent of any other Specified Account opened or maintained by the Company or Pledgor after the date hereof. To the extent requested by the Collateral Agent following the occurrence and continuation of any Default or the occurrence of any Event of Default, Pledgor shall, and shall cause the applicable depository or other institution to, execute and deliver to the Collateral Agent an account control agreement in form reasonably satisfactory to the Collateral Agent and in form and substance reasonably satisfactory to the Holder which provides among other things for the depository or other institution's agreement that it will comply with: (i) instructions originated by the Collateral Agent directing the disposition of the funds in each such Specified Account that is a deposit account; or (ii) entitlement orders originated by the Collateral Agent with respect to each such Specified Account that is a securities account, in each case without further consent by Pledgor. Notwithstanding the foregoing, any account control agreement which requires the Collateral Agent in its individual capacity to indemnify the depository or other institution other than out of the Collateral will not be reasonably satisfactory to the Collateral Agent.
- (b) Subject to Section 2(d), Pledgor covenants and agrees that upon declaration and payment of any dividend or distribution by any Subsidiary, it will cause such dividend or distribution to be paid and segregated into the Specified Account. Prior to the occurrence and continuance of an Event of Default, amounts so deposited in the Specified Account may be withdrawn and used for corporate purposes, including investments in or loans to Subsidiaries permitted under the Credit Agreement.

SECTION 5. REPRESENTATIONS, WARRANTIES AND CERTAIN COVENANTS. Pledgor represents, warrants and covenants as follows:

(a) The Equity Interests have been duly authorized and validly issued and are fully paid and non-assessable. No Equity Interests constituting interests in limited liability companies

constitute or are evidenced by certificated securities, unless such certificates have been delivered to the Collateral Agent.

- (b) Pledgor is the legal and beneficial owner of the Collateral free and clear of any Lien or other encumbrance except for the Lien created by this Agreement and any other Liens created in favor of the Collateral Agent and described in the Collateral Agency Agreement. There is no existing agreement, option, right or privilege capable of becoming an agreement or option pursuant to which Pledgor would be required to sell or otherwise dispose of any Equity Interest, except as described in the Collateral Agency Agreement.
- (c) Except for the delivery of any certificates or instruments representing Collateral to the Collateral Agent pursuant to this Agreement, filing of an appropriate financing statement with the Arizona Secretary of State, and any control agreement contemplated by Section 4(a), no other action is required to create or maintain the Lien of the Collateral Agent as a valid and perfected first priority Lien (pari passu with the other Secured Parties (as defined in the Collateral Agency Agreement)) in the Collateral to secure the Secured Obligations.
- (d) No authorization, approval or other action by, and no notice to or filing with, any Governmental Authority or regulatory body (except as set forth in Section 5(c)) is required either: (i) for the pledge by Pledgor of the Collateral pursuant to this Agreement or for the execution, delivery or performance of this Agreement by the Pledgor; or (ii) for the exercise by the Collateral Agent of its rights and the rights of any other Secured Parties provided for in this Agreement and the Collateral Agency Agreement or the remedies in respect of the Collateral pursuant to this Agreement and the Collateral Agency Agreement (except as may be required in connection with such disposition by Laws affecting the offering and sale of securities generally and except for compliance with requirements of the ACC Regulations as set forth in Sections 7(b) and 8).
- (e) The execution, delivery and performance of this Agreement does not and will not: (i) violate any provision of any Law (including Regulations T, U and X of the Board of Governors of the Federal Reserve System), order, writ, judgment, injunction, decree, determination or award applicable to Pledgor; (ii) result in breach of, or constitute a default under, any indenture, credit or loan or note agreement or any other agreement, lease or instrument to which Pledgor presently is a party, or by which it or its properties are bound or affected; or (iii) result in or require (other than pursuant to this Agreement) the creation or imposition of any Lien or other share or encumbrance upon or with respect to any properties now owned or hereafter acquired by Pledgor. Pledgor is not in violation of or default under any such Law or material provision of any such indenture, agreement, lease or instrument.
- (f) Schedule I correct sets forth the names of the Pledged Companies and the percentage of Equity Interests in the Pledged Companies owned by Pledgor and pledged by this Agreement.
- (g) Except for any such agreements in favor of the Collateral Agent for the benefit of the Secured Parties and the other Secured Parties (as defined in the Collateral Agency Agreement), Pledgor shall not enter into any agreement providing any Person with "control" (within the meaning of Sections 9-104 or 9-106 of the applicable UCC) of any Specified Account.
- (h) Pledgor is a corporation duly formed under the laws of Arizona, and is validly existing and in good standing under the laws of such jurisdiction. Pledgor has its chief executive office in the State of Arizona. Pledgor agrees to give the Collateral Agent at least 30 days' prior written notice before changing the state in which its chief executive office is located or the state in which it is organized, and prior to the effectiveness of any such change shall take all steps necessary to maintain the security interest provided for herein as a first-priority (pari passu with

the other Secured Parties (as defined in the Collateral Agency Agreement)) perfected security interest, including filing additional UCC financing statements or amendments as may be necessary or requested by (and subject to the rights of) the Collateral Agent.

- (i) Pledgor shall not: (i) sell, assign (by agreement, operation of law or otherwise) or otherwise dispose of, or grant any option with respect to, any Collateral; or (ii) create or permit to exist any Lien or security interest on or with respect to any Collateral, except for the Lien created by this Agreement and any other Lien in favor of the Collateral Agent for the benefit of the Secured Parties.
- (j) Pledgor shall not permit or cause to be issued any Equity Interests: (i) in substitution for any existing Equity Interests; and (ii) in addition to the existing Equity Interests, except following notice to the Collateral Agent and provided that immediately upon its acquisition (directly or indirectly) of any such substitute or additional securities, Pledgor will execute such documentation as is necessary to pledge or evidence the pledge or as may be requested by Collateral Agent or the Holder pledging, and evidencing the pledge hereunder of, such Equity Interests.
- (k) Pledgor shall, at its expense, protect and defend this Agreement, all rights of the Collateral Agent hereunder, and the Collateral against all claims and demands of other parties. Pledgor shall pay all claims and charges that in the reasonable opinion of the Collateral Agent or the Holder might prejudice, imperil or otherwise affect Collateral or the security interest therein. Pledgor shall promptly notify the Collateral Agent of any levy, distraint or other seizure by legal process or otherwise of any Collateral and of any threatened or filed claims or proceedings that might affect or impair this Agreement.
- SECTION 6. FURTHER ASSURANCES. Pledgor authorizes the Collateral Agent to file any financing statements covering the Collateral or any part thereof as the Collateral Agent may desire. Pledgor agrees that from time to time, at the expense of Pledgor, Pledgor will promptly execute and deliver all further documentation and take all further actions that the Collateral Agent or Holder may reasonably request, in order to perfect and protect any security interest granted or purported to be granted hereby or to enable the Collateral Agent to exercise and enforce its rights and remedies hereunder with respect to any Collateral. Pledgor will furnish to the Collateral Agent from time to time statements and schedules further identifying and describing the Collateral and such other reports in connection with the Collateral as the Collateral Agent or Holder may reasonably request, all in reasonable detail, Pledgor agrees, and agrees to cause each issuer of Equity Interests included in the Collateral, to mark its books and records to reflect the Lien of the Collateral Agent in the Collateral.

SECTION 7. VOTING AND DIVIDENDS.

- (a) So long as no Event of Default has occurred and is continuing:
- (i) Pledgor will be entitled to exercise any voting and other consensual rights pertaining to any Collateral for any purpose not inconsistent with this Agreement or any other Loan Documents.
- (ii) Pledgor will be entitled to receive and retain any dividends or distributions paid in respect of the Collateral; provided however, except as expressly permitted by the Credit Agreement, any (A) dividends or distributions paid or payable other than in cash in respect of, and instruments and other property received, receivable or otherwise distributed in respect of, or in exchange for, any Collateral, (B) dividends and other distributions paid or payable in cash in respect of any Collateral in connection with a partial or total liquidation or dissolution, and (C) cash paid, payable or otherwise distributed in redemption of, or in exchange for, any Collateral, shall be forthwith delivered to the Collateral Agent to hold as Collateral and shall, if received by

Pledgor, be received in trust for the benefit of the Collateral Agent, be segregated from the other property or funds of Pledgor, and be forthwith delivered to the Collateral Agent as Collateral in the same form as so received (with any necessary endorsement).

- (iii) The Collateral Agent shall execute and deliver (or cause to be executed and delivered) to Pledgor all such proxies and other instruments as Pledgor may reasonably request for the purpose of enabling Pledgor to exercise the voting and other rights it is entitled to exercise pursuant to Section7(a)(i) and to receive dividends it is authorized to receive and retain pursuant to Section7(a)(ii).
- (b) Subject to any requirements of ACC Regulations, on the occurrence and during the continuation of an Event of Default:
- (i) All rights of Pledgor to exercise voting and other consensual rights it would otherwise be entitled to exercise pursuant to Section 7(a)(i) will automatically cease, and the Collateral Agent will thereupon have the sole right to exercise such rights.
- (ii) All rights of Pledgor to receive the distributions and dividends it would otherwise be entitled to receive and retain pursuant to Section 7(a)(ii) will automatically cease, and the Collateral Agent will thereupon have the sole right to receive and hold as Collateral such dividends, distributions and interest.
- (iii) Any distributions and dividends received by Pledgor contrary to the provisions of Section 7(b)(ii) will be received in trust for the benefit of the Collateral Agent on behalf of the Secured Parties, shall be segregated from other funds of Pledgor and shall be forthwith paid over to the Collateral Agent on behalf of the Secured Parties as Collateral in the same form as received (with any necessary endorsement).

SECTION 8. REMEDIES. Subject to any requirements of ACC Regulations:

- (a) The Collateral Agent may exercise in respect of any Collateral, in addition to other rights and remedies provided for herein or otherwise available to it, all rights and remedies of a secured party on default under the UCC (whether or not the UCC applies to the affected Collateral), and the Collateral Agent may also, without notice except as specified below, sell the Collateral or any part thereof in one or more parcels at public or private sale, at any exchange, broker's board or at any of the Collateral Agent's offices or elsewhere, for cash, on credit or for future delivery, and on such other terms as the Collateral Agent may claim commercially reasonable. Pledgor agrees that at least ten days' notice to Pledgor of the time and place of any public sale or time after which any private sale is to be made will constitute reasonable notification. The Collateral Agent will not be obligated to make any sale of Collateral regardless of notice of sale having been given. The Collateral Agent may adjourn any public or private sale from time to time by announcement at the time and place fixed therefor, and such sale may, without further notice, be made at the time and place to which it was so adjourned.
- (b) Any cash held by the Collateral Agent as Collateral and all cash proceeds received by the Collateral Agent in respect of any sale of, collection from, or other realization on any Collateral shall be applied by the Collateral Agent against the Secured Obligations in such order as the Collateral Agent may elect, subject to the requirements of the Credit Agreement and Collateral Agency Agreement. Any surplus of such cash or cash proceeds held by the Collateral Agent and remaining after payment in full of all Secured Obligations and all other Secured Obligations (as defined in the Collateral Agency Agreement) shall be paid over to the Pledgor or to whomsoever may be lawfully entitled to receive such surplus.

- (c) All rights and remedies of the Collateral Agent expressed herein are in addition to all other rights and remedies possessed by the Collateral Agent or the Holder in the Loan Documents and any other agreement or instrument relating to the Secured Obligations.
- (d) In connection with a public or private sale of any Collateral, the Collateral Agent may disclose to prospective purchasers any non-public information available to the Collateral Agent which pertains to: (i) the issuer of any Collateral; or (ii) Pledgor, provided in the case of Pledgor such non-public information is material to said issuer, its financial condition or the Collateral.
- (e) Without in any way limiting the foregoing, on the occurrence and during the continuation of any Event of Default, the Collateral Agent shall have the right, in addition to all other rights provided herein or by Law, to direct the disposition of the funds in any Specified Account that is a deposit account, or entitlement orders originated by the secured party with respect to each such Specified Account that is a securities account, in each case without further consent by Pledgor.
- (f) If the Collateral Agent exercises its right to take possession of any Collateral, Pledgor shall also at its expense perform any other steps requested by the Collateral Agent or Holder to preserve and protect the security interest hereby granted in the Collateral, such as maintaining Collateral records and filing UCC financing and continuation statements.

SECTION 9. WAIVERS; PRIVATE SALES.

- Pledgor waives any right to require the Collateral Agent or Holder to: (i) proceed against any Person, including any other Obligor; or (ii) proceed against or exhaust any Collateral, or (iii) pursue any other remedy in Collateral Agent's or the Holder's power; and waives any defense arising by reason of any disability of Pledgor or any other Person. Until the Secured Obligations have been paid in full, Pledgor waives any right of subrogation, reimbursement, indemnification, and contribution (contractual, statutory or otherwise), including any right of subrogation under the Bankruptcy Code (Title 11, United States Code) or any successor statute, arising from the existence or performance of this Agreement, and Pledgor waives any right to enforce any remedy which the Collateral Agent now has or may hereafter have against Pledgor or any other Person, and waives any benefit of, and any right to participate in, any security now or hereafter held by the Collateral Agent. Pledgor authorizes the Collateral Agent, without notice or demand and without affecting Pledgor's liability hereunder, from time to time to: (a) renew, compromise, extend, accelerate or otherwise change the time for payment of, or otherwise change the terms of, any Secured Obligations, including increase or decrease of any rate of interest thereon; (b) receive and hold security, other than the Collateral herein described, for the payment of any Secured Obligations, and exchange, enforce, waive, release, fail to perfect, sell, or otherwise dispose of any Collateral or other security; and (c) release or substitute the Company, or any other guarantors of such Secured Obligations or any part thereof, or any other parties thereto.
- (b) Pledgor recognizes that the Collateral Agent may be unable to effect a public sale of any of the Collateral by reason of ACC Regulations and/or certain prohibitions contained in the Laws of any jurisdiction outside the United States or in the Securities Act and applicable state securities Laws, but may instead be compelled to resort to one or more private sales thereof to a restricted group of purchasers who will be obliged to agree, among other things, to acquire such Collateral for their own account for investment and not with a view to the distribution or resale thereof, or otherwise in accordance with the ACC Regulations. Pledgor agrees that any private sale may result in prices and other terms less favorable to the seller than if such sale were a public sale and, notwithstanding such circumstances, agrees that any such private sale shall, to the extent permitted by Law, be made in a commercially reasonable manner. The Collateral Agent will not be under any obligation to delay a sale of any Collateral for the period of time

necessary to permit the issuer of such securities to register such securities under the Laws of any jurisdiction outside the United States, under the Securities Act or under any applicable state securities Laws, even if the issuer would agree to do so. If the Collateral Agent is able to lawfully effect a public sale without registration of the Collateral under the laws of any jurisdiction outside the United States, under the Securities Act or under any applicable state securities Laws, then, subject to any applicable ACC Regulations, the Collateral Agent may, but will not be required to, conduct a public sale of the Collateral, rather than a private sale, if the Collateral Agent reasonably believes it would realize a higher sales price in a public sale.

SECTION 10. COLLATERAL AGENT'S DUTIES.

- (a) The powers conferred on the Collateral Agent hereunder are solely to protect the interests of the Secured Parties in the Collateral and shall not impose any duty on the Collateral Agent to exercise any such powers. Except for reasonable care in the custody of any Collateral in its possession and accounting for moneys actually received by it hereunder, the Collateral Agent will have no duty as to any Collateral or as to the taking of any necessary steps to preserve rights against prior parties or any other rights pertaining to any Collateral. The Collateral Agent will have exercised reasonable care in the custody and preservation of Collateral in its possession if the Collateral is accorded treatment substantially equal to that which the Collateral Agent accords its own property and collateral held for others in its capacity as a collateral agent, it being understood that the Collateral Agent will not have any responsibility for: (a) ascertaining or taking action with respect to calls, conversions, exchanges, maturities, tenders or other matters relative to any Collateral, whether or not the Collateral Agent has or is deemed to have knowledge of such matters; or (b) taking any necessary steps to preserve rights against any parties with respect to any Collateral.
- (b) Upon the appointment of a replacement collateral agent pursuant to the Collateral Agency Agreement, the Collateral Agent may transfer all of its interest and Liens in or any of the Collateral and be fully discharged thereafter from all liability and responsibility with respect to such Collateral so transferred, and the transferee shall be vested with all the rights and powers of the Collateral Agent hereunder with respect to such Collateral so transferred; but with respect to any of its interest and liens in Collateral not so transferred, the Collateral Agent will retain all rights and powers hereby given.

SECTION 11. OTHER RIGHTS.

- (a) The rights, powers and remedies given to the Collateral Agent by this Agreement shall be in addition to all rights, powers and remedies given to the Collateral Agent by virtue of any statute or rule of Law. Any forbearance or failure or delay by the Collateral Agent in exercising any right, power or remedy hereunder shall not be a waiver of such right, power or remedy, and any single or partial exercise of any right, power or remedy hereunder shall not preclude the further exercise thereof; and every right, power and remedy of the Collateral Agent shall continue in full force and effect until such right, power or remedy is specifically waived by a writing executed by the Collateral Agent. The rights and remedies of the Collateral Agent under this Agreement shall be cumulative and not exclusive of any other right or remedy which the Collateral Agent or any Holder may have.
- (b) This Agreement constitutes an assignment of rights only and not an assignment of any duties or obligations of Pledgor in any way related to the Collateral, and the Collateral Agent will have no duty or obligation to discharge any such duty or obligation. The Collateral Agent will have no responsibility for taking any necessary steps to preserve rights against any parties with respect to any Collateral or initiating any action to protect any Collateral against the possibility of a decline in market value. Neither the Collateral Agent nor any party acting as attorney for the Collateral Agent will be liable for any acts or omissions or for any error of

judgment or mistake of fact or Law other than its gross negligence or willful misconduct or negligence in the handling of funds.

In addition to any other powers of attorney contained herein, Pledgor hereby appoints the Collateral Agent, its nominee, and any other Person whom the Collateral Agent may designate, as Pledgor's attorney-in-fact, with full power and authority to sign Pledgor's name on verifications of Collateral; to send requests for verification of Collateral to obligors; to endorse Pledgor's name on any checks, notes, acceptances, money orders, drafts and other forms of payment or security that may come into the Collateral Agent's possession or on any assignments, stock powers or other instruments of transfer relating to any Collateral; to sign Pledgor's name on claims to enforce collection of any Collateral, on notices to and drafts against obligors, on schedules and assignments of Collateral, on notices of assignment and on public records; on the occurrence and during the continuance of an Event of Default to notify post office authorities to change the address for delivery of Pledgor's mail to an address designated by the Collateral Agent; upon the occurrence and during the continuance of an Event of Default to receive, open and dispose of all mail addressed to Pledgor; and to do all things necessary to carry out this Agreement. Pledgor hereby ratifies and approves all acts of any such attorney and agrees that neither the Collateral Agent nor any such attorney will be liable for any acts or omissions nor for any error of judgment or mistake of fact or Law other than such Person's gross negligence or willful misconduct or negligence in the handling of funds. The foregoing powers of attorney, being coupled with an interest, are irrevocable until the Secured Obligations have been fully paid and satisfied and all agreements of any Secured Party to extend credit to or for the account of Pledgor have expired or otherwise been terminated.

SECTION 12. INDEMNITY; WAIVER.

- Pledgor agrees to indemnify, pay and hold harmless, the Collateral Agent and each other Secured Party and any of their Related Parties (each an "Indemnitee"), from and against any and all Indemnified Liabilities, IN ALL CASES, WHETHER OR NOT CAUSED BY OR ARISING, IN WHOLE OR IN PART, OUT OF THE COMPARATIVE OR CONTRIBUTORY NEGLIGENCE OF SUCH INDEMNITEE; provided, Pledgor shall not have any obligation to any Indemnitee hereunder with respect to any Indemnified Liabilities to the extent such Indemnified Liabilities (x) arise from the gross negligence or willful misconduct, as determined by a court of competent jurisdiction by final and nonappealable judgment, of that Indemnitee, or (y) result from a claim brought by Pledgor against an Indemnitee for breach in bad faith of such Indemnitee's obligations hereunder or under any other Loan Document, as determined by a court of competent jurisdiction by final and nonappealable judgment. To the extent the undertakings to indemnify, pay and hold harmless in this Section 12 may be unenforceable in whole or in part because they are violative of any Law or public policy, Pledgor shall contribute the maximum portion it is permitted to pay and satisfy under Applicable Laws to the payment and satisfaction of all Indemnified Liabilities incurred by any Indemnitees. All amounts due under this Section 12(a) shall be payable promptly after demand therefor. For purposes hereof, "Related Parties" means, with respect to any Person, such Person's Affiliates and the partners, managers, directors, trustees, officers, employees or other personnel, counsel, agents and advisors of such Person and of such Person's Affiliates.
- (b) To the extent not prohibited by Applicable Law, Pledgor shall not assert, and Pledgor waives, any claim against the Collateral Agent and its respective Related Parties, on any theory of liability, for special, indirect, consequential or punitive damages (as opposed to direct or actual damages) (whether or not the claim therefor is based on contract, tort or duty imposed by any applicable legal requirement) arising out of, in connection with, as a result of, or in any way related to, this Agreement or any other Loan Document or any agreement or instrument contemplated hereby or thereby or referred to herein or therein, the transactions contemplated hereby or thereby, or the use of the proceeds thereof or any act or omission or event occurring in

10

connection therewith, and Pledgor waives, releases and agrees not to sue on any such claim or any such damages, whether or not accrued and whether or not known or suspected to exist in its favor. No Indemnitee shall be liable for any damages arising from the use by unintended recipients of any information or other materials distributed to it through telecommunications, electronic or other information transmission systems in connection with this Agreement or any other Loan Documents or the transactions contemplated hereby or thereby. In no event shall the Collateral Agent be responsible or liable for any failure or delay in the performance of its obligations hereunder arising out of or caused by; directly or indirectly, forces beyond its control, including strikes, work stoppages, accidents, acts of war or terrorism, civil or military disturbances, nuclear or natural catastrophes or acts of God, and interruptions, loss or malfunctions of utilities, communications or computer (software and hardware) services, it being understood that the Collateral Agent shall use reasonable best efforts which are consistent with accepted practices in the banking industry to resume performance as soon as practicable under the circumstances.

SECTION 13. INTERPRETATION.

- (a) In this Agreement, unless a clear contrary intention appears:
 - (i) the singular number includes the plural number and vice versa;
 - (ii) reference to any gender includes each other gender;
- (iii) the words "herein," "hereof" and "hereunder" and words of similar import refer to this Agreement as a whole and not to any particular Article, Section or other subdivision;
- (iv) reference to any Person includes such Person's successors and assigns but, if applicable, only if such successors and assigns are permitted by this Agreement, and reference to a Person in a particular capacity excludes such Person in any other capacity or individually, provided that nothing in this Section 13(a)(iv) is intended to authorize any assignment not otherwise permitted by this Agreement;
- (v) reference to any agreement, document or instrument means such agreement, document or instrument as amended, supplemented or modified and in effect from time to time in accordance with the terms thereof and, if applicable, the terms hereof, and reference to any note includes any note issued pursuant to any Loan Document in extension or renewal thereof and in substitution or replacement therefor;
- (vi) unless the context indicates otherwise, reference to any Article, Section or Schedule means such Article or Section hereof or Schedule hereto;
- (vii) "including" (with its correlative meaning "include") means including, without limiting the generality of any description preceding such term;
- (viii) with respect to the determination of any period of time, the word "from" means "from and including" and the word "to" means "to but excluding;"
- (ix) reference to any Law means such as modified, codified or reenacted, in whole or in part, and in effect from time to time; and
- (x) references to the Collateral Agent are to it in its capacity as Collateral Agent for the Secured Parties and the other Secured Parties (as defined in the Collateral Agency Agreement) under and as provided in the Collateral Agency Agreement.
- (b) Article and Section headings herein are for convenience only and shall not affect the construction hereof.

- (c) No provision of this Agreement shall be interpreted or construed against any Person because that Person or its legal representative drafted such provision.
- SECTION 14. AMENDMENTS. No amendment or waiver of any provision of this Agreement, nor consent to any departure by Pledgor herefrom, will be effective unless in writing and signed by Pledgor and the Collateral Agent, and then only in the specific instance and for the specific purpose for which given.
- SECTION 15. NOTICES. All notices and other communications provided for hereunder shall be in writing and sent: (a) by telecopy, facsimile or electronic mail if the sender on the same day sends a confirming copy of such notice by a nationally recognized overnight delivery service (charges prepaid); (b) by registered or certified mail with return receipt requested (postage prepaid); or (c) by a nationally recognized overnight delivery service (with charges prepaid). Any such notice must be sent:
- (i) if to the Bank or its nominee, to the Bank or nominee at the address specified for such communications in the Credit Agreement, or at such other address as the Bank or nominee shall have specified to the Company in writing;
- (ii) if to any other Holder, to such Holder at such address as such Holder shall have specified to the Company in writing;
- (iii) if to Pledgor, to Pledgor at the following address: 21410 North 19th Avenue, Suite 220, Phoenix, Arizona 85027 to the attention of: Michael J. Liebman, or at such other address as Pledgor shall have specified to the Collateral Agent and each Holder in writing; or
- (iv) if to the Collateral Agent, at 333 Commerce Street, Suite 900, Nashville, Tennessee 37201, Attention: W. Jones (Global Water Resources, Inc.), wally.jones@usbank.com, or at such other address as the Collateral Agent shall have specified to Pledgor in writing.

Notices under this Section 15 will be effective only when actually received.

SECTION 16. SEPARABILITY. If any clause, sentence, paragraph, subsection or Section of this Agreement is judicially declared to be invalid, unenforceable or void, such decision will not have the effect of invalidating or voiding the remainder of this Agreement, and the part or parts of this Agreement so held to be invalid, unenforceable or void will be deemed to have been stricken herefrom by the parties hereto, and the remainder of this Agreement will have the same force and effect as if such stricken part or parts had never been included herein

SECTION 17. COUNTERPARTS. This Agreement may be executed in any number of counterparts and by different parties in separate counterparts, each of which when so executed will be an original and all of which taken together will constitute one and the same Agreement. Signatures of parties transmitted by facsimile or electronic transmission will be their original signatures for all purposes.

SECTION 18. CONTINUING LIEN. This Agreement shall create a continuing security interest in the Collateral and shall: (a) remain in effect until payment in full of the Secured Obligations; (b) be binding on Pledgor, its successors and assigns; and (c) inure to the benefit of the Collateral Agent and other Secured Parties and their successors, transferees and assigns. Without limiting the foregoing clause (c), any Holder may assign or otherwise transfer all or a portion of its interests, rights and obligations under any Promissory Note in accordance with the Loan Documents. Upon the payment in full of the Secured Obligations and the other Secured Obligations (as defined in the Collateral Agency Agreement), Pledgor will be entitled to the return, on its request and at its expense, of such Collateral as shall not have been disposed of.

SECTION 19. SURVIVAL. All representations and warranties in this Agreement or made in writing by Pledgor in connection herewith will survive the execution and delivery of this Agreement and repayment of the Secured Obligations. Any investigation by any Secured Party will not diminish in any respect its right to rely on such representations and warranties.

SECTION 20. LAW. THIS AGREEMENT AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES HEREUNDER SHALL BE GOVERNED BY, AND SHALL BE CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK WITHOUT REGARD TO CONFLICT OF LAWS PRINCIPLES THEREOF THAT WOULD REQUIRE OR PERMIT THE APPLICATION OF THE LAWS OF ANOTHER JURISDICTION EXCEPT TO THE EXTENT THE PERFECTION OF THE SECURITY INTEREST HEREUNDER, OR REMEDIES HEREUNDER, IN RESPECT OF ANY PARTICULAR COLLATERAL ARE GOVERNED BY THE LAWS OF A JURISDICTION OTHER THAN THE STATE OF NEW YORK.

SECTION 21. JURISDICTION; WAIVER OF JURY TRIAL.

- (a) Pledgor irrevocably submits to the non-exclusive jurisdiction of any New York State or federal court sitting in the Borough of Manhattan, The City of New York, over any suit, action or proceeding arising out of or relating to the Loan Documents. To the fullest extent permitted by Applicable Law, Pledgor irrevocably waives and agrees not to assert, by way of motion, as a defense or otherwise, any claim that it is not subject to the jurisdiction of any such court, any objection that it may now or hereafter have to the laying of the venue of any such suit, action or proceeding brought in any such court and any claim that any such suit, action or proceeding brought in any such court has been brought in an inconvenient forum.
- (b) Pledgor consents to process being served by or on behalf of the Collateral Agent or any Holder in any suit, action or proceeding of the nature referred to in Section 21(a) by hand delivery, reputable overnight commercial delivery service or by mailing a copy thereof by registered or certified or express mail (or any substantially similar form of mail), postage prepaid; return receipt requested, in each case to it at its address described in Section 15 or at such other address of which the Collateral Agent or such Holder shall then have been notified pursuant to Section 15. Pledgor agrees that such service upon receipt: (i) will be in every respect effective service of process on it in any such suit, action or proceeding; and (ii) will, to the fullest extent permitted by Applicable Law, be valid personal service on and personal delivery to it. Notices hereunder will be conclusively presumed received as evidenced by a delivery receipt furnished by the United States Postal Service or any reputable commercial delivery service.
- (c) Nothing in this Section 21 shall affect the right of the Collateral Agent or any Holder to serve process in any manner permitted by Law, or limit any right the Collateral Agent or Holder may have to bring proceedings against Pledgor in the courts of any appropriate jurisdiction or to enforce in any lawful manner a judgment obtained in one jurisdiction in any other jurisdiction.
- (d) THE PARTIES HERETO HEREBY WAIVE TRIAL BY JURY IN ANY ACTION BROUGHT ON OR WITH RESPECT TO OR RELATED TO ANY LOAN DOCUMENT.

SECTION 22. FINAL AGREEMENT. THIS AGREEMENT AND ANY OTHER LOAN DOCUMENTS EXECUTED IN CONNECTION HEREWITH REPRESENT (WITH THE COLLATERAL AGENCY AGREEMENT) THE FINAL AGREEMENT BETWEEN THE PARTIES AND MAY NOT BE CONTRADICTED BY EVIDENCE OF ANY PRIOR, CONTEMPORANEOUS, OR SUBSEQUENT ORAL AGREEMENTS OF THE PARTIES. THERE ARE NO UNWRITTEN ORAL AGREEMENTS BETWEEN THE PARTIES.

SECTION 23. LLC INTERESTS. Pledgor represents and warrants that: (a) any Pledged Company that is a limited liability company has not issued any certificate or other instrument to evidence any of Pledgor's limited liability company interests in such Pledged Company; (b) the limited liability company interest in any such Pledged Company have not been dealt in or traded on any securities exchange or in securities markets; and (c) the operating agreement of any such Pledged Company does not provide that any limited liability company interest in any such Pledged Company shall be a security governed by Article 8 of the Uniform Commercial Code. Pledgor will not: (A) permit any Pledged Company that is a limited liability company to issue any certificate or other instrument to evidence any of Pledgor's limited liability company interests in such Pledged Company; (B) permit the limited liability company interest in any such Pledged Company to be dealt in or traded on any securities exchange or in securities markets; or (C) allow the operating agreement of any such Pledged Company to provide that any limited liability company interest in any such Pledged Company shall be a security governed by Article 8 of the Uniform Commercial Code.

[SIGNATURE PAGE FOLLOWS]

SIGNATURE PAGE TO PLEDGE AND SECURITY AGREEMENT

IN WITNESS WHEREOF, the Pledgor has executed and delivered this Agreement effective as of the date first above written.

PLEDGOR:

GLOBAL WATER HOLDNGS, INC.

By: /s/ Michael J. Liebman

Name: Michael J. Liebman

Title: First Vice President, Secretary and Treasurer

COLLATERAL AGENT:

U.S. BANK TRUST COMPANY, NATIONAL ASSOCCIATION, not in its

individual capacity, but solely as Collateral Agent

By: /s/ Gregory M. Jackson

Name: Gregory M. Jackson

Title: Vice President

Schedule I PLEDGED COMPANIES

Issuer	Class of Shares	Number of Shares	Percentage of Outstanding Shares
Global Water – Farmers Water Company, Inc.	Common	25	100%
Global Water – Ocotillo Water Company, Inc.	Common	1,000	100%
Global Water – Saguaro District Water Company, Inc.	Common	1,000	100%